The Good Wife? Reputation Dynamics and Financial Decision-Making Inside the Household^{*}

Nina BuchmannPascaline DupasYale UniversityPrinceton University and NBER

Roberta Ziparo Aix-Marseille School of Economics

September 27, 2024

Abstract

We study reputation dynamics within the household in a setting where women regularly receive transfers from their husbands for household purchases. We propose a signaling model in which wives try to maintain a good reputation in the eyes of their husband to receive high transfers. This leads them to (a) avoid risky purchases (goods with unknown returns); and (b) knowingly over-use low-return goods to hide bad purchase decisions—we call this the *intra-household sunk cost effect*. We present supportive evidence for the model from a series of experiments with married couples in rural Malawi.

Keywords: Intra-household model, Signaling, dynamic games, experiment, technology adoption. JEL code: D13, J16, O12.

^{*}We are immensely grateful to Aaron Wolf and Maxim Bakhtin for outstanding research assistance, and to all our study participants for their time and attention. The data collection was done with the assistance of IKI Malawi. The research protocols were approved by the Stanford IRB and the Malawi National Committee on Research in the Social Sciences and Humanities. The experimental designs were pre-registered in the AEA RCT registry (ID 4069). We gratefully acknowledge funding from Stanford University. We thank three anonymous referees and the co-editor, as well as Renaud Bourlès, Lukas Bolte, Marcel Fafchamps, Yucheng Liang, Madeline McKelway, Muriel Niederle, Agathe Pernoud, Collin Raymond, David Romer, Erik Snowberg, Alessandra Voena, Leeat Yariv and multiple seminar audiences for very helpful comments and feedback. All errors are our own. Buchmann: nina.buchmann@yale.edu; Dupas: pdupas@princeton.edu; Ziparo: roberta.ziparo@univ-amu.fr.

1 Introduction

Women worldwide rely transfers for on spousal general household expenditures—especially in low-income countries, where women's earnings are limited by a number of factors such as labor-intensive home production (Jayachandran, 2015). Little is known about what drives the size of such transfers and women's discretion over how to spend them. The household bargaining literature primarily attributes the transfer size to spouses' outside options (Manser and Brown, 1980; Chiappori, 1988, 1992). By contrast, this paper studies the role of information asymmetries and women's intra-household reputation in explaining transfer size. In particular, we examine whether the size of discretionary transfers to wives is driven in part by husbands' beliefs that their wives will utilize these transfers wisely, and how this creates incentives for wives to maintain an intra-household reputation as expert investors—even at efficiency costs to the household.

We study the role of women's intra-household reputation both theoretically in a signaling model and empirically in a series of experiments involving over 2,600 spouses in rural Malawi. The key intuition for the reputation dynamics we study is the following. If the husband only gives an allowance to his wife if he believes her to be a savvy investor, then the wife has the incentive to distort information about her real investment expertise. In particular, the wife may: i) under-invest in risky but potentially high-return goods to avoid non-savvy purchases of unproductive goods ("lemons"), and ii) exert costly effort to hide that she has purchased a lemon. Thus, women's reputation concerns may limit their willingness to try out new technologies or abandon non-productive technologies. This has important implications for understanding women's agency and well-being, as well as for designing effective anti-poverty programs such as technology adoption campaigns.

We propose a signaling game in which the husband infers his wife's market expertise from her household goods purchase and usage decisions. The husband decides whether to make a discretionary transfer ("allowance") to the wife, and the wife decides which household good to purchase and whether to use the good after purchase. The wife can purchase either a safe or a risky good. The safe good has a known usage return. The risky good can have either high or zero usage return—for example, a new cookstove advertised as more efficient could truly be more efficient, or it could be no better than the existing one. Wives vary in their expertise in assessing the return of risky goods: Expert wives are able to discern the return of a risky good before purchasing it; non-expert wives only learn the return of the risky good after purchasing it. The husband does not observe the return of the goods himself but learns about his wife's expertise by observing his wife's investment and usage decisions.

In this environment, non-expert wives might i) under-invest in potentially high-return goods and ii) over-use zero-return goods (lemons). First, non-expert wives might underinvest in the risky good, i.e., sometimes purchase the safe good even if its expected return is lower than that of the risky good, to mimic the investment rate of expert wives who purchase the safe good whenever the risky good is zero-return. Second, non-expert wives might exert costly effort to use zero-return goods and hide that they purchased a lemon. This rationalizes a behavior empirically equivalent to the sunk cost fallacy, the tendency to follow through on an endeavor after having invested time, effort, or money into it. The relative importance of these two distortions—under-investment and over-use of lemons—depends on the cost of hiding the purchase of a lemon. If the cost of hiding is low (for example, it is possible to pretend a new stove is efficient by hiding how much firewood is collected and used), non-experts under-invest and hide their investment mistakes. If the cost of hiding is high—including goods for which the husband can directly observe the productivity (for example, it is quasi-impossible to hide that a manual irrigation pump does not have enough suction depth to be useful on one's land), non-experts under-invest even more (completely shying away from risky investments) but do not hide investment mistakes.

We conduct three experiments in Southern Malawi to examine the basic mechanisms of the model and show how these mechanisms could impact real-world behavior. The *transfer experiment*, conducted with 1,093 husbands, suggests that husbands consider their wives' previous investment decisions when deciding how much money to allocate to them. The *signaling experiment*, conducted with 1,093 wives, suggests that wives are willing to incur substantial financial losses to avoid sending a negative signal about their ability to assess the return of market goods to their husbands. The *market experiment*, conducted with a new sample of 675 wives shopping at local markets, suggests that wives are less willing to use spousal transfers to purchase unknown goods without being able to signal that the good is of high return.

The *transfer experiment* tests whether husbands' transfers respond to their wives' perceived expertise. Husbands play a dictator game with a multiplier with their wives and are randomly assigned to a "salience" treatment, in which they recall examples of their wife's (potential lack of) market expertise right *before* the transfer decision. Consistent with the model premise, the salience treatment decreases the share transferred from the husband to the wife by 13% when he rates the wife as having low market expertise, but does not decrease the share transferred when he rates the wife as having high market expertise.

The *signaling experiment* tests whether women internalize intra-household reputation concerns in their investments and usage decisions. We ask wives first to play a quiz discerning high- from low-quality goods (e.g., natural sponge vs. plastic sponge) and then to decide i) whether to share their quiz scores with their husbands in exchange for additional survey compensation ("investing"), and ii) how many quiz answers to correct at a given hiding fee before sharing the score ("hiding"). Consistent with the model predictions, we find that wives expecting a lower score ("non-experts") invest as much as experts but hide mistakes when the cost of hiding is low, and invest significantly less and do not hide mistakes when the cost of hiding is high. The resulting inefficiency cost is quite large. Non-expert wives forgo 125 Malawian Kwacha (MWK) in experimental earnings on average—about 36% of daily income. Specifically, 35% of non-expert wives do not invest at all (i.e., they request that we do not share their score with their husband and thus forgo the additional compensation), and 24% invest but pay to hide mistakes.

The market experiment tests whether intra-household reputation concerns influence women's real-life investment decisions. In a new sample of married women shopping alone at local markets, we elicit women's willingness to exchange some of their survey compensation for an unfamiliar good with high return. We experimentally vary whether the purchase may affect their reputation; specifically, we experimentally vary whether the husband will know that the good is high-return by attaching an "effectiveness" sticker to the good and whether the husband will know that the wife received the good for free by attaching a "donated" sticker to the good. Consistent with the model predictions, non-expert wives, when compared to expert wives, have a 25% lower willingness to pay for the good absent any sticker, but this gap disappears if either of the two stickers removing the reputation risk is attached.

The model predicts that reputation concerns only affect spouses' behavior if the wife's reputation is still above the threshold above which discretionary transfers occur. Consistently, across all three experiments, the patterns predicted by the model are observed only in couples in which wives receive high (above median) monthly transfers from their husbands, that is, in couples in which reputation dynamics are at play. This also suggests that our findings are unlikely to be explained by experimenter demand effects.

We provide theoretical and empirical evidence for reputation dynamics among low-income couples in Malawi but the proposed mechanism could be at play in many different settings. For instance, a California husband might continue using the yogurt maker that produces tasteless yogurt to maintain his reputation as an investor—and be able to buy a Sourdough bread maker in the future. The model could also apply to parents and teenage children, or international migrants and those receiving their remittances. However, we see the dynamics we model as particularly consequential for poor households in low-income countries, where the subordinate position and financial dependence of one spouse relative to the other are still common and exacerbate the issue.

Understanding the extent to which reputation dynamics influence women's ability to experiment with new technologies is also an important step toward understanding the types of policies and programs that can influence technology adoption. The intra-household dynamics we describe may be one of the factors behind the low adoption of new technologies targeting women, such as preventive health products (Cohen and Dupas, 2010; Meredith et al., 2013), improved cookstoves (Berkouwer and Dean, 2021), etc. They could, for example, explain the experimental finding that the marketing of antimalarial bednets in the presence of both spouses increases the purchase rate by 7ppt (+20%) compared to targeting the wife or husband alone (Dupas, 2009). Our findings provide suggestive evidence that marketing campaigns promoting new technologies specifically to women, as many non-governmental organizations do, may generate negative consequences for women who face reputation risks when asked to make investment decisions on their household's behalf.

We contribute to four strands of literature. First, we contribute to the literature on intra-household resource allocations. We connect to the vast literature on bargaining within the household started by Manser and Brown (1980) and Chiappori (1988, 1992). The main friction considered to date in this literature is limited commitment, with transfers within the household determined by the outside options of both spouses. As Doepke and Tertilt (2016) wrote, "An alternative friction that has received much less attention so far is private information within the household." The hiding of income (Hoel, 2015a; Boltz et al., 2016), spending (de Laat, 2014) and savings (Anderson and Baland, 2002; Ashraf, 2009; Dupas and Robinson, 2013; Schaner, 2015) has been well documented. A nascent literature has considered the implications of different preferences on information diffusion within the household (Apedo-Amah et al., 2020; Ashraf et al., 2022). We study the impact of information asymmetries on intra-household resource allocations.

Second, we contribute to the literature on reputation inside the household. Reputation concerns have been proposed as an explanation for intimate partner violence (Tauchen et al., 1991) and tough parenting (Hao et al., 2008; Fu and Pantano, 2015; Hotz and Pantano, 2015): husbands and parents can have a strategic incentive to build a reputation as non-lenient, and thereby dissuade certain behaviors (Hao et al., 2008; Fu and Pantano, 2015; Hotz and Pantano, 2015). We suggest that reputation mechanisms between spouses can also affect financial decision-making and may matter for the design of social policies.

Third, we contribute to the literature on dynamic signaling (Noldeke and van Damme, 1990; Swinkels, 1999; Kremer and Skrzypacz, 2007) and to the literature on incentives in organizations and markets (Dewatripont et al., 1999; Bar-Isaac, 2003).

Finally, our model rationalizes a behavior that is observationally equivalent to the sunk cost fallacy, the greater tendency to continue an endeavor once an investment in money, effort, or time has been made (Arkes and Blumer, 1985). Under standard economic models, this behavior is irrational: once the expense has been incurred, it should be irrelevant to the decision to go on. The idea that such behavior may be driven by the rational need to save face when future payouts are at stake has previously been modeled in the context of firm managers by Kanodia et al. (1989) and Prendergast and Stole (1996). To our knowledge, we are the first paper to suggest the existence of such a phenomenon in a setting where the cost of hiding private information is borne by the agent (the wife)—while in the firm setting, the escalation costs are borne by the principal (the firm).

The remainder of this paper is structured as follows. Section 2 presents the existing

evidence that motivated the paper. Section 3 presents the model and derives a set of testable predictions. Section 4 presents the empirical setting. Section 5, Section 6, and Section 7 describe the transfer, the signaling, and the market experiment, respectively, each designed to test different theoretical premises or predictions. Section 8 tests for heterogeneity by the size of the discretionary transfers from the husband to the wife, and Section 9 concludes.

2 Motivating Evidence: Intra-Household Dynamics and Technology Adoption

In this section, we provide motivating evidence from two randomized experiments and qualitative surveys that suggest that intra-household reputation concerns might contribute to i) the low adoption of new technologies observed in resource-constrained settings and ii) usage behavior akin to the sunk cost fallacy. Low adoption has been documented for technologies that can help improve health (e.g., bed nets or water treatment products), reduce effort costs inside the household (e.g., efficient cookstoves), or increase agricultural yields (e.g., high-yield seed varieties).¹ The literature so far has focused on a number of explanations for why adoption of these products is low despite their returns seemingly far outweighing their costs: lack of information (Hussam et al., 2021), liquidity constraints (Cohen and Dupas, 2010; Tarozzi et al., 2014; Berkouwer and Dean, 2022), inattention (Berkouwer and Dean, 2022) and procrastination (Banerjee et al., 2010). These explanations treat the household as a unit. However, in many contexts, *husbands* are responsible for the generation of income for the household dynamics may contribute to the low adoption of certain household production technologies.

The first experimental study we revisit suggests that demand for a new type of antimalarial bed nets in Kenya is higher when both spouses jointly rather than either of the spouses alone are provided information about the bed nets (Dupas, 2009). In the experiment, either the wife alone, the husband alone, or both spouses together were randomly selected to receive the opportunity to buy the bednet at a randomized price (the randomized price was the focus of Dupas (2009)'s analysis). First, the pre-selected spouse(s) received information about a new type of antimalarial bed net not yet available in the market, a long-lasting insecticidetreated net branded as "Olyset[®]," described as more effective than older generation bed nets. Second, the pre-selected spouse(s) received a voucher for a subsidized Olyset[®]. Third, the pre-selected spouse(s) had three months to redeem their vouchers.

Take-up of the bednet was significantly higher (+7ppts, or 20%) among households in which both spouses were jointly given the information (Figure 1). This is despite the fact

¹See Dupas and Miguel (2017) and Magruder (2018) for reviews in health and agriculture, respectively.

that all participants had three months to redeem the vouchers, i.e., women had ample time to convince their husbands to give them money for the new product, and men to ask their wives if they were interested in the new product. The fact that take-up was higher only when *both spouses* received the same information is consistent with a model in which wives face a reputation risk when returns are uncertain but are expected to cast the deciding vote.



Figure 1: Bed Net Investment Decision in Kenya (2007)

Notes: Data from Dupas (2009). The experiment took place in Kenya in 2007. The sample is limited to households who had to pay a non-zero price for the bed net (N=1,222). The spouse offered the voucher was randomized.

The second experimental study we revisit suggests behavior akin to the sunk cost fallacy: Married but not unmarried women are more likely to use a new water treatment product in Zambia when they paid for the product than when they received the product for free (Ashraf et al., 2010). In the experiment, female household heads willing to purchase the product (called 'Clorin') were randomly offered a discount such that some received the product for free while others had to pay a positive amount.²

Restricting the data to women who had never used Clorin before, we find that married but not unmarried women were more likely (+7ppts, or 18%) to put the product to use if they had to pay some non-zero price for it (Figure 2). This could be suggestive that the sunk cost fallacy operates at the household level: married women feel compelled to use the Clorin for which they paid in order to "justify" their purchase, even if the product appears not well suited for their household.³ However, as single and married women may differ in many dimensions, the finding that price has a differential impact on their usage patterns,

²Ashraf et al. (2010) use this design to test for the sunk cost fallacy and fail to find any evidence that those who ultimately had to pay more were more likely to put the product to use.

³Indeed, more than half of married women who got the Clorin for free report *not* using it at the follow-up visit—this is consistent with the finding that many households dislike the chlorinated taste that can result from water treatment (Dupas et al., 2023).

while suggestive of a potential intra-household sunk cost fallacy, could reflect some other differences between these two types of households. In the next section, we thus generate testable predictions for a formal model, and in the rest of the paper, we describe experiments specifically designed to test these predictions.



Figure 2: Clorin Usage in Zambia, by Final Price paid

Notes: Data from Ashraf et al. (2010). The graph shows adjusted means from OLS regressions with Huber-White robust SEs. The experiment took place in Zambia in 2006. The sample is limited to women with no prior experience using Clorin and who expressed willingness to pay the initial price quoted (N married=388, N single=88). The final price was randomized.

Our last piece of motivating evidence comes from two sets of qualitative interviews. The first set was conducted with a small group of women in Ghana in 2017, and provided useful insights as we wrote the model. When asked what their husband's reaction would be if they brought home a new product, one woman said "he might reduce the amount he will give next time", and another one said "he will be okay as long as the product is useful." A second, larger set of interviews was conducted in Kenya in 2023, i.e., after we wrote the model and ran the Malawi experiments. We surveyed 132 married men and 209 married women. One third of men report they would reduce the allowance to their wife if she bought any product that did not work and 40% of men if she bought an expensive product that did not work. Men also report that their *neighbors* would reduce their wives' allowances by 55% and 61%. respectively, which we take as evidence for social desirability bias in own reports, which are likely an underestimate. In turn, 56% of women report that they consider their husband's reaction when deciding what to buy; and 69% report that either they or their neighbor would have a harder time getting money from their husband after buying a bad product. 31% of women report they can *pretend to use* the bad product and another 27% report they can *use* the bad product in order to prevent their husband from finding out that the product they bought does not work well.

3 A Signaling Model with Endogenous Budget Allocations

In this section, we propose an intra-household signaling model with endogenous budget allocations. First, we describe the set-up with two spouses, one ("husband") who chooses whether to make a transfer and one ("wife") who chooses which good to purchase with the transfer and whether to use it. Second, we analyze the wife's optimal purchase and usage decisions without and with reputation concerns. Finally, we derive testable predictions.

3.1 Setup

Husband's Choice There are two periods: $t \in \{1,2\}$.⁴ In each period, the husband chooses between making a transfer $T_t \in \{0,1\}$ to his wife and investing himself in a project that has value ω for himself and 0 for the wife.⁵ This transfer is in addition to a basic transfer determined by outside options as in the bargaining literature, which we normalize to zero.

Wife's Choices If the wife receives the transfer, she makes both an investment (purchase) and an effort (usage) choice. First, the wife decides whether to buy a safe household good, $g_t = 0$ (e.g., a well-known grain, medicine, or food, or even just savings), or a new risky household good, $g_t = 1$ (e.g., a new grain with potentially higher returns, a new medicine advertised to have fewer side effects, or a new food advertised to be more nutritious).⁶ Second, the wife decides whether to exert effort to use the good she bought, $e_t \in \{0, 1\}$.

The safe good has productivity $\eta_t = \eta^S$. The risky good has productivity $\eta_t = \eta^R$ with probability $\lambda \in (0, 1)$ and productivity $\eta_t = 0$ with probability $1 - \lambda$. The value of the good depends on its productivity and on the wife's usage decision: $y_t = \eta_t e_t$. Using the safe good or the productive risky good does not entail any cost, c = 0. Using the unproductive risky

⁴An alternative modeling choice in games that are played so frequently that the horizon approaches only very slowly and is thus ignored (i.e., it does not enter people's strategic calculations) is an infinite horizon (see section 4.1. in Mailath and Samuelson (2007)). We sketched an infinite-horizon model (available upon request) that we conjecture generates identical predictions.

⁵The investment of the husband if he keeps the transfers can be in either a public or a private good. The evaluation of the value for the wife being 0 is a normalization. The key assumption is that the value of not receiving the allowance is *lower* for the wife than that of receiving the allowance. This assumption reflects the substantial evidence that husbands and wives make different public goods investments (e.g., children's education, Thomas (1990, 1993); Duflo (2003)) and, thus, the wife prefers to have direct control over the allowance (see Afzal et al. (2022a) on demand for agency in the household).

⁶We assume that the types of goods for which wives receive transfers relate to specific purchases/types of investment for which women have a comparative advantage (due to, e.g., differences in information or differences in the opportunity cost of time). This implies that husbands never purchase such goods themselves. However, our model also generalizes to a model of altruistic preferences, in which the husband makes a transfer to the wife, which she uses to buy goods for private consumption. Finally, the model also generalizes to settings where spouses have private income as long as there is some collective sharing of resources and spouses need to decide how to split the shared resources. The model applies as long as the opportunity cost of assigning more resources to women depends on their ability to manage these resources.

good requires the wife to bear a hiding cost c > 0 (for example, pretending a new stove is efficient by hiding how much firewood is collected and used).

Wife Types As investors, wives can be either experts or non-experts, $\theta \in \{E, NE\}$. A wife's expertise is private information to the wife (she learned it through experimentation before marriage).⁷ The expert wife observes the productivity of the risky good, η_t^R , before the purchase decision, while the non-expert wife observes it only after the purchase decision.

The husband updates his beliefs that the wife's type is E, her reputation $P_t \in (0, 1)$, based on the wife's investment and usage decisions, but might not directly observe the productivity of the good.⁸

Payoffs Both spouses enjoy the utility of the goods, but only the wife bears the cost of usage. The husband's utility in each period is

$$U_t^H = \begin{cases} y_t & \text{if } T_t = 1\\ \omega & \text{if } T_t = 0 \end{cases}$$

where ω is the husband's outside option.⁹ The wife's utility each period is

$$U_t^W = \begin{cases} y_t - ce_t & \text{if } T_t = 1\\ 0 & \text{if } T_t = 0. \end{cases}$$

The total utility is a discounted sum of period utilities:

$$U^{i} = U_{1}^{i} + \beta U_{2}^{i} \text{ for } i \in \{H, W\}.$$

Strategies We focus on Perfect Bayesian Equilibria, which require sequential rationality and the beliefs to be determined by Bayes' rule whenever possible. We solve the model by deriving optimal strategies of the husband and both types of wives for all possible starting priors (Appendix Subsection A.1).

Assumptions We impose the following assumptions on the parameters of the model:

⁷With many different types of goods, wives might be experts in some domains and non-experts in others, e.g., a wife could be an expert in farming investments but not in health investments. Thus, one should think of the model as applying to each domain separately, e.g., the husband decides independently whether to make transfers for farming purchases and whether to make transfers for health purchases.

⁸The assumption that the productivity of the good is not immediately observable to husbands is not as extreme as it may seem: as will become clear later, goods whose low productivity can be observed over time are considered to have very high hiding cost, i.e., they are too costly to hide, so they will not be purchased in equilibrium.

⁹To focus on the role of reputation concerns, we assume that the outside option of both spouses is constant. This assumption implies that the bargaining power of the two spouses is constant apart from the reputation dynamics analyzed in the paper.

1.
$$\beta \geq \frac{\lambda \eta^R - \eta^S}{\lambda(\omega - \lambda \eta^R)}$$

2. $\lambda \eta^R > \eta^S > 0$
3. $\omega > \lambda \eta^R, \, \omega < \lambda \eta^R + (1 - \lambda) \eta^S$

Assumption 1 states that the wife is sufficiently patient and hence has reason to care about her reputation.¹⁰ It also ensures that the husband always prefers to have a wife with a higher reputation. Assumption 2 states that the risky good is better than the safe good in expectation, and both are useful. Assumption 3 states that the husband's outside option is better than the safe or the risky good in expectation, but worse than buying the risky good if and only if it is productive. As a result of this assumption, the husband prefers to make the transfer if he knows that the wife is an expert and not to make the transfer if he knows that the wife is not an expert.¹¹

3.2 Optimal Strategies Without Reputation Concerns

We first consider the equilibrium strategies at t = 2. Since this is the last period of the game, the wife does not care about her future reputation, and the strategies serve as a benchmark for the spouses' behavior without reputation concerns.

Lemma 1. At t = 2, the expert wife buys the risky good if and only if it is productive. She always uses the good. The non-expert wife always buys the risky good. She uses the good if and only if it is productive. The husband uses a threshold strategy.

Intuitively, without reputation concerns, everyone plays their static optimal action (the proof is provided in Appendix Subsection A.2). The expert wife buys the risky good whenever it is productive. She always uses the good. The non-expert wife always buys the risky good because it gives a higher payoff in expectation. She uses the good only if it is productive. The husband compares the expected payoff from making the transfer with the outside option. As the expected payoff from the transfer depends on the wife's reputation, the husband makes the transfer if and only if the wife's reputation is sufficiently high.

The equilibrium strategies of both types of wives without reputation concerns correspond to their first-best actions as they maximize household welfare (the weighted sum of the utilities of both spouses) conditional on the transfer. The first-best action of the expert wife (investing

¹⁰It implies that $\beta \lambda \eta^R (1-\lambda) \ge \lambda \eta^R - \eta^S$, which affects the wife's strategy at intermediate levels of costs, as we discuss below.

¹¹Assumptions 1 and 3 are key for the main results of the model. We impose assumption 2 because it fits the low take-up of high-return goods discussed in Section 2 and because, for ethical concerns, we wanted to offer only high-return goods in our experiments. However, the reputation dynamics we highlight are at play even in the opposite scenario (i.e., if this assumption is reversed). Assumption 2 also implicitly assumes risk neutrality in order to study a new mechanism that is unrelated to risk preferences. However, we control for risk preferences across all three experiments.

in the risky good if and only if it is productive) implies that transfers to the expert wife maximize household welfare for any welfare weight placed on the wife as $\lambda \eta^R + (1 - \lambda)\eta^S > \omega$ by assumption. The first-best action of the non-expert wife (always investing in the risky good) implies that transfers to the non-expert wife maximize household welfare for any welfare weight $\geq \frac{\omega - \lambda \eta^R}{\omega}$ (in which case the household welfare with the transfer, $2\lambda \eta^R$, is at least as large as the household welfare without the transfer, $\leq 2\left(1 - \frac{\omega - \lambda \eta^R}{\omega}\right)\omega$).

3.3 Optimal Strategies With Reputation Concerns

Next, we consider the equilibrium strategies at t = 1. As these strategies incorporate reputation concerns, they serve as the basis for testable predictions.

The expert wife, who does not risk losing her reputation as she does not make investment mistakes, uses the same strategy as in the case without reputation concerns. She always buys the risky good if it is productive, which happens at rate λ , and the safe good if it is not. Then, she always uses the purchased good on the equilibrium path.

The non-expert wife risks losing her reputation and transfers from purchasing the risky good at a rate different from λ or from buying the unproductive good and may thus not play her static-optimal action in period 1.

The husband updates his beliefs about his wife's type from his wife's purchase and usage decisions using Bayes' rule. He transfers to the wife if her reputation is at least P_1^* . He always transfers to the expert wife who always plays her static optimal action in period 1 and may or may not transfer to the non-expert wife based on the wife's reputation. As the non-expert wife does not play her static optimal action in period 1, the husband's expected payoff from being married to a non-expert wife is lower in period 1 than in period 2 and $P_2^* < P_1^*$.

The equilibrium strategies of the non-expert wife and the cutoff strategy of the husband P_1^* depend on the hiding cost c. Figure 3 provides a graphical illustration of the equilibria, with each row corresponding to a hiding cost level.

Proposition 1. Suppose the hiding cost is high. The expert wife buys the risky good if and only if it is productive (this happens at rate λ). She always uses the good. The non-expert wife buys the risky good at a rate less than λ unless her reputation is very low. She only uses the good if it is productive. The husband uses a threshold strategy.

This case is depicted in the top row of Figure 3 (the proof as well as the formal definition of low hiding cost and of the husband threshold are provided in Appendix Subsection A.3).

When the hiding cost is high, the non-expert wife does not use the unproductive good because the usage cost would outweigh the reputation benefit.¹²

¹²Note that if the husband can directly observe the productivity of a good, it has hiding cost ∞ .



Figure 3: Hiding Cost

Notes: The figures depict the equilibria of the model: they show the probability of buying the risky good and the probability of using the low-return good, conditionally on buying it in equilibrium, by the cost of using the good for each type of wife. P_1^* is the reputation threshold below which the husband stops giving transfers in the first period, and P_2^* is the reputation threshold below which the husband stops giving transfers in the second period.

As the expert wife never purchases an unproductive good and the non-expert wife does not hide the unproductive good, buying the risky good involves a high risk of losing all reputation. Therefore, the non-expert wife has an incentive to never invest in the risky good and the purchase of the risky good sends a positive signal about the wife's type.

The non-expert wife's equilibrium investment rate balances the first-period payoff gain from the positive signal from investing in the risky good with the expected second-period reputation and payoff loss from potentially making an investment mistake. If the wife's firstperiod reputation is far above the threshold below which transfers will stop in the secondperiod, the expected reputation loss from investing in the risky good is high, and the nonexpert wife invests in the risky good at probability $0.^{13}$ As her reputation approaches the

 $^{^{13}}$ This result relies on the assumption that expert wives make no investment mistakes. If expert wives made (few) mistakes, then *high-reputation* non-expert wives could invest more than expert wives if their expected payoff gain from investing in the risky good outweighed their expected reputation loss. However,

threshold, the non-expert wife increases her investment rate (as the reputation gain from investing into the risky good increases relative to the expected reputation loss from making an investment mistake) until it reaches exactly λ at the threshold so that the husband does not update his beliefs at all. When her reputation is below the threshold, the expected reputation loss from investing in the risky good is low (as it is unlikely the wife will receive transfers in the second period), and the non-expert wife's investment rate is above λ .

Proposition 2. Suppose the hiding cost is low. The expert wife buys the risky good if and only if it is productive (this happens at rate λ). She always uses the good. The non-expert wife buys the risky good at a rate of at least λ but less than 1 unless her reputation is very high. She uses the good if it is productive or if it is unproductive and her reputation is not too low. The husband uses a threshold strategy.

This case is depicted in the bottom row of Figure 3. We describe first the non-expert wife's usage and investment strategies and then the husband's threshold strategy (the proof as well as the formal definition of low hiding cost and of the husband threshold are provided in Appendix Subsection A.4). When the hiding cost is low, the non-expert wife uses the unproductive good and bears the hiding cost to maintain her reputation. She only does not use the good if her reputation is so low that using the unproductive good does not help improve the reputation sufficiently.

The non-expert wife's first-period payoff is increasing in her investment rate (as the expected payoff of the risky good is larger than that of the safe good). Therefore, the non-expert wife has an incentive to always invest in the risky good (at a rate higher than λ) and the purchase of the risky good sends a negative signal about the wife's type. If the non-expert wife invested in the risky good at a rate lower than λ , then purchasing the risky good would be a positive signal about the wife's type. Together with the higher first-period payoff, this would give non-expert wives an incentive to increase their investment rate. Thus, it is not an equilibrium that the non-expert wife invests in the risky good at a rate lower than λ .

The non-expert wife's equilibrium investment rate balances the first-period payoff gain from investing in the risky good with the second-period payoff loss from hiding an investment mistake. If the wife's first-period reputation is far above the threshold below which transfers will stop in the second-period, the reputation gain of investing in the safe good is low, and the non-expert wife invests in the risky good at probability 1. As her reputation approaches the threshold, the non-expert wife reduces her investment rate until it reaches exactly λ at

this would not be a sustainable strategy in a repeated game: Both investing in the risky good and making an investment mistake would signal one to be non-expert and quickly reduce the non-expert wife's reputation below the threshold below which transfers stop. We thus consider the assumption that expert wives never make investment mistakes in a two-period model to be a good approximation of a repeated game in which expert wives make sufficiently fewer (even if not zero) mistakes than non-expert wives.

the threshold so that the husband does not update his beliefs at all. When her reputation is below the threshold, the expected second-period payoff loss from hiding an investment mistake is low (as it is becoming less likely the wife will receive transfers in the second period and thus does not need to hide investment mistakes), and the non-expert wife's investment rate is above λ .

The husband's threshold strategy P_1^* is higher than in the case with high hiding cost (i.e., transfers stop earlier) because being married to a non-expert wife is more costly for the husband in period 1 as he is less likely to find out his wife's type from an investment mistake.

Proposition 3. Suppose the hiding cost is intermediate. The expert wife buys the risky good if and only if it is productive (this happens at rate λ). She always uses the good. The non-expert wife buys the risky good at a rate less than λ unless her reputation is very low. She only uses the good if it is productive or if it is unproductive and her reputation is not too low. The husband uses a threshold strategy.

This case is depicted in the middle row of Figure 3. When the hiding cost is intermediate, the non-expert wife uses an investment strategy like that described in Proposition 1 (high cost) and the same usage strategy as the one described in Proposition 2 (low cost). The proof is provided in Appendix Subsection A.5. When the hiding cost is intermediate, the non-expert wife uses the unproductive good and bears the hiding cost to maintain her reputation. However, as the hiding of investment mistakes is more costly, the non-expert wife shies away from investing into the risky good as much as possible.

The husband's threshold strategy P_1^* is higher than in the case with low hiding cost (i.e., transfers stop earlier) because the husband's expected payoff from being married to a non-expert wife is lower as the wife invests in the risky good with a lower probability.

3.4 Testable Predictions

The results from the previous subsection give us predictions that we empirically test through a series of experiments in Malawi. We test the following key premise and predictions:

Premise 1. Husbands' financial transfers to their wives respond to their beliefs about their wives' expertise as investors.

Prediction 1. When the hiding cost is low, non-expert wives invest no less than expert wives (but less than what would be optimal in the absence of reputation concerns). When the hiding cost is intermediate or high, non-expert wives invest less than expert wives (under-investment).

Prediction 2. Non-expert wives, conditional on investing, hide investment mistakes when the hiding cost is low or intermediate (over-use of lemons). **Prediction 3.** The investment rates of all wives maximize present payoffs when reputation is not at stake, i.e., if the wives do not use the husband transfer for the investment or there is no uncertainty about the quality of the risky good.

Prediction 4. The husband's transfer choice is not affected by the wife's reputation once the wife's reputation is too low. Non-expert wives whose reputation is below the threshold below which transfers stop in the second period do not invest less when the hiding cost is intermediate or high and do not hide investment mistakes.

The rest of the paper tests these predictions. We present the empirical setting in Section 4. In Section 5, we test Premise 1 in an experiment in which husbands transfer money to their wives and we randomly vary the salience of the wife's market expertise reputation. In Section 6, we use a complementary experiment to test Predictions 1 and 2. We do so by randomly varying the hiding cost and studying investment and hiding for wives. In Section 7, we test Predictions 1 and 3 in an experiment involving a real-life purchase decision: we offer an unfamiliar good to women while they are running errands at the market, randomly varying the hiding and reputation costs of the good. Finally, in Section 8, we test Prediction 4 by testing whether results across all three experiments are driven by couples with high transfers from husband to wife.

4 Empirical Setting: Couples in Rural Malawi

The transfer experiment and the signaling experiment were done side-by-side with 1,093 married monogamous couples between May and July 2019. The couples were sampled from 36 villages in Neno district, in Southern Malawi. We selected dwellings randomly and enrolled households in which both spouses were available to participate in an hour-long survey administered separately to the husband and the wife.¹⁴ The surveys administered included standard questions on household demographics, schooling, and employment, as well as a module on expenditures and budget decisions inside the household, recent transfers from the husband to the wife, and financial literacy. In addition, we elicited respondents' performance on six math questions to test respondents' ability to solve everyday math problems, and on 12 Raven's Progressive Matrices (Cattell, 1963) to measure respondents' reasoning ability. The husband survey embedded the *transfer experiment*. The wife survey embedded the *signaling experiment*. Everything took place at the couples' homes, with one survey or speaking with the husband while the other spoke with the wife. Interviews were held outdoors and far enough apart to respect complete confidentiality for both spouses.

 $^{^{14}{\}rm Enumerators}$ used the "left-hand" rule to sample dwellings, as described in Online Appendix Subsection B.1.

Variable	Obs	Mean	Std. Dev.
Years married	1092	9.91	8.48
N of Children	1093	2.63	1.56
Husband's age	1093	35.83	10.16
Husband's education	1093	6.77	3.54
Husband's avg. income last two months (MWK, H's report)	1093	29770.05	33075.29
Husband's avg. income last two months (MWK, W's report)	1093	15506.03	23030.8
Wife's age	1091	30.37	8.93
Wife's education	1093	5.68	3.27
Wife's avg. income last two months (MWK, W's report)	1093	10659.82	17556.52
Wife's avg. income last two months (MWK, H's report)	1093	4967.04	10458.39
Avg. transfers (H to W) last two months (MWK, W's report)	1092	4911.73	8097.11
Avg. transfers (H to W) last two months (MWK, H's report)	1093	8451.97	11411.97

Table 1: Couple characteristics, transfer and signaling experiments

Notes: Kwacha values are winsorized at 3SDs. They represent averages over the preceding two months.

Table 1 provides summary statistics on the couples surveyed. They have been married for an average of 10 years and have 2.6 children. Husbands are an average of 36 years old, have 6.8 years of education, and have earned an average of Malawian Kwacha (MWK) 29,770 (approx. USD 42) per month in the preceding two months (conditional on working). Wives are an average of 30 years old, have 5.7 years of education, and have earned an average of MWK 10,660 (approx. USD 15) per month in the preceding two months (conditional on working). Husbands report transferring an average of MWK 8,452 (approx. USD 12, 28% of their income) per month to their wives in the preceding two months.¹⁵ Interestingly, both wives and husbands substantially underestimate their spouses' income, suggesting that there are indeed substantial information frictions inside the household.

5 Does Reputation Matter for Budget Shares? The Transfer Experiment

We test that husbands' financial transfers to their wives respond to husbands' beliefs about their wives' expertise according to Premise 1 using both transfers reported by husbands and observed in the *transfer experiment*. First, we describe our measure of husbands' perception of their wives' expertise and how it correlates with transfers as reported by husbands. Second, we explain how we experimentally manipulate the salience of the wives' types. Finally, we present dictator game transfers to wives perceived as either experts or non-experts.

¹⁵Wives report transfers that are half in magnitude, suggesting that they might omit substantial transfers that husbands considered. We thus use husbands' reports of transfers whenever available.

5.1 Measuring Market Expertise Reputation

During the survey with the husband, we elicited his beliefs about his wife's market expertise. From this, we construct the wife's Market Expertise Reputation (MER) index, which takes the values 0 to 4, depending on how many of the following questions the husband affirmed: i) his wife has never bought anything that did not work as advertised ("Purchases", 86%), ii) his wife is never tempted by marketing advertisement ("Tempted", 80%), iii) his wife can manage money well compared to other women in the community ("Manage", 70%), and iv) his wife can do calculations correctly in her head when she requests change in the market ("Change", 95%). The distribution of the MER index is as follows: 0.6% of women have an MER of 0, 3.5% have an MER of 1, 13% have an MER of 2, 31% have an MER of 3, and 52% have an MER of 4.

When asked to recall their wives' purchase behavior, husbands were asked to provide examples of instances when the wife was tempted by marketing advertisements and instances when a good purchased by the wife did not work as advertised. Online Appendix Table C.1 lists 50 randomly selected answer choices for each question. The main "flaw" of non-expert wives appears to be gullibility, in the sense that they get easily fooled by vendors. Examples of direct quotes from the husband surveys include: "She bought a drug that wasn't effective at all. She got carried away by what the vendor was telling her"; "She bought a pair of shoes that were not of the required foot size because a vendor told her it will fit"; "She bought atelic 'super dust' that didn't work"; "She bought second-hand burglar bars, which were painted to conceal the rust"; "She was given short trousers by the vendor instead of a skirt."

Consistent with Premise 1, we find that the amount of money a wife receives from her husband is positively correlated with her intra-household reputation as an expert. Controlling for husband and wife characteristics, reported average transfers in the previous two months are monotonically increasing in the wife's MER (Table A.1). On average, women with an MER of 0 or 1 receive MWK 8,187, women with an MER of 2 MWK 8,198, women with an MER of 3 MWK 9,031, and women with an MER of 4 MWK 9,497 (column 1). To verify that men transfer discretionary funds to their wives according to a threshold strategy, we define women as "Low MER" if they have an MER of 0, 1, or 2 (17%). Indeed, having a low MER is associated with a decrease in reported average transfers in the previous two months by MWK 1,086 (13%, column 2). These results are unchanged when we control for an indicator that is 1 if the wife has a below-median "General Ability Reputation" (GAR), a mean effects index (see Kling et al. (2007)) of the husband's beliefs of the wife's scores on six math questions and 12 raven matrices (columns 3-4). We observe little correlation between women's GAR and transfers in the last two months. This is consistent with the model primitive that women could be experts in some domains and non-experts in others and that transfers only respond to women's reputation in finance-related domains. We also find a significant, negative, and large correlation between a low MER and another measure of budget control: whether the husband reports that his wife has access to cash and savings (column 5).¹⁶

The fact that most husbands (83%) believe their wives are experts suggests that either the share of experts in the population is very high, or that men perceive some non-experts as experts as predicted by the model. Nevertheless, the MER as reported by husbands seems correlated with true types: 73% of wives with a low MER report that the statement "I buy things that I later regret because I bought them on impulse" applies to them. This share falls to 61% among wives with a high MER (the p-value of the difference is <0.01).¹⁷

5.2 The Transfer Experiment - Experimental Design

We complement the observational results with a "transfer experiment" in which husbands play a dictator game with multiplier with their wives. The observational correlations provide suggestive evidence in favor of Premise 1 but do not nail causation. For example, the correlation between reputation and transfers could be driven by recall bias: men who hold their wives in high esteem could be more likely to remember a transfer to their wives. We thus test whether husbands' transfers respond to experimentally increasing the salience of the wife's reputation before the transfer choice.

Husbands were offered to choose what share of their experiment compensation of MWK 600 would be doubled and transferred to their wives in one of two treatment conditions. We randomly assigned husbands to either:

- *Salience Treatment:* Husbands played the game at the end of the survey, immediately following the MER module asking them to recall their wife's purchase behavior.
- *Control:* Husbands played the game early in the survey, before the MER module asking them to recall their wife's purchase behavior.

While our model predicts that husbands should transfer less to women with a non-expert reputation in a standard dictator game (as we have already shown they do observationally), we multiply all transfers to the wife to give all husbands an incentive to transfer to their wives. By randomizing how "top of mind" the wife's (potential lack of) market expertise is at the time the husbands make their transfer choice, we can then obtain a causal estimate of the importance of reputation in the husband's allocation decision. To reduce the risk of experimenter demand effects, the husband's decision was not observed by the surveyor toward whom demand effects might be largest: husbands privately placed the transfer to the wife

 $^{^{16}}$ As we show in the next subsection, our results in the *transfer experiment* are also consistent with an equilibrium in threshold strategies with a threshold MER of 3.

¹⁷The correlation between high MER and risk-aversion of the wife is close to 0, suggesting that high MER does not simply proxy risk aversion. The results are robust to controlling for a measure of risk aversion.

in an envelope, which was then handed (with the husband bearing witness) to the surveyor speaking with the wife.

Husbands were explained the dictator game as follows:

"As promised, we are going to give you 600 kwacha for participating in the survey. Here is the 600, please count it to make sure it's correct. But before you pocket it, I am going to offer you a chance to give some of this 600 to your wife. Here is how it will work. You will choose how much of the 600 you want to give to your wife. Whatever you choose to give her, we will double. So if you give 20 to your wife, we will give her 40 and you keep 580. If you give 400, we will give your wife 800 and you keep 200. [...] If you choose to give money to your wife, she will get it right away. We will not tell her where this money is coming from: We will only tell her that this is part of the survey. If you choose to give 0, we will not tell her anything at all."

We estimate the following equation:

$$T_i = \alpha + \beta_1 Low MER_i + \beta_2 S_i + \beta_3 (Low MER_i \times S_i) + \beta'_4 X_i + \mu_e + \delta_c + \lambda_v + \epsilon_i$$
(1)

where T_i is the dictator game transfer of husband *i* to his wife, $LowMER_i$ is a binary variable equal to 1 if the wife's MER score is below 3, and S_i is the assignment of husband *i* to the salience treatment. ϵ_i are Huber-White robust standard errors. We include enumerator fixedeffects μ_e , compensation fixed effects δ_c , and version fixed effects λ_v (see technical Online Appendix subsection B.2 for details on what the different compensations and versions are). We show results both without and with a vector of predictive individual controls X_i .

The treatment and control groups appear balanced within each MER group (Online Appendix Table C.2)—even though we were unable to stratify by MER level since we elicited the MER at the same time as we implemented the experiment.

5.3 The Transfer Experiment: Results

The results are presented in Figure 4. Making the wife's lack of market expertise salient decreases the transfer share by 9ppts (13%) among women with a low MER, but it does not change anything for women with a high MER. In other words, the salience module made husbands of low-MER wives think twice and adjust their transfer downward from what it would have been (results in table form are presented in Table A.2, and the breakdown by MER component is shown in Online Appendix Table C.3.)¹⁸

¹⁸We do not observe that husbands transfer more to high-MER wives without the salience treatment or that the salience treatment significantly increases transfers to high-MER wives. Husbands transfer 69% of the funds to their wives on average—a relatively large share compared to the outcomes of similar games in most other contexts, which are usually not played within couples (Andreoni and Vesterlund, 2001; Andreoni and Miller, 2002; Jakiela, 2013). Given the multiplier, all husbands willing to transfer anything to their wives may already have done so. This is particularly plausible given that money is fungible; that is, husbands could have later deducted the money from a future transfer to the wife. Thus, given the incentive to transfer to wives (which allows us to detect a reduction in transfers as predicted by our model), we only observe that the salience treatment reduces transfers to low-MER wives after reminding the husband about the wife's previous investment mistakes. By contrast, the fact that husbands do not transfer everything to their wives

We conduct several robustness analyses in Table A.2. First, we show robustness to including controls. Second, as we observed in Subsection 5.1 that transfers do not react to wives' general ability reputation, we rule out experimenter demand effects by verifying that the salience treatment did not reduce transfers to wives considered non-experts on the general ability domain (low-GAR wives). Third, we rule out that anger about their wives' previous investment mistakes caused husbands to reduce their transfers by verifying that husbands who may have gotten angry due to another section of the survey (those who scored poorly on the math and raven's quiz) did not reduce transfers. Finally, we rule out that the salience treatment works through other characteristics of the household/match, specifically, whether the wife has an income or the couple has fewer than three children.





Notes: The graph shows adjusted means from OLS regressions with Huber-White robust SEs. The "Salience" treatment was randomized. Low MER is an indicator that takes the value 1 if the wife has an MER of 0, 1, or 2 (see main text). Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{LowMER} , control mean+ $\beta_{Salience}$, and control mean+ β_{LowMER} + $\beta_{Salience}$ + $\beta_{Salience}$ + $\beta_{Salience}$, we show 95% confidence intervals based on the estimated standard errors of β_{LowMER} , $\beta_{Salience}$, and β_{LowMER} + $\beta_{Salience}$ + $\beta_{Salience}$, respectively. Significance from testing equal transfers to high-MER and low-MER wives in control (β_{LowMER} = 0) or in salience (β_{LowMER} + $\beta_{Salience}$ ×LowMER = 0). $p < 0.10^*$, $p < 0.05^{**}$, $p < 0.01^{***}$.

is consistent with the literature on inefficient intra-household allocations. Inefficient resource allocations are well-documented both in real-life agricultural production (Udry, 1996; Dercon and Krishnan, 2000; Andrews et al., 2015; Guirkinger et al., 2015) and in lab-in-the field settings (Kebede et al., 2014; Hoel, 2015b; Iversen et al., 2011). Inefficient allocations might be explained by spouses' demand for agency (Afzal et al., 2022b) and willingness to control and hide resources (Ashraf, 2009; Almas et al., 2018; Jakiela and Ozier, 2015). Both explanations suggest the existence of contracting problems in the household (see discussion by Bakhtiar et al., 2022; Apedo-Amah et al., 2020).

6 Are Women Strategic? The Signaling Experiment

Having shown that husbands act on their beliefs about their wife's expertise as a buyer, we next test Prediction 1—*under-investment*, and Prediction 2—*over-use of lemons* in the *signaling experiment*. This experiment was implemented with the wives of the men who participated in the transfer experiment. First, we describe our measure of wives' perceived expertise and how we experimentally vary wives' hiding costs. Second, we present the investment and hiding choices of both expert and non-expert wives.

6.1 Experimental Design

First, we asked wives to assess the quality (productivity) of different goods in a "quality quiz" to measure their market expertise. In each of six rounds (and two practice rounds), we showed wives a high- and a low-quality version of a good (as determined by our local research team: on several occasions, the low-quality good was a counterfeit good), and asked them to identify the high-quality version. The six goods were a sponge, a water bottle, a razor, a toothbrush, flour, and a candle, and the order in which the goods were presented was randomized. On average, both women and men identified 4.2 high-quality goods correctly.

Second, we elicited wives' beliefs about their score on the quiz to measure their own perceived market expertise. We elicited wives' prior distributions about their score using beans to represent probabilities and visual aids to represent the support (Delavande et al., 2011). That is, for every possible score between 0 to 6, we asked the wife to state her perceived probability that she had received that score using the beans.

Third, we asked wives to decide whether to share their quality quiz scores with their husbands for additional compensation ("investing") and how many mistakes on the quiz to correct against a fee before sharing the score ("hiding"). Specifically, we offered each wife an additional compensation of MWK 200 (approx. USD 0.30, 50% of the original survey compensation of MWK 400, or USD 0.60) if we could inform her husband about her corrected score on the quality quiz at the end of the survey. For the score to be meaningful to the husbands, we administered the same quiz to them as part of the survey during the transfer experiment.¹⁹ However, to ensure that the experiment did not cause any conflict between husbands and wives, the husbands did not learn that the wives had also completed the quiz if the wives chose not to share their scores. This differs from the theoretical set-up in which investing in the safe good does send a signal about the wife's type.²⁰ If the wives chose to

¹⁹Men and women performed equally well on the quality quiz. See detailed scores in Online Appendix Table C.4. Also, consistent with the idea that some non-experts are able to hide their type, we find that husbands whose wives have a high MER are more likely to overestimate their wife's score on the quiz, compared to husbands whose wife has a low MER (55% vs. 43%, p<0.001).

²⁰Ethical concerns are also the reason why we did not simply market "lemons" to wives in order to directly test the over-use of lemons prediction. We believe that our experimental design is the best possible test of

share their scores, husbands received the final scores of the wives (after hiding) without being informed that the wives had been able to pay to improve their scores.

Our design allows us to measure wives' investment and hiding choices as a function of their perceived expertise and the cost of hiding mistakes. We elicited wives' choices in the following order: First, we provided the wife with the unit price she would have to pay to correct a mistake in the quality quiz. We clearly spelled out how many answers she could afford using the experimental payments (the participation fee for the extra activity itself + the participation fee for the survey). We also made it very clear to the wife that she could not purchase answers with her own funds. Second, we asked the wife to decide whether to tell her final (post-hiding) score to her husband. Third, if the wife was willing to share her score for money, we elicited how many mistakes she wanted to correct under each possible scenario (i.e., in case she had 0, 1, 2, ... questions correct). Since the hiding cost was deducted from the compensation fee yet to be paid out, the elicitation was incentive-compatible (the wife could not renege on her correction choice after seeing her score). Finally, after deciding whether to participate in the investment activity and how many mistakes to correct, wives were informed about their initial score as well as the final score to be given to their husbands.

We randomly assigned wives to one of the following hiding costs:

- Low hiding cost: The low cost of MWK 100 (approx USD 0.15) per corrected question allowed wives to correct up to two mistakes when using the additional compensation from participating in the activity and up to six (i.e., all possible) mistakes when using the additional compensation from the activity as well as their survey compensation.
- Intermediate hiding cost: The intermediate cost of MWK 200 (approx. USD 0.30) per corrected question allowed wives to correct up to one mistake when using the additional compensation from participating in the activity and up to three mistakes when using the additional compensation from the activity as well as their survey compensation.
- *High hiding cost:* The high cost of MWK 300 (approx. USD 0.45) per corrected question allowed wives to correct no mistake when using the additional compensation from participating in the activity and up to two mistakes when using the additional compensation from the activity as well as their survey compensation.

In addition, we classified wives according to their prior distributions about their scores:²¹

• "Self-Identified Expert": Women with a mean prior (averaged across the 7 possible scores) of 5 or 6 (61% of the sample). These are women at lower (perceived) risk of sending a "bad" signal to their husbands if they choose to participate without hiding.²²

the model predictions that respect the Belmont Report principle of beneficence.

²¹Since we did not tell women their scores before eliciting their choice, a wife's belief about her own score is the signal she thought the husband would likely receive.

 $^{^{22}}$ This is smaller than the share of wives *perceived* as experts by their husbands (wives with high MER)

• "Self-Identified Non-Expert": Women with a mean prior (averaged across the 7 possible scores) of below 5 (39% of the sample). These are women at greater (perceived) risk of sending a "bad" signal to their husbands if they choose to participate without hiding.²³

We test Predictions 1 and 2 of the signaling model by randomly varying the hiding cost. Specifically, given that participating in the game generated a non-negative payout with certainty, the only rational reason for a wife to *not* participate in the activity for compensation or to participate but pay to hide mistakes is to avoid sending a bad signal about her market expertise.²⁴ By observing non-zero rates of non-participation and hiding, we can already assert that women are concerned about their reputation. By randomly varying the hiding cost, we test whether

- Prediction 1: non-expert wives invest no less than expert wives when the cost of hiding is low but less when the cost of hiding is intermediate or high (*under-investment*).
- Prediction 2: non-expert wives, conditional on investing, hide more than expert wives (*over-use of lemons*) when the hiding cost is low or intermediate.

We estimate the impact of the intermediate and high hiding cost on the participation and hiding behavior by self-identified expertise using the following equation:

$$Y_i = \alpha + \beta_1 N E_i + \beta_2 I C_i + \beta_3 H C_i + \beta_4 (N E_i \times I C_i) + \beta_5 (N E_i \times H C_i) + \beta_6' X_i + \mu_e + \delta_c + \epsilon_i$$
(2)

where Y_i is outcome for wife *i*. NE_i is an indicator that is 1 if the wife self-identifies as non-expert (her average prior about her performance is at most 4 out of 6) and IC_i and HC_i are indicators that are 1 if wife *i* is assigned to the intermediate or high hiding cost. ϵ_i are Huber-White robust standard errors. As above, we include enumerator fixed-effects μ_e , and compensation fixed effects δ_c .²⁵

In addition, we create exogenous variation in quiz performance to address potential concerns regarding the endogeneity of the wife's self-identified expertise. For example, one might be concerned about differences in reporting by type (e.g., non-experts forgo more because they are worse at math). We thus randomized wives into two versions of the quiz:

• *Hard version*: Wives did the quality quiz without hints.

in Section 5, further suggesting the presence of information frictions: husbands perceive some non-expert as expert wives.

 $^{^{23}}$ Self-identified non-expert wives have a significantly lower score than self-identified expert wives (3.9 vs. 4.1, p<0.10). Note that this is not mechanical—women could be off about their own expertise, but the data suggests they are not. Results of the signaling experiment are almost identical when using the modal, minimum, or maximum prior.

²⁴Wives might try to avoid sending a bad signal to their husbands or those to whom the husband might communicate the wife's score. However, our heterogeneity results (discussed in Section 8) are consistent with signaling to the husband rather than others outside the household.

²⁵Controlling for whether the husband was assigned to the salience treatment in the transfer experiment described in Section 5 does not change the results, as expected since the flow of the wife survey and embedded experiment were completely independent of that of the husband's.

• *Easy version*: Wives were provided hints to help them discern the high-quality good during the quiz.

Online Appendix Table C.4 documents that the hints succeeded in increasing wives' performance in the quiz: the hints significantly increased the share of wives who correctly discerned the high-quality good for each of the six pairs, thereby increasing the average score by 1.1 points from 3.6 points (out of 6) in the hard version to 4.7 points in the easy version. In addition, the hints also significantly increased wives' priors about their performance. To verify that this also reduced wives' perceived risk of sending a bad signal, we elicited their second-order beliefs (beliefs about their husbands' beliefs about the wife's score in the quiz), using the visual handouts shown in Online Appendix Figure C.1. As intended, the hints substantially reduced the share of wives whose first-order prior was *lower* than their second-order prior, i.e., who believed their husband would update his belief about his wife's market expertise *downward* if she participated in the investment activity: 31% of wives playing the hard version vs. 22% of wives playing the easy version.

This randomization allows us to test whether wives with an exogenously lower performance participate *less* in the investment activity when hiding is costly, and pay to hide their mistakes when hiding is cheap. We do so by estimating the following equation:

 $Y_i = \alpha + \beta_1 Hard_i + \beta_2 IC_i + \beta_3 HC_i + \beta_4 (Hard_i \times IC_i) + \beta_5 (Hard_i \times HC_i) + \beta_6' X_i + \mu_e + \delta_c + \lambda_s + \epsilon_i + \delta_c + \lambda_s + \epsilon_i + \delta_c + \lambda_s +$

where $Hard_i$ is an indicator that is 1 if the wife was assigned to the hard quiz and IC_i and HC_i are indicators that are 1 if the wife was assigned to the intermediate or high hiding cost. ϵ_i are Huber-White robust standard errors.

6.2 Results

Consistent with theoretical Prediction 1, non-experts do not participate less when the hiding cost is low but participate less when the hiding cost is intermediate or high (Figure 5). Only around 75% of women decide to participate even if the hiding cost is low but rates do not differ by self-identified expertise.²⁶ However, the intermediate and high hiding costs decrease the participation rate of non-expert wives by 12.5ppts (-16%) and 11.2ppts (-15%) respectively (Table A.3, Panel A, column 2). Overall, self-identified non-experts forgo 71 MWK in participation fee on average (about 50% more than self-identified experts, Table A.3, Panel A, column 3).

 $^{^{26}}$ Online Appendix Figure C.2 documents that participation rates are increasing substantially in the women's reported risk preferences. All our results are robust to controlling for risk preferences (we show results with controls in Online Appendix Table C.5).



Figure 5: Game participation in the signaling experiment

Notes: The graph shows adjusted means from OLS regressions with Huber-White robust SEs. Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{NE} , control mean+ β_{IC}/β_{HC} , and control mean+ $\beta_{NE}+\beta_{IC}/\beta_{HC}+\beta_{IC/HC\times NE}$. We show 95% confidence intervals based on the estimated standard errors of β_{NE} , β_{IC}/β_{HC} , and $\beta_{NE}+\beta_{IC}/\beta_{HC}+\beta_{IC/HC\times NE}$, respectively. Significance from testing equal participation of expert and non-expert wives when the hiding cost is low ($\beta_{NE} = 0$), intermediate ($\beta_{NE} + \beta_{IC\times NE} = 0$), or high ($\beta_{NE} + \beta_{HC\times NE} = 0$). $p < 0.10^*, p < 0.05^{**}, p < 0.01^{***}$.

Consistent with theoretical Prediction 2, non-expert wives who participate hide significantly more than expert wives who participate when the hiding cost is low or intermediate: they correct 0.25 more errors when the hiding cost is low and 0.22 more errors when the hiding cost is intermediate, thus paying substantially more in hiding fees (Figure 6 and Table A.3, Panel A, column 6).²⁷ Conditional on participating, non-expert wives still have a significantly lower score than expert wives (-0.36 points, Table A.3, Panel A, column 4).²⁸ However, final scores (sent to husbands) are statistically indistinguishable between non-expert and expert wives (Table A.3, Panel A, column 7).

Note that this set of findings makes it unlikely that our results are explained by experimenter demand effects. In the presence of experimenter demand effects, we would expect non-expert wives (who would have to be responding more to experimenter demand effects than expert wives) to react similarly (at least in direction) to the intermediate and high hiding costs for both participation and errors corrected. That is, non-expert wives would pay attention to the prices and stop correcting and participating either when the cost of hiding is intermediate or the cost of hiding is high. It is implausible that they would change their behavior in response to the prices consistent with the specific pattern that is predicted by our theory, i.e., they participate less when the hiding cost is intermediate or

²⁷We focus on binary types in our analysis to match our proposed model. However, we show in Online Appendix Figure C.3 that forgone earnings are decreasing close to linearly in women's mean prior score.

²⁸Note that the difference in scores between experts and non-experts is smaller when the hiding cost is high since, as predicted, fewer non-experts decide to participate in the game for money.

high but hide less only when the hiding cost is high.



Figure 6: Hiding in the signaling experiment

Notes: Sample restricted to wives who choose to participate in the activity. The graph shows adjusted means from OLS regressions with Huber-White robust SEs. Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{NE} , control mean+ β_{IC}/β_{HC} , and control mean+ $\beta_{NE}+\beta_{IC}/\beta_{HC}+\beta_{IC/HC\times NE}$. We show 95% confidence intervals based on the estimated standard errors of β_{NE} , β_{IC}/β_{HC} , and $\beta_{NE}+\beta_{IC}/\beta_{HC}+\beta_{IC/HC\times NE}$, respectively. Significance from testing equal hiding of expert and non-expert wives when the hiding cost is low ($\beta_{NE} = 0$), intermediate ($\beta_{NE} + \beta_{IC\times NE} = 0$), or high ($\beta_{NE} + \beta_{HC\times NE} = 0$). $p < 0.10^*, p < 0.05^{**}, p < 0.01^{***}$.

Wives forgo experimental earnings on both the extensive (participation) and intensive (hiding) margin in order to avoid sending what they think will be a bad signal. Self-identified non-experts forgo 30% more earnings (MWK +25, from a base of 83) than experts when the hiding cost is low, 63% more (MWK +53, from a base of 84) when the hiding cost is intermediate, and 40% more (MWK +39, from a base of 97) when the hiding cost is high (Figure 7, Panel A, and Table A.3, column 8).²⁹

We do not find that our results are driven by two potential correlates of being non-expert, lower education or self-esteem (Table C.6). Women with low education (fewer than 6, the median, 46% of women) play less but do not differentially so across hiding costs and do not correct more errors. Women with low self-esteem (wives who believe that their quality quiz score is lower than their actual score, 46% of women) play less when the hiding cost is high (even though not significantly so) and do not correct more errors.

²⁹Since our predictions and key estimations concern the interaction between expertise status and the randomly assigned hiding cost, the relevant "balance check" is within each expertise group. We show this in Online Appendix Table C.7.



Figure 7: Total forgone earnings in the signaling experiment

standard errors of β_{NE}/β_{Hard} , β_{IC}/β_{HC} , and $\beta_{NE}+\beta_{IC}/\beta_{HC}+\beta_{IC/HC\times NE/Hard}$, respectively. Significance from testing equal forgone earnings of expert and non-expert wives when the hiding cost is low $(\beta_{NE}/\beta_{Hard} = 0)$, intermediate $(\beta_{NE}/\beta_{Hard} + \beta_{IC\times NE/Hard} = 0)$, or high

(P N E / P II u u = 0), intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P II u u + P I C N E / H u u = 0), of intermediate (P N E / P I u u + P I C N E / H u u = 0), of intermediate (P N E / P I u + 0).

 $(\beta_{NE}/\beta_{Hard} + \beta_{HC \times NE/Hard} = 0). \ p < 0.10^*, p < 0.05^{**}, p < 0.01^{***}.$

We find similar results by exogenous variation in performance (Table A.3, Panel B). The harder version of the quiz succeeded in exogenously lowering women's scores from 4.6 to 3.5 points (column 1). Wives participate as much in the hard as in the easy quiz if the hiding cost is low. However, the intermediate and high hiding costs decrease the participation rates in the hard quiz by 11.8ppts (-16%) and 6.3ppts (-9%) respectively (Table A.3, Panel B, column 2). In addition, wives who participate hide significantly more in the hard than the easy quiz when the hiding cost is low or intermediate (Table A.3, Panel B, column 5). Overall, wives in the hard quiz forego 14% more earnings (MWK +12, from a base of 85) when the hiding cost is intermediate, and 35% more (MWK +48, from a base of 78) when the hiding cost is high (Figure 7, Panel B, and Table A.3, Panel B, column 8). This is consistent with wives in the hard quiz being more likely to believe that their husbands will update their beliefs downwards and wanting to avoid sending this bad signal.³⁰

³⁰Overall, our results using the exogenously varied scores are slightly noisier, which might be due to some random imbalances. For example, wives in the hard quiz with intermediate or high hiding costs have a higher

We also present results by two alternative measures of expertise, the wife's quality quiz score and the accuracy of her beliefs (Online Appendix Table C.10). First, we find that wives with a low quality quiz score (5, the median, 39% of women) do not play less when the price of hiding is high but hide more mistakes. Second, we find that wives who have inaccurate beliefs about their quality quiz score (they are off by a score of more than 1, the median, 40% of women) play less when the price of hiding is high but do not hide more mistakes.

Financial or Non-Financial Reputation Incentives? While theoretically plausible, we consider it unlikely that our experimental results are driven by wives trying to maintain their intra-household reputation to avoid non-financial consequences of a low reputation, such as domestic violence or emotional abuse. The model encompasses such an alternative: transfers could be not only financial but also in-kind (with abuse being a negative transfer).³¹ From an ethical standpoint, however, the implications of our experimental design are quite different across the two interpretations. Specifically, if the risk of abuse increases in response to poor investment choices, could our signaling experiment have put our participants at risk? We piloted the protocols extensively in settings where women could freely share with us their concerns, and abuse was never brought up.³² Our enumerators, after sharing the wife's performance with the husband (with the wife present), systematically witnessed husbands congratulating their wives on their good performance (the wife's final reported score was 4.6 out of 6 on average, higher than the husbands' average of 4.2). Furthermore, the finding that reputation matters for transfers in the transfer experiment, as well as our heterogeneity analyses, are much better aligned with the hypothesis that women attempt to maintain their reputation to receive financial transfers (rather than to avoid abuse). We find strong heterogeneity by transfers, suggesting that women who could lose more financially also respond more to our experiments (see Section 8). Finally, the 2015/2016 Malawi DHS shows that spousal violence is decreasing in both the wife's and husband's level of education (p. 285–286, NSO and ICF (2017)) but we do not find the playing and hiding patterns by either the wife's or husband's median education.

market expertise reputation, have been married for fewer years, and have a lower math score and education (see Online Appendix Table C.8.).

 $^{^{31}}$ It could also be that people care about their reputation for self-image reasons. An ideal way to test this would have been to implement a version of experiment 2 with a quiz in a domain completely unrelated to household finances. We did not do such a placebo experiment. Nonetheless, the fact that we see heterogeneity by size of intra-household transfers (see Section 8) suggests that self-image concerns would need to be higher for wives in households with higher transfers for this mechanism to explain the entirety of the results, and this seems unlikely.

³²Domestic abuse is present but not widespread in our study setting: The share of adult women in Neno District who reported experiencing physical or sexual intimate partner violence in the past 12 months is 12% (NSO and ICF, 2017), compared to 25% in Kenya (KNBS, Ministry of Health and ICF, 2017), 27% in Bangladesh (Bangladesh Bureau of Statistics, 2016), 9% in Guatemala (MSPAS, INE and ICF, 2017) and 5% in the US (Smith et al., 2018).

7 The Market Experiment

The market experiment tests Prediction 1 for real-life investment rates—do non-experts invest no less than experts when the hiding cost is low but less when the hiding cost is intermediate or high?, and Prediction 3—do non-expert and expert wives invest at the same rate (the payoff-maximizing rate) when their reputation is not at stake? Following the completion of our lab-in-the-field experiments, we conducted short surveys and a field experiment with 675 married women in monogamous relationships, recruited while they were shopping at one of six local markets in Zomba district in July 2019.³³ We did not conduct the market experiment with the wives in the transfer and signaling experiment because the market experiment required us to present the wives with goods that they could have acquired while shopping alone at the local markets.³⁴ Therefore, we recruited women who were shopping alone at the market.

The married women in the market experiment have characteristics similar to those of the first two experiments (Online Appendix Table C.9). They have been married for an average of 10 years and have 2.5 children. They are an average of 30 years old and have 7.3 years of education. They have earned an average income of MWK 15,117 (approx. USD 23) in the preceding two months, are married to husbands whom they report have earned an average income of MWK 20,397 (approx. USD 30) in the preceding two months, and report average transfers from their husbands of MWK 11,293 (approx. USD 17) in the preceding two months.

We first describe our measure of wives' expertise and the experimental design and then present the investment choices of both expert and non-expert wives.

7.1 Experimental Design

We use an alternative measure of expertise as we could not administer the quality quiz used in the signaling experiment and elicit women's beliefs about their scores on that quiz.³⁵ To identify this alternative, we used a random forest algorithm to predict the perceived score in experiment 2 from all exogenous husband and wife characteristics.³⁶ The random forest

 $^{^{33}}$ We initially aimed to recruit 1000 married women, as pre-registered, but due to delays, we did not complete the data collection before the research associate overseeing the field work had to leave the country for a new position.

 $^{^{34}\}mathrm{It}$ is extremely common for women to go to the market on their own in this context.

³⁵There are two reasons for this: (i) the quiz could have created tensions vis-a-vis market vendors, some of which were selling some of the low-quality goods in the quiz (since we procured them from local markets); and (2) going through the quiz takes quite some time because lengthy instructions need to be given.

 $^{^{36}}$ We chose a random forest algorithm as it minimizes over-fitting for out-of-sample predictions (Breiman, 2001). We tuned the hyper parameters, the number of trees and the variables considered at each split, using a grid search. We tried values between 25 and 400 in steps of 25 for the number of trees and between one and the number of independent values for the number of variables. We chose the algorithm that minimizes the out-of-bag error rate: 250 trees and two variables. The out-of-bag error rate is 0.93 (on a scale of 0–6) with these optimal hyper parameters.

algorithm chose the wife's second-order beliefs about her husband's beliefs about her market math score (elicited using the visual handout shown in Online Appendix Figure C.1) as the most important predictor of the wife's perceived score in experiment 2. Given this finding, we only administered the 6 market math questions to the market experiment sample and elicited second-order beliefs.

- "Non-Experts": Women with a second-order belief about their math score of at most 4 out of 6 (44% of the sample).
- *"Experts"*: Women with a second-order belief about their math score of 5 or 6 out of 6 (56% of the sample).

We also show results when classifying women with a below-median random-forest predicted perceived score as non-experts (we also show results using this classification for experiment 2).

Using a Becker-DeGroot-Marschak (BDM) mechanism (Becker et al., 1964), we elicited women's willingness to trade part or all the cash value of their survey compensation of MWK 1,000 (approximately USD 1.5) for an unfamiliar good. This generated an estimate of their willingness to pay for the unfamiliar good without the amount of cash they had at the time of the survey being a constraint.

We randomly offered women one of two unfamiliar goods with different hiding costs:

- Low hiding (time) cost: An airtight crop storage bag purchased in Blantyre, the second largest city in Malawi. These bags are hermetically sealed to protect harvested grains (e.g., maize, red beans) from insect pests. They are used to store grains for months on end. While the returns of these bags are substantial, they were unknown to the women. The usage cost of a storage bag is low (in terms of the model, the hiding cost is low). This is because once the bag has been filled, even if it did not truly protect from pests, leaving the grains in the bag for the rest of the season would have no cost (since the status quo is to store the grains in non-sealed bags). In addition, if some of the grain rots, the wife can sort through and throw it away while the husband is absent.
- *High hiding (time) cost:* A child picture book (imported from overseas by the research team): Either Richard Scarry's "A Day at the Airport" picture book, or the illustrated "Lift-the-flap" Animal ABC baby book by Jonny Lambert. Such books are relatively expensive (USD 10 before shipping costs) and totally unavailable in Malawi: Even low-quality picture books are completely absent from even markets in the capital city. The evidence on the benefits of showing books and describing pictures to very young children is strong (but underestimated by parents worldwide). Board books are meant to be looked at/shown/read to children over and over again. The usage cost (hiding

cost) is thus high because the good needs to be used repeatedly as it is obvious if it stays on the shelf for too long without being used.³⁷

We implemented the intervention on market days so that, in case the respondent brought the good home, the husband would infer that the respondent had bought the good at the market.

To test Predictions 1 and 3 of the model, women were further randomly allocated to one of four sticker groups:

- Donated: We put a "donated by Stanford University" sticker on the good.
- *Effectiveness:* We put a sticker on the good describing its proven effectiveness (e.g., describing the positive effects of reading/looking at picture books with children).
- Both: We put both Donated and Effectiveness stickers on the good.
- None: We put no stickers on the good.

The stickers are shown in Online Appendix Figures B.4 and B.5.³⁸ When deciding whether and how much to invest, the women could see the stickers. Hence they knew what information would be available to their husband. Specifically:

- The donated sticker gave the woman the guarantee that the spouse would see that the good was acquired at no financial cost to the household (as it indeed was, since it was given in exchange for her time). In such a case, the reputation mechanism in the model is not at play as the cost of the good to the husband is 0, i.e., the payoff to the husband cannot be negative. The prediction is that investment rates should differ across experts and non-experts absent the sticker, but not in the presence of the sticker.
- The effectiveness sticker aimed to eliminate uncertainty about the quality of the risky good to the husband ($\lambda = 1$). The prediction is that investment rates should differ across experts and non-experts absent the sticker, but not in the presence of the sticker.³⁹

Importantly, to ensure that the stickers did not work by changing women's beliefs, all women in all treatment arms were shown "flyer" versions of the stickers and read the information on the stickers by the enumerators before making their investment decisions. However, only in the sticker treatment arms were the stickers attached to the goods, and women could take the information home.

 $^{^{37}}$ Because we had to bring the books by plane in a suitcase (there is no reliable shipping service to Malawi), we were only able to offer the books to 26% of the women in the sample.

 $^{^{38}}$ Even though literacy is high in our setting (88% among adult men and 78% among adult women in Zomba district (NSO and ICF, 2017)), we attempted to make the sticker content as clear as possible using images.

³⁹To minimize a potential concern that the effectiveness sticker could be interpreted as a marketing ploy by husbands, we purposefully designed the stickers as information leaflets added to the products externally. This contrasts with traditional advertisements that are commonly integrated into product packaging.

We estimate the impact of the stickers on the willingness to pay using the following equation:

$$WTP_{i} = \alpha + \beta_{1}NE_{i} + \beta_{2}D_{i} + \beta_{3}Eff_{i} + \beta_{4}(D\&Eff_{i}) + \beta_{5}(NE_{i} \times D_{i}) + \beta_{6}(NE_{i} \times Eff_{i}) + \beta_{7}(NE_{i} \times D\&Eff_{i}) + \beta_{8}'X_{i} + \mu_{e} + \delta_{m} + \epsilon_{i}$$

$$(3)$$

where WTP_i is woman *i*'s willingness to pay for the good. NE_i is an indicator that is 1 if the woman is classified as a non-expert wife, D_i is an indicator that is 1 if woman *i* is assigned to the donated treatment arm, Eff_i is an indicator that is 1 if woman *i* is assigned to the effectiveness treatment arm, and $D\& Eff_i$ is an indicator that is 1 if woman *i* is assigned to both stickers (donated and effectiveness). ϵ_i are Huber-White robust standard errors. We include enumerator fixed-effects μ_e and market fixed effects δ_m and show estimations with and without adjusting for individual controls X_i .⁴⁰

Wife characteristics are broadly balanced by treatment arms, though not perfectly (see Online Appendix Table C.11). Results are similar both with and without controls.⁴¹

7.2 Results

Consistent with Prediction 1, non-expert wives have a lower willingness to invest than experts without stickers but not with the stickers (Figure 8; the full estimation results are shown in Table A.4). For simplicity, we pool both stickers into one "sticker arm" since their predicted effects are of the same sign and their observed effects cannot be distinguished from each other (Table A.4). Without the stickers, expert wives have an average willingness to pay of MWK 351 and non-expert wives have an average willingness to pay of MWK 265 (-25%, column 2). Neither the donated nor the effectiveness sticker affect the willingness to pay of expert wives whose investment decisions already maximize their present payoffs, but any sticker increases the average willingness to pay of non-expert wives by MWK 93 (+26%, column 5), such that the willingness to pay of expert and non-expert wives is not statistically different if the good is offered with either of the two stickers.⁴² All results are robust to

 $^{^{40}}$ We did not specify that we would focus on heterogeneity by expertise in the AEA RCT registration for the study, nor did we specify how we would proxy expertise. The former was an oversight (our model *is* about types and all our predictions are about types). The latter was that we did not have data from experiment 2 yet by the time we registered since we registers all three experiments at once. For transparency, we show the results estimating the effects of the stickers on the pooled sample of expert and non-expert wives in columns 6-9 of Table A.4.

⁴¹The most concerning imbalance is that expert wives assigned to the effectiveness sticker treatment have been married longer, hence have larger families, and received a significantly lower transfer from their husbands in the previous two months. This is not the case for non-expert wives assigned to that arm. This means that a differential impact of the effectiveness sticker by expertise could possibly be due to these differences, especially those in income. These could not explain differences in the impacts of the 'donated sticker', however.

⁴²The stickers (insignificantly) decrease the willingness-to-pay of expert wives, and the effect of the two combined stickers is in the same direction as each sticker alone, although somewhat muted and not significant. Anecdotally, it seems that expert wives perceived the goods as "tempered" with the stickers, and the two

omitting our vector of controls and controlling for market fixed effects.



Figure 8: Market experiment: Willingness to pay, by wife's expertise

Notes: The graph shows adjusted means from OLS regressions with Huber-White robust SEs. The dependent variable is the willingness to pay in Malawian Kwacha. Non-Expert is an indicator that takes the value 1 if the wife has a second-order belief about her math score of at most 4 out of 6. Any Sticker is an indicator that takes the value 1 if either the donated or effectiveness sticker was attached to the good. Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{NE} , control mean+ $\beta_{AnySticker}$, and control mean+ $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$. We show 95% confidence intervals based on the estimated standard errors of β_{NE} , $\beta_{AnySticker}$, and $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$, respectively. Significance from testing equal willingness to pay of expert and non-expert wives in control ($\beta_{NE} = 0$) or in the sticker treatments ($\beta_{NE}+\beta_{NE\times AnySticker} = 0$). $p < 0.10^*, p < 0.05^{**}, p < 0.01^{***}$.

Consistent with Prediction 3, we find stronger effects among women offered a book (high hiding cost) than women offered the bags (low hiding cost, Figure 9). The investment gap between expert and non-expert wives absent a sticker is much greater when the hiding cost is high than when it is low. These findings seem to be again inconsistent with experimenter demand effects, which would not predict differential behavior for the two goods.⁴³

The investment gap between expert and non-expert wives is greater when we increase the salience of the husband-wife relationship (Figure 10), suggesting that it is indeed due to intra-household reputation concerns (as opposed to, for example, concerns about friends or neighbors). In half of the sample the BDM was played before the survey and in half of the sample the BDM was played after the survey, i.e., after eliciting the wife's second-order

stickers combined occupied too much space on the goods and therefore made them less attractive.

⁴³The results could possibly be due to differences in preferences of husbands and wives for the goods, with husbands particularly disliking the book without stickers. Shying away from bringing home a good that the husbands dislike could be the result of reputation dynamics or simply due to the wife caring about her husband's utility. We view the likelihood that husbands have preferences over the presence of stickers as unlikely, however.

beliefs and asking her about previous transfers and her financial decision-making inside the household.⁴⁴



Figure 9: Market experiment: Willingness to pay, by wife's expertise and hiding cost

Notes: The graph shows adjusted means from OLS regressions with Huber-White robust SEs. Regressions are run separately for bags (left Panel) and books (right Panel). The dependent variable is the willingness to pay in Malawian Kwacha. Non-Expert is an indicator that takes the value 1 if the wife has a second-order belief about her math score of at most 4 out of 6. Any Sticker is an indicator that takes the value 1 if either the donated or effectiveness sticker was attached to the good. Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{NE} , control mean+ $\beta_{AnySticker}$, and control mean+ $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$. We show 95% confidence intervals based on the estimated standard errors of β_{NE} , $\beta_{AnySticker}$, and

 $\beta_{NE} + \beta_{AnySticker} + \beta_{NE \times AnySticker}$, respectively. Significance from testing equal willingness to pay of expert and non-expert wives in control ($\beta_{NE} = 0$) or in the sticker treatments ($\beta_{NE} + \beta_{NE \times AnySticker} = 0$). $p < 0.10^*, p < 0.05^{**}, p < 0.01^{***}$. The p-values on the interactions "Any Sticker × Book" and "Non-Expert

 $p < 0.10^{\circ}, p < 0.05^{\circ\circ}, p < 0.01^{\circ\circ\circ}$. The p-values on the interactions "Any Sticker × Book" and "Non-Expert × Any Sticker × Book" in the fully interacted model are 0.40 and 0.14, respectively.

Finally, we note that all results presented in this section are robust to using the randomforest predicted types instead of the second-order beliefs (Figures A.1 to A.3).

Overall, the findings of the market experiment are in line with the idea that women internalize potential reputation costs when making real-life investment decisions. These results suggest that women might have a limited ability to experiment with new technologies unless it is ensured that they are able to credibly convey certain information to their husbands.⁴⁵

⁴⁴This also alleviates concerns that the stickers might operate by changing the wife's beliefs about the goods. The salience effect as well as the heterogeneity by discretionary transfer size shown in the following section further rule out another potential mechanism: the stickers may help the wife convince the husband to use (or let her use) the product.

⁴⁵Communication alone (just letting the husband and wife communicate) may not be effective in improving the ability of women to experiment with new technology: the existing evidence on the limited ability of

Figure 10: Market experiment: Willingness to pay, by wife's expertise and relationship salience



Notes: The graph shows adjusted means from OLS regressions with Huber-White robust SEs. Regressions are run separately in the control treatment (left Panel) and the relationship salience treatment (right Panel). The dependent variable is the willingness to pay in Malawian Kwacha. Non-Expert is an indicator that takes the value 1 if the wife has a second-order belief about her math score of at most 4 out of 6. Any Sticker is an indicator that takes the value 1 if either the donated or effectiveness sticker was attached to the good. Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{NE} , control mean+ $\beta_{AnySticker}$, and control mean+ $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$. We show 95% confidence intervals based on the estimated standard errors of β_{NE} , $\beta_{AnySticker}$, and $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$, respectively. Significance from testing equal willingness to pay of expert and non-expert wives in control ($\beta_{NE} = 0$) or in the sticker treatments ($\beta_{NE}+\beta_{NE\times AnySticker} = 0$). $p < 0.10^*, p < 0.05^{**}, p < 0.01^{***}$. The p-values on the interactions "Any Sticker × Salience" and "Non-Expert × Any Sticker × Salience" in the fully interacted model are 0.94 and 0.74, respectively.

8 Heterogeneity by Discretionary Transfer Size

In this section, we test Prediction 4 of the model by assessing whether the results in all three experiments are driven by households in which women still receive discretionary transfers from their husbands. Our model predicts that spouses' behavior should stop responding to the wife's reputation once the reputation has fallen below the threshold above which discretionary transfers occur in the second period.

To test Prediction 4, we compare experimental results in households in which the wife receives only "subsistence level" transfers (i.e., transfers for basic household necessities) and

spouses to convey effective information is growing (Conlon et al., 2022; Fehr et al., 2022; Bjorkman Nyqvist et al., 2023), especially in contexts in which spouses may have conflicting interests (Ashraf et al., 2022) or contracting problems (Apedo-Amah et al., 2020). This suggests that policies providing external tools to reduce information gaps between spouses may be beneficial.
households in which she receives additional discretionary transfers for investments.⁴⁶ This corresponds to households below or above the median transfer size. We use husbands' reports where available (experiments 1 and 2) as wives' reporting could be correlated with the wife's type. Transfers reported by husbands respond substantially to wives' perceived expertise, with a low MER reducing transfers by 13% (-MWK 1076, Appendix Table A.1).



Figure 11: Heterogeneity by discretionary transfer size

Notes: The graph shows the coefficients and confidence intervals from OLS regressions with Huber-White robust SEs. Low/High Transfers correspond to below/above the median. Rows 1 to 3 control for enumerator and compensation fixed effects (and version fixed effects for the transfer experiment) as well as the wife and the husband's age, education, average income in the last two months, variability of income (whether income is the same in most months or varies a lot), risk preferences, math and raven scores, and years married, number of children and number of household members, and MER index. Controls are as reported by the husband in the transfer and signaling experiment and as reported by the wife in the market experiment. Row 4 controls for enumerator and market fixed effects as well as the wife's age, education, average income in the last two months, risk preferences, math score, as well as the husband's average income in the last two months, years married, and the number of children and household members. Coefficients are presented as percentage point deviations from the control means.

Consistent with Prediction 4, we observe significant estimates only in households in which the wife's reputation is still above the threshold, and women thus still receive discretionary transfers (Figure 11, all coefficients are shown in percentage point deviations from the control mean). First, only husbands who transfer to their wives for non-necessities investments (high transfers) reduce their transfers in the salience treatment if their wife has a low MER in the transfer experiment.⁴⁷ Second, only non-expert wives who still receive transfers from their husbands reduce their participation in the high hiding cost treatments (as the prediction is the same for both intermediate and high hiding costs, we pool both costs, however, the results are the same for both costs individually) or hide mistakes to avoid sending a signal in the

⁴⁶We use transfers to classify households since we have this information for all three experiments.

⁴⁷All husbands should use the dictator game to transfer for basic necessities (given the multiplier) and reduce their own private transfers to the wife afterwards.

signaling experiment.⁴⁸ Finally, only non-expert wives who still receive transfers from their husbands increase their willingness to pay for the unfamiliar goods in the sticker treatments in the market experiment (we show results using the random-forest predicted expertise in Online Appendix Figure A.1). Taken together, our results thus provide strong evidence for the external relevance of the experiments as spouses' behaviors seem to be driven by reallife reputation concerns inside the household (and not, for example, experimenter demand effects).

We also find that husbands and wives in households with more children, a proxy of budget tightness, react more across the three experiments (even though the differences in reaction are not statistically different; see Online Appendix Figure C.8).

9 Conclusion

This paper offers a new perspective on some potential dynamics at play between spouses in contexts where women are specialized in household production and at least partly depend on their husband's income. We develop a signaling model in which a woman's access to the household budget varies with her husband's perceptions of her skills as an investor. Our theoretical results thus suggest that, to maintain control over a greater share of the budget, women may experiment too little when they have difficulty assessing the productivity of new goods and technologies or incur costs to hide bad purchase decisions. Hence, our model could explain behavior akin to the sunk cost fallacy—using a product even after one has realized it does not have positive returns—within the realm of neoclassical economics.

Three experiments were designed to test specific pieces of the theory. The transfer experiment shows that husbands whose wives made bad market choices in the past transfer less to their wives in a dictator game with a multiplier if asked to recall these choices just before playing the game. The signaling experiment suggests that women forego earnings to avoid sending a bad signal about their investment skills. Finally, the market experiment suggests that wives are less willing to purchase unknown goods without being able to signal that the good is free or of high return. In all three experiments, results are driven by couples in which wives still receive transfers from their husbands—providing additional evidence for the external relevance of our experimental findings.

Our paper brings to light novel insights about the role of dynamic reputation concerns and how they could influence both investment and usage decisions. In particular, we present experimental evidence suggesting that reputation concerns could lead to two distortions: under-investment in potentially high-return goods and over-use of unproductive goods

⁴⁸Note that we also observe that non-experts with low transfers hide (insignificantly) more. This could be due to the binary transfer measure being a noisy proxy of cutoff reputation or wives who lost their discretionary transfers being concerned about their reputation for reasons other than transfers.

(lemons). Although our results show that the behavior of both men and women responds to the reputation of women and is *consistent* with women signaling to their husbands to maintain their budget share, we cannot conclusively rule out that women are not signaling to others instead. We hope that this study opens the door to future researchers interested in firmly establishing the relationship between intra-household reputation and financial transfers and in providing evidence for the existence of what we coin the "intra-household sunk cost fallacy".

From a policy point of view, the mechanisms we propose could explain the relatively low willingness to pay for high-return investments observed in many programs and experiments aimed at women (Cohen and Dupas, 2010; Meredith et al., 2013). Campaigns promoting new technologies or goods could potentially be more successful and pose a smaller reputation risk to women if they involved both spouses or ensured that women have the means to credibly convey information about the benefits of the goods to their spouses. We leave it to future research to evaluate and test the policy implications of the novel insights our theory provides.

Finally, while this paper focused on the husband as principal and the wife as an agent given the prevailing context of gender inequality, there is no reason why the mechanism would not be completely symmetric in a context where spousal roles are reversed. When both spouses earn equal income, reputation may still matter for the share of the budget that one has control over. For example, a husband who purchased a bench press that was used only twice in the past year may face resistance when he next suggests buying a treadmill. The welfare implications of such dynamics are likely much less stark in contexts where households can afford bench presses and treadmills, however, as compared to contexts with limited and unequal consumption such as the one we consider in rural Malawi.

References

- Afzal, Uzma, Giovanna d'Adda, Marcel Fafchamps, and Farah Said, "Intrahousehold Consumption Allocation and Demand for Agency: A Triple Experimental Investigation," American Economic Journal: Applied Economics, July 2022, 14 (3), 400– 444.
- _ , _ , _ , and _ , "Intrahousehold Consumption Allocation and Demand for Agency: A Triple Experimental Investigation," American Economic Journal: Applied Economics, July 2022, 14 (3), 400–444.
- Almas, Ingvild, Alex Armand, Orazio Attanasio, and Pedro Carneiro, "Measuring and Changing Control: Women's Empowerment and Targeted Transfers," *Economic Journal*, July 2018, 128 (612), 609–639.
- Anderson, Siwan and Jean-Marie Baland, "The Economics of Roscas and Intrahousehold Resource Allocation," *The Quarterly Journal of Economics*, 2002, 117 (3).
- Andreoni, James and John Miller, "Giving according to GARP: An experimental test of the consistency of preferences for altruism," *Econometrica*, 2002, 70 (2), 737–753.

- **and Lise Vesterlund**, "Which is the fair sex? Gender differences in altruism," *The Quarterly Journal of Economics*, 2001, *116* (1), 293–312.
- Andrews, Martyn J., Jennifer Golan, and Jann Lay, "Inefficiency of Male and Female Labor Supply in Agricultural Households: Evidence from Uganda," *American Journal of Agricultural Economics*, 2015, 97 (3), 998–1019.
- **Apedo-Amah, Marie Christine, Habiba Djebbari, and Roberta Ziparo**, "Gender, Information and the Efficiency of Household Production Decisions: An Experiment in Rural Togo," IZA Discussion Papers 12948, Institute of Labor Economics (IZA) 2020.
- Arkes, Hal and Catherine Blumer, "The psychology of sunk cost," Organizational Behavior and Human Decision Processes, 1985, 35 (1), 124–140.
- Ashraf, Nava, "Spousal control and intra-household decision making: An experimental study in the Philippines," *The American Economic Review*, 2009, pp. 1245–1277.
- _, Erica M. Field, Alessandra Voena, and Roberta Ziparo, "Gendered Spheres of Learning and Household Decision Making over Fertility," Technical Report, Working Paper December 2022.
- _ , James Berry, and Jesse M Shapiro, "Can higher prices stimulate product use? Evidence from a field experiment in Zambia," *American Economic Review*, 2010, 100 (5), 2383–2413.
- Bakhtiar, Mehrab, Markus Goldstein Marcel Fafchamps, Kenneth Leonard, , and S. Papieni, "Women's Empowerment and the Intrinsic Demand for Agency: Experimental Evidence from Nigeria," MIMEO 2022.
- Banerjee, Abhijit Vinayak, Esther Duflo, Rachel Glennerster, and Dhruva Kothari, "Improving immunisation coverage in rural India: clustered randomised controlled evaluation of immunisation campaigns with and without incentives," *BMJ*, 2010, 340.
- **Bangladesh Bureau of Statistics**, "Report on violence against women (VAW) survey 2015," Technical Report 2016.
- Bar-Isaac, Heski, "Reputation and Survival: Learning in a Dynamic Signalling Model," *Review of Economic Studies*, 2003, 70 (2), 231–251.
- Becker, Gordon M, Morris H DeGroot, and Jacob Marschak, "Measuring utility by a single-response sequential method," *Behavioral science*, 1964, 9 (3), 226–232.
- Berkouwer, Susanna B and Joshua T Dean, "Credit, attention, and externalities in the adoption of energy efficient technologies by low-income households," Unpublished manuscript, University of Pennsylvania, Philadelphia, PA, 2021.
- and _ , "Credit, attention, and externalities in the adoption of energy efficient technologies by low-income households," American Economic Review, 2022, 112 (10), 3291–3330.
- Boltz, Marie, Karine Marazyan, and Paola Villar, "Income Hiding and Informal Redistribution: A Lab in the Field Experiment in Senegal," *Working Paper*, 2016.
- Breiman, Leo, "Random Forests," Machine Learning, 2001, 45 (1), 5–32.
- Cattell, Raymond B, "Theory of fluid and crystallized intelligence: A critical experiment.," Journal of educational psychology, 1963, 54 (1), 1.
- Chiappori, Pierre-André, "Rational Household Labor Supply," *Econometrica*, 1988, 56 (1), 63–89.
- _, "Collective Labor Supply and Welfare," Journal of Political Economy, June 1992, 100 (3), 437–467.
- Cohen, Jessica and Pascaline Dupas, "Free Distribution or Cost-Sharing? Evidence from a Randomized Malaria Prevention Experiment," *The Quarterly Journal of Economics*, 02

2010, 125(1), 1-45.

- Conlon, John J, Malavika Mani, Gautam Rao, Matthew Ridley, and Frank Schilbach, "Learning in the Household," 2022.
- de Laat, Joost, "Household allocations and endogenous information: The case of split migrants in Kenya," *Journal of Development Economics*, 2014, 106, 108–117.
- Delavande, Adeline, Xavier Giné, and David McKenzie, "Measuring subjective expectations in developing countries: A critical review and new evidence," *Journal of development economics*, 2011, 94 (2), 151–163.
- **Dercon, Stephan and Pramila Krishnan**, "In Sickness and in Health: Risk Sharing Within Households in Rural Ethiopia," *Journal of Political Economy*, 2000, *108*, 688–727.
- **Dewatripont, Mathias, Ian Jewitt, and Jean Tirole**, "The economics of career concerns, part I: Comparing information structures," *The Review of Economic Studies*, 1999, *66* (1), 183–198.
- Doepke, Matthias and Michèle Tertilt, "Families in macroeconomics," in "Handbook of macroeconomics," Vol. 2, Elsevier, 2016, pp. 1789–1891.
- **Duflo, Esther**, "Grandmothers and Granddaughters: Old-Age Pensions and Intrahousehold Allocation in South Africa," *The World Bank Economic Review*, 2003, 17 (1), 1–25.
- **Dupas, Pascaline**, "What matters (and what does not) in households' decision to invest in malaria prevention?," *American Economic Review*, 2009, 99 (2), 224–30.
- and Edward Miguel, "Impacts and determinants of health levels in low-income countries," in "Handbook of economic field experiments," Vol. 2, Elsevier, 2017, pp. 3– 93.
- and Jonathan Robinson, "Why don't the poor save more? Evidence from health savings experiments," American Economic Review, 2013, 103 (4), 1138–1171.
- _, Basimenye Nhlema, Zachary Wagner, Aaron Wolf, and Emily Wroe, "Expanding Access to Clean Water for the Rural Poor: Experimental Evidence from Malawi," American Economic Journal: Economic Policy, February 2023, 15 (1), 272–305.
- Fehr, Dietmar, Johanna Mollerstrom, and Ricardo Perez-Truglia, "Listen to Her: Gender Differences in Information Diffusion within the Household," Working Paper 30513, National Bureau of Economic Research September 2022.
- Fu, Chao and Juan Pantano, "Parental Reputation," Working Paper, 2015.
- Guirkinger, Catherine, Jean-Philippe Platteau, and Tatiana Goetghebuer, "Productive inefficiency in extended agricultural households: Evidence from Mali," *Journal* of Development Economics, 2015, 116 (C), 17–27.
- Hao, Lingxin, V Joseph Hotz, and Ginger Z Jin, "Games parents and adolescents play: Risky behaviour, parental reputation and strategic transfers," *The Economic Journal*, 2008, 118 (528), 515–555.
- Hoel, Jessica B, "Heterogeneous households: A within-subject test of asymmetric information between spouses in Kenya," Journal of economic behavior & organization, 2015, 118, 123–135.
- Hoel, Jessica B., "Heterogeneous households: A within-subject test of asymmetric information between spouses in Kenya," *Journal of Economic Behavior & Organization*, 2015, 118 (C), 123–135.
- Hotz, V Joseph and Juan Pantano, "Strategic parenting, birth order, and school performance," Journal of population economics, 2015, 28 (4), 911–936.
- Hussam, Reshmaan, Abu S Shonchoy, Chikako Yamauchi, and Kailash Pandey, "Translating Information Into Action: A Public Health Experiment In Bangladesh,"

Technical Report 2021.

- Iversen, Vegard, Cecile Jackson, Bereket Kebede, Alistair Munro, and Arjan Verschoor, "Do Spouses Realise Cooperative Gains? Experimental Evidence from Rural Uganda," World Development, 2011, 39 (4), 569–578.
- Jakiela, Pamela, "Equity vs. efficiency vs. self-interest: on the use of dictator games to measure distributional preferences," *Experimental Economics*, 2013, 16 (2), 208–221.
- **and Owen Ozier**, "Does Africa Need a Rotten Kin Theorem? Experimental Evidence from Village Economies," *The Review of Economic Studies*, 09 2015, *83* (1), 231–268.
- Jayachandran, Seema, "The Roots of Gender Inequality in Developing Countries," Annual Review of Economics, 2015, 7 (1), 63–88.
- Kanodia, C, R Bushman, and John Dickhaut, "Escalation errors and the sunk cost effect an explanation based on reputation and information asymmetries," *Journal of Accounting Research*, 1989, 27 (1), 59–77.
- Kebede, Bereket, Marcela Tarazona, Alistair Munro, and Arjan Verschoor, "Intra-household Efficiency: An Experimental Study from Ethiopia," *Journal of African Economies*, 2014, 23 (1), 105–150.
- Kling, Jeffrey R, Jeffrey B Liebman, and Lawrence F Katz, "Experimental analysis of neighborhood effects," *Econometrica*, 2007, 75 (1), 83–119.
- KNBS, Ministry of Health and ICF, "Kenya Demographic and Health Survey 2014," 2017.
- Kremer, Ilan and Andrzej Skrzypacz, "Dynamic signaling and market breakdown," Journal of Economic Theory, March 2007, 133 (1), 58–82.
- Magruder, Jeremy R, "An assessment of experimental evidence on agricultural technology adoption in developing countries," Annual Review of Resource Economics, 2018, 10, 299–316.
- Mailath, George and Larry Samuelson, Repeated Games and Reputations: Long-Run Relationships 01 2007.
- Manser, M. and Martin Brown, "Marriage and Household Decision-Making: A Bargaining Analysis," *International Economic Review*, 1980, 21 (1), 31–44.
- Meredith, Jennifer, Jonathan Robinson, Sarah Walker, and Bruce Wydick, "Keeping the doctor away: Experimental evidence on investment in preventative health products," *Journal of Development Economics*, 2013, 105, 196 – 210.
- MSPAS, INE and ICF, "Encuesta Nacional de Salud Materno Infantil 2014-2015. Informe Final. Guatemala," 2017.
- Noldeke, Georg and Eric van Damme, "Signalling in a Dynamic Labour Market," *Review* of Economic Studies, 1990, 57 (1), 1–23.
- **NSO and ICF**, "Malawi Demographic and Health Survey 2015-16," Technical Report, National Statistical Office Malawi and ICF, Zomba, Malawi 2017.
- Nyqvist, Martina Bjorkman, Seema Jayachandran, and Celine Zipfel, "A Mother's Voice: Impacts of Spousal Communication Training on Child Health Investments," 2023.
- Prendergast, Canice and Lars Stole, "Impetuous Youngsters and Jaded Old-Timers: Acquiring a Reputation for Learning," *Journal of Political Economy*, 1996, 104 (6).
- Schaner, Simone, "Do Opposites Detract? Intrahousehold Preference Heterogeneity and Inefficient Strategic Savings," American Economic Journal: Applied Economics, April 2015, 7 (2), 135–74.
- Smith, Sharon G, Xinjian Zhang, Kathleen C Basile, Melissa T Merrick, Jing Wang, Marcie jo Kresnow, and Jieru Chen, "The national intimate partner and

sexual violence survey: 2015 data brief-updated release," 2018.

- Swinkels, Jeroen M., "Education Signalling with Preemptive Offers," Review of Economic Studies, 1999, 66 (4), 949–970.
- Tarozzi, Alessandro, Aprajit Mahajan, Brian Blackburn, Dan Kopf, Lakshmi Krishnan, and Joanne Yoong, "Micro-loans, insecticide-treated bednets, and malaria: evidence from a randomized controlled trial in Orissa, India," *American Economic Review*, 2014, 104 (7), 1909–1941.
- Tauchen, Helen V, Ann Dryden Witte, and Sharon K Long, "Domestic violence: A nonrandom affair," *International Economic Review*, 1991, pp. 491–511.
- **Thomas, Duncan**, "Intra-household resource allocation: An inferential approach," *Journal* of human resources, 1990, pp. 635–664.
- _ , "The Distribution of Income and Expenditure within the Household," Annales d'Économie et de Statistique, 1993, (29), 109–135.
- Udry, Christopher, "Gender, Agricultural Production, and the Theory of the Household," Journal of Political Economy, 1996, 104 (5), 1010–1046.

Appendix Figures and Tables



Figure A.1: Market experiment: Willingness to pay, by wife's random-forest predicted expertise

Notes: The graph shows adjusted means from OLS regressions with Huber-White robust SEs. The dependent variable is the willingness to pay in Malawian Kwacha. Non-Expert is an indicator that takes the value 1 if the random-forest predicted perceived score on the quality quiz is at most 4.8 (the median) out of 6. Any Sticker is an indicator that takes the value 1 if either the donated or effectiveness sticker was attached to the good. Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{NE} , control mean+ $\beta_{AnySticker}$, and control mean+ $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$. We show 95% confidence intervals based on the estimated standard errors of β_{NE} , $\beta_{AnySticker}$, and $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$, respectively. Significance from testing equal willingness to pay of expert and non-expert wives in control ($\beta_{NE} = 0$) or in the sticker treatments ($\beta_{NE}+\beta_{NE\times AnySticker} = 0$). $p < 0.10^*, p < 0.05^{**}, p < 0.01^{***}$.

Figure A.2: Market experiment: Willingness to pay, by wife's random-forest predicted expertise and hiding cost



Notes: The graph shows adjusted means from OLS regressions with Huber-White robust SEs. Regressions are run separately for bags (left Panel) and books (right Panel). The dependent variable is the willingness to pay in Malawian Kwacha. Non-Expert is an indicator that takes the value 1 if the random-forest predicted perceived score on the quality quiz is at most 4.8 (the median) out of 6. Any Sticker is an indicator that takes the value 1 if either the donated or effectiveness sticker was attached to the good. Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{NE} , control mean+ $\beta_{AnySticker}$, and control mean+ $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$. We show 95% confidence intervals based on the estimated standard errors of β_{NE} , $\beta_{AnySticker}$, and $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$, respectively. Significance from testing equal willingness to pay of expert and non-expert wives in control ($\beta_{NE} = 0$) or in the sticker treatments ($\beta_{NE}+\beta_{NE\times AnySticker} = 0$). $p < 0.10^*, p < 0.05^{**}, p < 0.01^{***}$. The p-values on the interactions "Any Sticker × Book" and "Non-Expert × Any Sticker × Book" in the fully interacted model are 0.49 and 0.34, respectively.

Figure A.3: Market experiment: Willingness to pay, by wife's random-forest predicted expertise and relationship salience



Notes: The graph shows adjusted means from OLS regressions with Huber-White robust SEs. Regressions are run separately in the control treatment (left Panel) and the relationship salience treatment (right Panel). The dependent variable is the willingness to pay in Malawian Kwacha. Non-Expert is an indicator that takes the value 1 if the random-forest predicted perceived score on the quality quiz is at most 4.8 (the median) out of 6. Any Sticker is an indicator that takes the value 1 if either the donated or effectiveness sticker was attached to the good. Each bar is the sum of the control mean and the relevant regression coefficients, i.e., control mean, control mean+ β_{NE} , control mean+ $\beta_{AnySticker}$, and control mean+ $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$. We show 95% confidence intervals based on the estimated standard errors of β_{NE} , $\beta_{AnySticker}$, and $\beta_{NE}+\beta_{AnySticker}+\beta_{NE\times AnySticker}$, respectively. Significance from testing equal willingness to pay of expert and non-expert wives in control ($\beta_{NE} = 0$) or in the sticker treatments ($\beta_{NE}+\beta_{NE\times AnySticker} = 0$). $p < 0.10^*$, $p < 0.05^{**}$, $p < 0.01^{***}$. The p-values on the interactions "Any Sticker × Salience" and "Non-Expert × Any Sticker × Salience" in the fully interacted model are 0.89 and 0.77, respectively.

	Avg. transf	fers in the la	ast two montl	ns (MWK)	Access to cash $(\%)$
	(1)	(2)	(3)	(4)	(5)
MER=2	11.357		10.855		
	(1910.735)		(1911.101)		
MER=3	844.885		840.298		
	(1817.044)		(1819.637)		
MER=4	1310.761		1301.580		
	(1801.829)		(1807.818)		
Low MER		-1085.503		-1075.501	-8.993
		(843.682)		(849.411)	(3.980)
Low GAR			-58.394	-87.695	-7.167
			(669.120)	(666.459)	(2.979)
Control Mean	8186.59	8509.49	8509.49	8186.59	70.31
Observations	1093	1093	1093	1093	1092

Table A.1: Correlations between reputation and transfers from the husband to the wife in the previous two months (MWK) as well as the share of wives who have access to cash and savings

Notes: The table shows results from OLS regressions with Huber-White robust SEs. The data is winsorized at 3 SDs (1.7% of the data). MER=2, MER=3, and MER=4 are binary variables that take the value 1 if the woman has an MER of 2 (13%), 3 (31%), or 4 (52%) respectively. Access to cash is an indicator that takes the value 1 if the husband reports that his wife has "access to cash and savings". The regressions control for the wife and the husband's age, education, average income in the last two months (as reported by the husband), variability of income (whether income is the same in most months or varies a lot), risk preferences, math, and raven scores, as well as years married, number of children and household members and enumerator fixed effects.

	(1)	(2)	(3)	(4)	(5)	(6)
Low MER	0.343	-0.472	0.492	0.173	0.243	0.445
	(3.374)	(3.395)	(3.339)	(3.389)	(3.381)	(3.359)
Salience	2.143	1.635	2.128	3.237	3.117	0.876
	(1.720)	(1.741)	(2.224)	(2.298)	(1.965)	(2.171)
Low MER*Salience	-9.184	-9.311	-8.786	-8.917	-8.985	-9.286
	(4.231)	(4.181)	(4.226)	(4.241)	(4.220)	(4.222)
Low GAR	· /	. ,	-3.763	. ,	. ,	· /
			(2.309)			
Low GAR*Salience			0.030			
			(2.993)			
Low Husband GAR				-2.566		
				(2.310)		
Low Husband GAR*Salience				-2.313		
				(2.991)		
No Wife Income					-0.085	
					(2.680)	
No Wife Income*Salience					-3.322	
					(3.389)	
Few Children						-1.835
						(2.331)
Few Children*Salience						2.486
						(3.026)
Control Mean	68.889	68.889	70.608	71.285	68.823	68.946
P-value (Expert vs. Non-Expert, Control)	0.919	0.890	0.883	0.959	0.943	0.895
P-value (Expert vs. Non-Expert, Salience)	0.001	0.000	0.002	0.001	0.001	0.001
Observations	1093	1093	1093	1093	1093	1093
Benchmark specification from Figure 4	\checkmark					
Including controls		\checkmark				
Testing for experimenter demand effect			\checkmark			
Testing for effect of the husband's mood				\checkmark		
Testing for effect of wife's income					\checkmark	
Testing for effect of children						\checkmark

Table A.2: Transfer experiment: Effect of reputation salience on amount (%) transferred from the husband to the wife

Notes: The table shows results from OLS regressions with Huber-White robust SEs. All regressions include enumerator, compensation and version fixed effects. Market Expertise Reputation (MER) defined as before. Column 2 tests for robustness when including additional controls (see Table A.1 notes for the list of controls). Column 3 tests for experimenter demand effects by assessing the impact of the salience treatment by the wife's General Ability Reputation (GAR). GAR is the normalized mean of the husband's beliefs about the wife's correct answers in a math test, and a raven game. Low GAR is a binary variable that takes the value 1 if the woman has a General Ability Reputation below the median, and 0 otherwise. Column 4 tests whether the salience treatment works through making husbands angrier by assessing the impact of the salience treatment by the function (GAR). This is the normalized mean of the husband's General Ability Reputation (GAR). This is the normalized mean of the husband's General Ability Reputation (GAR). This is the normalized mean of the husband's beliefs about his correct answers in a math test and a raven game. Low Husband GAR is a binary variable that takes the value 1 if the husband has a general Ability reputation below the median, and 0 otherwise. Columns 5 and 6 test whether the salience treatment works through other characteristics of the household/match, specifically, whether the wife has an income or the couple has fewer than three children.

	Whole sample (N=1093)			Participation sample (N=786)				Whole sample (N=1093)
			Panel A:	By price				
	Initial score	Participate (%)	Forgone comp.	Initial score	# Errors corrected	Hiding fee	Final score	Total forgone
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Non-Expert	-0.246	-4.080	8.160	-0.361	0.247	23.297 (11.043)	-0.114	24.540 (10.774)
Intermediate Cost	0.168	4.578	(9.011) -9.156 (7.747)	(0.117) 0.237 (0.120)	-0.179	10.223	(0.139) 0.059 (0.130)	1.413
High Cost	(0.118) 0.113	(3.874) -0.348	(7.747) 0.695 (7.795)	(0.129) 0.060 (0.125)	(0.074) -0.238	(10.973) 17.802 (12.500)	(0.130) -0.178	(10.750) 13.976
Non-Expert*Intermediate Cost	(0.108) -0.000	(3.897) -12.476	(7.795) 24.952	(0.125) -0.043	(0.068) -0.028	(12.590) 21.570	(0.128) -0.071	(11.155) 27.280
Non-Expert*High Cost	(0.184) 0.119 (0.176)	(6.840) -11.188	(13.681) 22.375 (12.176)	(0.207) 0.214 (0.208)	(0.140) -0.243 (0.126)	(21.012) -16.702 (22.260)	(0.216) -0.029 (0.218)	(17.511) 0.128 (17.445)
Mean (Low Cost & Expert) P-value (Expert vs. Non-Expert, Intermediate Cost) P-value (Expert vs. Non-Expert, High Cost)	$\begin{array}{c} (0.170) \\ 4.149 \\ 0.060 \\ 0.293 \end{array}$	$\begin{array}{c} (0.388) \\ 76.349 \\ 0.001 \\ 0.001 \end{array}$	$\begin{array}{c} (13.176) \\ 47.303 \\ 0.001 \\ 0.001 \end{array}$	$\begin{array}{c} (0.208) \\ 4.196 \\ 0.005 \\ 0.320 \end{array}$	$\begin{array}{c} (0.120) \\ 0.462 \\ 0.015 \\ 0.957 \end{array}$	$\begin{array}{c} (22.209) \\ 46.196 \\ 0.012 \\ 0.734 \end{array}$	$\begin{array}{c} (0.218) \\ 4.658 \\ 0.205 \\ 0.338 \end{array}$	82.573 0.000 0.074

Table A.3: Outcomes in the signaling experiment

			Panel B:	By price a	and difficult	y of the qu	ıiz	
	Initial score	Participate (%)	Forgone comp.	Initial score	# Errors corrected	Hiding fee	Final score	Total forgone
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Harder Version	-1.120	6.836	-13.672	-1.065	0.305	30.657	-0.760	11.536
	(0.112)	(4.433)	(8.866)	(0.129)	(0.100)	(10.257)	(0.144)	(10.379)
Intermediate Cost	0.116	5.113	-10.225	0.111	-0.195	0.953	-0.084	-7.833
	(0.104)	(4.529)	(9.059)	(0.121)	(0.069)	(9.692)	(0.124)	(10.501)
High Cost	0.102	-2.289	4.579	0.148	-0.219	9.460	-0.072	8.528
	(0.105)	(4.584)	(9.168)	(0.122)	(0.071)	(13.026)	(0.126)	(11.565)
Harder Version [*] Intermediate Cost	0.082	-11.762	23.524	0.149	0.039	38.063	0.188	43.937
	(0.163)	(6.516)	(13.031)	(0.186)	(0.128)	(19.217)	(0.201)	(17.133)
Harder Version [*] High Cost	0.152	-6.313	12.626	-0.002	-0.204	4.131	-0.206	13.044
	(0.154)	(6.351)	(12.702)	(0.178)	(0.119)	(20.887)	(0.192)	(17.261)
Mean (Low Cost & Easier Version)	4.594	71.875	56.250	4.601	0.406	40.580	5.007	85.417
P-value (Easier vs. Harder Version, Intermediate Cost)	0.000	0.303	0.303	0.000	0.000	0.000	0.000	0.000
P-value (Easier vs. Harder Version, High Cost)	0.000	0.909	0.909	0.000	0.107	0.054	0.000	0.074

Notes: The table shows results from OLS regressions with Huber-White robust SEs. Non-Expert is an indicator that takes the value 1 if the wife reports an average weighted score that is lower than 5 (39% of women). The weighted average is calculated as the average across all scores, weighted by the probability assigned to each score by the woman (each woman placed 10 beans on the 7 different scores). Forgone comp. is the amount of money wives left on the table by opting out of the game. All regressions include enumerator and compensation fixed effects. The p-value is the p-value from a Wald test comparing outcomes between Non-Expert and Expert wives or between the hard and the easy version when the hiding cost is high.

Table A.4: Market experiment: Willingness to pay, by wife's expertise

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Non-Expert	-89.782	-86.941	-78.570	-89.219	-86.257	-77.216			
-	(36.868)	(36.999)	(36.288)	(36.769)	(36.875)	(36.149)			
'Donated' Sticker	-32.684	-37.235	-38.032						
	(37.855)	(38.007)	(37.849)						
'Donated'*Non-Expert	113.829	107.463	115.436						
	(55.173)	(55.354)	(54.859)						
'Effectiveness' Sticker	-31.535	-32.110	-39.847						
	(40.388)	(39.376)	(38.914)						
'Effectiveness'*Non-Expert	121.058	120.382	136.007						
	(59.744)	(59.461)	(58.813)						
'Donated'&'Effectiveness' Stickers	-45.036	-49.979	-45.003						
	(39.776)	(38.544)	(38.505)						
('Donated'&'Effectiveness')*Non-Expert	48.426	52.750	59.247						
	(53.765)	(53.607)	(52.779)						
Any Sticker				-35.841	-39.075	-40.023	0.228	-3.843	0.538
				(31.694)	(30.828)	(30.643)	(22.955)	(22.591)	(22.134)
Any Sticker*Non-Expert				94.123	92.504	102.193			
				(44.053)	(44.298)	(43.178)			
Mean(Control & Expert)	350.802	350.802	350.802	350.802	350.802	350.802	350.802	350.802	350.802
P-value (Expert vs. Non-Expert, Control)	0.015	0.019	0.031	0.016	0.020	0.033			
P-value (Expert vs. Non-Expert, Donated)	0.559	0.613	0.379						
P-value (Expert vs. Non-Expert, Effective)	0.507	0.470	0.217						
P-value (Expert vs. Non-Expert, D+E)	0.926	0.940	0.693						
P-value (Expert vs. Non-Expert, Any Sticker)				0.841	0.796	0.307			
Observations	675	675	675	675	675	675	675	675	675
Market FE									
Controls		•	• •		•	•		•	• •

Notes: The table shows results from OLS regressions with Huber-White robust SEs. The dependent variable is the willingness to pay in Malawian Kwacha. Non-Expert is an indicator that takes the value 1 if the wife reports that her husband has a prior about her math score below 5 (44%). All regressions include enumerator fixed effects. Market fixed effects are dummies for the different markets in which the enumerators recruited married women. Controls include the wife's age, education, average income in the last and previous month, risk preferences, math score, as well as the husband's average income in the last and previous month, years married, and the number of children and household members.

Online Appendix

Table of Contents

\mathbf{A}	The	eory Appendix	1
	A.1	Strategies	1
	A.2	Proof of lemma 1	1
	A.3	Proof of proposition 1	2
	A.4	Proof of proposition 2	6
	A.5	Proof of Proposition 3	10
в	Tec	hnical Appendix	14
	B.1	Sampling	14
	B.2	Compensation and Prices	14
	B.3	Defining the Salience Treatment in the Transfer Experiment	14
С	\mathbf{Em}	pirical Appendix	15
	C.1	Figures	15
	C.2	Tables	21

A Theory Appendix

A.1 Strategies

Before describing strategies, we introduce the notation for information sets, at which the husband and the wife make choices. Denote $h_t^H \in H_t^H$ the information set of the husband in period t. At t = 1, the husband has only one information set. At t = 2, h_t^H is characterized by the husband's observations — the wife's purchase and usage choices at t = 1. Denote $h_{t,g}^{NE} \in H_{t,g}^{NE}$ the non-expert wife's information set when she makes an investment choice and $h_{t,e}^{NE} \in H_{t,e}^{NE}$ when she makes a usage choice in period t. At t = 1, the non-expert wife has only one information set when making the investment choice; when making the usage choice, her information set is described by the productivity of the purchased good. At t = 2, the non-expert wife's information sets also depend on the history she observes. Similarly, denote $h_{t,g}^E \in H_{t,g}^E$ the expert wife's information set when she makes an investment choice and $h_{t,e}^E \in H_{t,e}^E$ when she makes a usage choice in period t. In contrast to the non-expert wife, the expert wife's information set when making the investment choice depends on the productivity of the risky good, which she observes.

Denote the agent *i*'s strategy σ^i . For convenience, we also introduce notation for behavior strategies at each information set, i.e., $\sigma_t^H : H_t^H \to [0, 1], \sigma_{t,g}^E : H_{t,g}^E \to [0, 1], \sigma_{t,e}^E : H_{t,e}^E \to [0, 1],$ $\sigma_{t,g}^{NE} : H_{t,g}^{NE} \to [0, 1], \sigma_{t,e}^{NE} : H_{t,e}^{NE} \to [0, 1]$ map histories to the probability of an action (making a transfer, buying the risky good, or using the purchased good).

Let $P_2(h_2^H)$ be the wife's reputation at the information set h_2^H . Since the wife observes more information than the husband, she also knows $P_2(h_2^H)$ in the second period. To simplify notation, we just write P_2 for the wife's reputation at t = 2.

We focus on a Perfect Bayesian Equilibrium, which requires sequential rationality and the beliefs to be determined by Bayes' rule whenever possible. The Bayes' rule is the following:

$$P_2(P_1, g_1 = 1, e_1 = 1) = \frac{P_1 \sigma_{1,g}^E(h_{1,g}^E) \sigma_{1,e}^E(h_{1,e}^E)}{P_1 \sigma_{1,g}^E(h_{1,g}^E) \sigma_{1,e}^E(h_{1,e}^E) + (1 - P_1) \sigma_{1,g}^{NE}(h_{1,g}^{NE}) \sigma_{1,e}^{NE}(h_{1,e}^{NE})}$$

A.2 Proof of lemma 1

 $\begin{array}{l} Proof. \mbox{ Consider }t=2. \mbox{ This is the last period, so everyone plays their static best response.}\\ \mbox{For the wife, the investment strategies are } \sigma^{NE}_{2,g}(h^{NE}_{2,g}) &= 1 \mbox{ for any } h^{NE}_{2,g}, \mbox{ and } \sigma^{E}_{2,g}(h^{E}_{2,g}) &= \begin{cases} 1 \mbox{ if } \eta^{R}_{2} = \eta^{R} \\ 0 \mbox{ if } \eta^{R}_{2} = 0 \end{cases} & . \mbox{ The usage strategies are } \sigma^{E}_{2,e}(h^{E}_{2,e}) &= \begin{cases} 1 \mbox{ if } \eta_{2} \neq 0 \\ 0 \mbox{ if } \eta_{2} = 0 \end{cases},\\ \sigma^{NE}_{2,e}(h^{NE}_{2,e}) &= \begin{cases} 1 \mbox{ if } \eta_{2} \neq 0 \\ 0 \mbox{ if } \eta_{2} = 0 \end{cases}. \mbox{ The husband's payoff is increasing in } P_{2}: \ \lambda \eta^{R} + P_{2}(1-\lambda)\eta^{S}. \end{cases}$

Therefore, the husband's best response is $\sigma_2^H(h_2^H) = \begin{cases} 1 \text{ if } P_2 > P_2^* \\ [0,1] \text{ if } P_2 = P_2^* \\ 0 \text{ if } P_2 < P_2^* \end{cases}$, where P_2^* is

defined by
$$\lambda \eta^R + P_2(1-\lambda)\eta^S = \omega$$
, i.e., $P_2^* = \frac{\omega - \lambda \eta^R}{(1-\lambda)\eta^S}$.

A.3 Proof of proposition 1

Proof. Suppose the hiding cost is sufficiently high: $c > \beta \lambda \eta^R$. Lemma 1 pins down the equilibrium strategies at t = 2, except for the husband's strategy when he is indifferent between making the transfer or not, i.e., at h_2^H such that $P_2 = P_2^*$. For these cases, let the husband randomize with probabilities $\tilde{\sigma}_2^H(h_2^H) \in [0, 1]$, which are defined further in the proof for various histories. At t = 1, equilibrium strategies are the following. The expert wife invests iff the risky good is productive and always uses the good unless it is unproductive and the hiding cost is too high:

$$\sigma_{1,g}^{E}(h_{1,g}^{E}) = \begin{cases} 1 \text{ if } \eta_{1}^{R} \neq 0\\ 0 \text{ if } \eta_{1}^{R} = 0 \end{cases}; \\ \sigma_{1,e}^{E}(h_{1,e}^{E}) = \begin{cases} 1 \text{ if } \eta_{1} \neq 0 \text{ or } c < \beta(\lambda\eta^{R} + (1-\lambda)\eta^{S})\\ 0 \text{ if } \eta_{1} = 0 \text{ and } c \ge \beta(\lambda\eta^{R} + (1-\lambda)\eta^{S}) \end{cases}$$

The non-expert wife invests with a probability that is decreasing in reputation, always uses productive and safe goods, and never uses unproductive goods.

$$\sigma_{1,g}^{NE}(h_{1,g}^{NE}) = \begin{cases} 0 & \text{if } P_1 > \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*} \\ \frac{P_2^* - P_1 + \lambda P_1(1-P_2^*)}{(1-P_1)P_2^*} \le \lambda & \text{if } P_1 \in \left[P_2^*, \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*}\right]; \\ \frac{P_2^* - P_1 + \lambda P_1(1-P_2^*)}{(1-P_1)P_2^*} > \lambda & \text{if } P_1 < P_2^* \\ \frac{\sigma_{1,e}^{NE}(h_{1,e}^{NE}) = \begin{cases} 1 & \text{if } \eta_1 \neq 0 \\ 0 & \text{if } \eta_1 = 0 \end{cases} \end{cases}$$

Note, that the husband makes a transfer at t = 1 if the expected payoff is higher than the outside option. Let $V_t(P_t)$ be the expected value of the husband at time t if he has a belief P_t . Note that

$$V_2(P_2) = \begin{cases} \omega & \text{if } P_2 < P_2^* \\ \lambda \eta^R + P_2(1-\lambda)\eta^S & \text{if } P_2 \ge P_2^* \end{cases}$$

Using productive and safe goods is weakly dominant for both wife types. When the good is unproductive, $e_1 = 0$ is optimal for the non-expert wife because the hiding cost is greater than the highest continuation payoff: $c > \beta \lambda \eta^R$. For the expert wife, it is sometimes optimal to use the unproductive good (off-path) if the continuation payoff is larger than the cost, i.e., if $c < \beta (\lambda \eta^R + (1 - \lambda) \eta^S)$. Next, consider the purchase decision.

We will show that the strategies of the wife and the husband form an equilibrium for different values of P_1 for which the wife has different investment strategies: $P_1 > \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*}$ and $P_1 \le \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*}$.

For all values of P_1 , we:

- 1. calculate the husband's Bayesian on-equilibrium and off-equilibrium posteriors $P_2(P_1|g_1, e_1)$ for $g_1 \in \{0, 1\}$ and $e_1 \in \{0, 1\}$ given the wife's strategies,
- 2. show that the wife has no profitable deviation in her usage choice and calculate the husband's transfer strategy in period 2 that makes the wife indifferent between different usage choices in case she has a mixed strategy, and

- 3. show that the wife has no profitable deviation in her investment choice and calculate the husband's transfer strategy in period 2 that makes the wife indifferent between different investment choices in case she has a mixed strategy,
- 4. calculate the husband's transfer strategy in period 1.

First, suppose
$$P_1 > \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*}$$
.

- Given the strategies, the updated reputation is $P_2 > P_2^*$.
- For the investment choice, it is optimal for the non-expert wife to buy the safe good because $\eta^S + \beta \lambda \eta^R \ge \lambda \eta^R + \beta \lambda^2 \eta^R$. This condition holds by assumption that $\beta \lambda \eta^R (1 \lambda) \ge \lambda \eta^R \eta^S$ (assumption 1). For the expert wife, it is optimal to buy the risky good if $\eta_1^R = \eta^R$ because $\eta^R + \beta(\lambda \eta^R + (1 \lambda)\eta^S) > \eta^S + \beta(\lambda \eta^R + (1 \lambda)\eta^S)$. For the expert wife, it is optimal to buy the safe good if $\eta_1^R = 0$ because $\eta^S + \beta(\lambda \eta^R + (1 \lambda)\eta^S) > \max\{0, -c + \beta(\lambda \eta^R + (1 \lambda)\eta^S)\}$.
- For the husband's strategy, any posterior P_2 lies above P_2^* . In this region, $V_2(P_2)$ is linear in P_2 , so $\mathbb{E}V_2(P_2) = \lambda \eta^R + \mathbb{E}P_2(1-\lambda)\eta^S = \lambda \eta^R + P_1(1-\lambda)\eta^S = V_2(P_1)$. Thus, the husband needs to only compare first-stage payoffs from $T_1 = 1$ and $T_1 = 0$. The first-stage payoff is increasing in P_1 :

$$\mathbb{E}[U_1^H(T_1=1)|P_1] = P_1(\lambda\eta^R + (1-\lambda)\eta^S) + (1-P_1)\eta^S$$

Moreover, at the lower end of the interval, at $P_1 = \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*}$, the husband prefers to make the transfer:

$$\mathbb{E}\left[U_{1}^{H}(T_{1}=1)|P_{1}=\frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+P_{2}^{*}}\right]-\omega \\ =\frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+P_{2}^{*}}(\lambda\eta^{R}+(1-\lambda)\eta^{S})+\frac{(1-\lambda)(1-P_{2}^{*})}{(1-\lambda)(1-P_{2}^{*})+P_{2}^{*}}\eta^{S}-\omega \\ \propto P_{2}^{*}(\lambda\eta^{R}+(1-\lambda)\eta^{S})+(1-\lambda)(1-P_{2}^{*})\eta^{S}-\omega+\omega\lambda(1-P_{2}^{*}) \\ =P_{2}^{*}(\lambda\eta^{R}+(1-\lambda)\eta^{S})+(1-P_{2}^{*})\lambda\eta^{R}+(1-P_{2}^{*})(1-\lambda)(\eta^{S}-\lambda\eta^{R})-\omega+\omega\lambda(1-P_{2}^{*}) \\ =(1-P_{2}^{*})(\lambda(\omega-\lambda\eta^{R})-(1-\lambda)(\lambda\eta^{R}-\eta^{S}))\geq 0 \quad (\text{by assumption 1}).$$

Thus, $T_1 = 1$ in this interval.

Second, suppose
$$P_1 \le \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*}$$

- Given the strategies, $P_2(P_1|g_1 = 0, e_1 = 1) = P_2^*$, $P_2(P_1 \ge \frac{P_2^*}{(1 P_2^*)(1 \lambda) + 1} \equiv \overline{P_1}|g_1 = 1, e_1 = 1) \ge P_2^*$ and $P_2(P_1 < \overline{P_1}|g_1 = 1, e_1 = 1) < P_2^*$. Denote $\kappa_1^S \equiv \sigma_2^H(h_2^H)$ when $P_1 \ge \overline{P_1}$ and $g_1 = 0$. Denote $\kappa_2^S \equiv \sigma_2^H(h_2^H)$ when $P_1 < \overline{P_1}$ and $g_1 = 0$.
- For the investment choice, the non-expert wife mixes if $\eta^S + \beta \kappa_1^S \lambda \eta^R = \lambda \eta^R + \beta \lambda^2 \eta^R$ and $\eta^S + \beta \kappa_2^S \lambda \eta^R = \lambda \eta^R$. These conditions pin down the husband's mixing probabilities: $\kappa_1^S = \frac{\lambda \eta^R - \eta^S + \beta \lambda^2 \eta^R}{\beta \lambda \eta^R}$ and $\kappa_2^S = \frac{\lambda \eta^R - \eta^S}{\beta \lambda \eta^R}$. For the expert wife, it is optimal to

buy the risky good if $\eta_1^R = \eta^R$ because $\eta^R + \beta(\lambda\eta^R + (1-\lambda)\eta^S) > \eta^S + \beta\kappa_1^S(\lambda\eta^R + (1-\lambda)\eta^S)$ and $\eta^R > \eta^S + \beta\kappa_2^S(\lambda\eta^R + (1-\lambda)\eta^S)$. The only strategies that form an equilibrium are the husband randomizing for $g_1 = 0$ and transferring with probability 1 for $g_1 = 1$, as otherwise when $P_1 = \bar{P}_1$, and $P_2 = P_2^*$ and the non-expert wife's investment rate is $> \lambda$, the non-expert wife would have the incentive to deviate by decreasing her investment rate. For the expert wife, it is optimal to buy the safe good if $\eta_1^R = 0$ because $\eta^S + \beta\kappa_1^S(\lambda\eta^R + (1-\lambda)\eta^S) > \max\{0, -c + \beta(\lambda\eta^R + (1-\lambda)\eta^S)\}$ and $\eta^S + \beta\kappa_2^S(\lambda\eta^R + (1-\lambda)\eta^S) > 0$.

- For the husband's strategy, we will look at three regions separately: $P_1 \in \left[P_2^*, \frac{P_2^*}{(1-\lambda)(1-P_2^*)+P_2^*}\right], P_1 \in (\overline{P}_1, P_2^*), \text{ and } P_1 \leq \overline{P}_1:$
 - First, suppose $P_1 \in \left[P_2^*, \frac{P_2^*}{(1-\lambda)(1-P_2^*)+P_2^*}\right]$. We look at the difference between the sum of the first-stage and continuation payoffs under the transfer and no transfer. The expected first-stage payoff if $T_1 = 1$ is

$$\mathbb{E}[U_1^H(T_1 = 1)|P_1] = P_1(\lambda \eta^R + (1 - \lambda)\eta^S) + (1 - P_1)(\sigma_{1,g}^{NE}\lambda \eta^R + (1 - \sigma_{1,g}^{NE})\eta^S)$$

= ... = $\lambda \eta^R + P_1 \frac{1 - \lambda}{P_2^*} (\eta^S - (1 - P_2^*)\lambda \eta^R)$
 $\geq \lambda \eta^R + (1 - \lambda)[\eta^S - (1 - P_2^*)\lambda \eta^R]$

The difference in expected first-stage payoffs under $T_1 = 1$ and $T_1 = 0$ is

$$\Delta \mathbb{E} U_1^H = \lambda \eta^R + P_1 \frac{1-\lambda}{P_2^*} (\eta^S - (1-P_2^*)\lambda \eta^R) - \omega$$

which is increasing in P_1 . The expected (discounted) second-stage continuation payoff if $T_1 = 1$ is

$$\beta \mathbb{E}[U_2^H(T_1 = 1)|P_1] = \beta [V_2(P_2(P_1|g = 1, e = 1))Pr(g = 1, e = 1) + \omega Pr(g = 0) + \omega Pr(g = 1, e = 0)]$$

The expected (discounted) continuation payoff if $T_1 = 0$ can also be written in a similar way using the linearity of payoff and $\mathbb{E}P_2 = P_1$:

$$\beta \mathbb{E}[U_2^H(T_1=0)|P_1] = \beta V_2(P_1)$$

= $\beta [V_2(P_2(P_1|g=1,e=1))Pr(g=1,e=1) + \omega Pr(g=0) + \lambda \eta^R Pr(g=1,e=0)]$

The difference in the expected continuation payoffs under $T_1 = 1$ and $T_1 = 0$ is

$$\beta \Delta \mathbb{E} U_2^H = \beta Pr(g = 1, e = 0)(\omega - \lambda \eta^R)$$
$$= \beta \frac{P_2^* - P_1 + \lambda P_1(1 - P_2^*)}{P_2^*} (1 - \lambda)(\omega - \lambda \eta^R)$$

which is decreasing in P_1 . The husband chooses $T_1 = 1$ iff $\Delta \mathbb{E} U_1^H + \beta \Delta \mathbb{E} U_2^H \ge 0$.

First, we show that $\Delta \mathbb{E} U_1^H + \beta \Delta \mathbb{E} U_2^H \ge 0$ at the lower end, $P_1 = P_2^*$:

$$\begin{split} \Delta \mathbb{E} U_1^H &+ \beta \Delta \mathbb{E} U_2^H \\ &= \lambda \eta^R + (1-\lambda)(\eta^S - (1-P_2^*)\lambda \eta^R) - \omega + \beta \lambda (1-P_2^*)(1-\lambda)(\omega - \lambda \eta^R) \\ &= \lambda \eta^R + (1-\lambda)(\eta^S - (1-P_2^*)\lambda \eta^R) \\ &- (\lambda \eta^R + P_2^*(1-\lambda)\eta^S) + \beta \lambda (1-P_2^*)(1-\lambda)(\omega - \lambda \eta^R) \\ &= (1-\lambda)(1-P_2^*)(\beta \lambda (\omega - \lambda \eta^R) - (\lambda \eta^R - \eta^S)) \ge 0, \end{split}$$

where the inequality holds by assumption 1. Second, we show that $\Delta \mathbb{E} U_1^H + \beta \Delta \mathbb{E} U_2^H$ is monotonically increasing in P_1 :

$$\begin{split} &\frac{\partial}{\partial P_1} (\Delta \mathbb{E} U_1^H + \beta \Delta \mathbb{E} U_2^H) \\ &= \frac{1-\lambda}{P_2^*} (\eta^S - (1-P_2^*)\lambda\eta^R) + \beta \frac{1-\lambda}{P_2^*} (\lambda(1-P_2^*) - 1)(\omega - \lambda\eta^R) \\ &\propto \eta^S - (1-P_2^*)\lambda\eta^R + \beta P_2^*(1-\lambda)\eta^S (\lambda(1-P_2^*) - 1) \\ &= P_2^* [\lambda\eta^R - \beta\eta^S (1-\lambda)(1-\lambda(1-P_2^*))] - (\lambda\eta^R - \eta^S) \\ &= \beta\lambda(\omega - \lambda\eta^R) \left[\frac{\eta^R}{\beta(1-\lambda)\eta^S} - \frac{1-\lambda(1-P_2^*)}{\lambda} \right] - (\lambda\eta^R - \eta^S) \\ &\geq \beta\lambda(\omega - \lambda\eta^R) \left[\frac{\eta^R}{(1-\lambda)\eta^S} - \frac{1-\lambda(1-P_2^*)}{\lambda} \right] - (\lambda\eta^R - \eta^S) \\ &= \beta\lambda(\omega - \lambda\eta^R) \left[\frac{\eta^R}{(1-\lambda)\eta^S} - \frac{1-\lambda(1-P_2^*)}{\lambda} \right] - (\lambda\eta^R - \eta^S) \\ &= \beta\lambda(\omega - \lambda\eta^R) \left[1 + \frac{\lambda\eta^R + \lambda^2\eta^R - \lambda\omega - \eta^S + \lambda\eta^S}{\lambda(1-\lambda)\eta^S} \right] - (\lambda\eta^R - \eta^S) > 0, \end{split}$$

where we have used that $P_2^* = \frac{\omega - \lambda \eta^R}{(1 - \lambda) \eta^S}$ and the inequality holds under assumption 1. Thus, the husband chooses $T_1 = 1$ at all $P_1 \ge P_2^*$.

- Next, suppose $P_1 \in (\overline{P}_1, P_2^*)$. Following the same approach as above, the difference in expected first-stage payoffs under $T_1 = 1$ and $T_1 = 0$ is

$$\Delta \mathbb{E} U_1^H = \lambda \eta^R + P_1 \frac{1-\lambda}{P_2^*} (\eta^S - (1-P_2^*)\lambda \eta^R) - \omega$$

which is increasing in P_1 . The difference in the expected continuation payoffs under $T_1 = 1$ and $T_1 = 0$ is

$$\begin{split} \beta \Delta \mathbb{E} U_2^H &= \beta Pr(g = 1, e = 1)(\lambda \eta^R + P_2(P_1|g = 1, e = 1)(1 - \lambda)\eta^S - \omega) \\ &= \beta (P_1 + (1 - P_1)\sigma_{1,g}^{NE})\lambda(\lambda \eta^R + P_2(P_1|g = 1, e = 1)(1 - \lambda)\eta^S - \omega) \\ &= \beta \frac{P_1 P_2^* + P_2^* - P_1 + \lambda P_1(1 - P_2^*)}{P_2^*}\lambda \times \\ &\left(\lambda \eta^R + \frac{P_1 P_2^*}{P_1 P_2^* + P_2^* - P_1 + \lambda P_1(1 - P_2^*)}(1 - \lambda)\eta^S - \omega\right) \\ &= \beta \lambda \left[(\omega - \lambda \eta^R) \frac{(1 - \lambda)P_1(1 - P_2^*) - P_2^*}{P_2^*} + \eta^S(1 - \lambda)P_1 \right] \end{split}$$

which is also increasing in P_1 . Thus, $\Delta \mathbb{E} U_1^H + \beta \Delta \mathbb{E} U_2^H$ is monotonically increasing in P_1 in this interval. At the lower end, if $P_1 = \overline{P}_1$, then $P_2(P_1|g = 1, e = 1) = P_2^*$, so $\beta \Delta \mathbb{E} U_2^H = 0$. For the first-stage payoff, $\Delta \mathbb{E} U_1^H < 0$. Therefore, the husband chooses $T_1 = 0$ at $P_1 = \overline{P}_1$ and switches to $T_1 = 1$ at some higher P_1 .

- Finally, suppose $P_1 \leq \overline{P}_1$. On the equilibrium path, updated reputation is always below P_2^* . Then, following the same argument as in Proposition 1, the husband needs to only consider the first-stage payoff. The expected first-stage payoff is worse than the outside option:

$$\mathbb{E}[U_{1}^{H}(T_{1}=1)|P_{1}] = P_{1}(\lambda\eta^{R} + (1-\lambda)\eta^{S}) + (1-P_{1})(\sigma_{1,g}^{NE}\lambda\eta^{R} + (1-\sigma_{1,g}^{NE})\eta^{S})$$

$$\leq P_{1}(\lambda\eta^{R} + (1-\lambda)\eta^{S}) + (1-P_{1})\lambda\eta^{R}$$

$$< P_{2}^{*}(\lambda\eta^{R} + (1-\lambda)\eta^{S}) + (1-P_{2}^{*})\lambda\eta^{R}$$

$$\leq \lambda\eta^{R} + (1-\lambda)\eta^{S} = \omega$$

Thus, the husband also chooses $T_1 = 0$ at all $P_1 < \overline{P}_1$.

Combining the three intervals for the husband, we conclude that the husband uses a threshold strategy:

$$T_1(P_1) = \begin{cases} 1 \text{ if } P_1 \ge P_1^* \\ 0 \text{ if } P_1 < P_1^* \end{cases}$$

where $P_1^* \in \left(\frac{P_2^*}{(1-P_2^*)(1-\lambda)+1}, P_2^*\right).$

A.4 Proof of proposition 2

Proof. Suppose the hiding cost is sufficiently small: $c \leq \frac{\lambda \eta^R - \eta^S}{1 - \lambda}$. Lemma 1 pins down the equilibrium strategies at t = 2, except for the husband's strategy when he is indifferent between making the transfer or not, i.e., at h_2^H such that $P_2 = P_2^*$. For these cases, let the husband randomize with probabilities $\tilde{\sigma}_2^H(h_2^H) \in [0, 1]$, which are defined further in the proof for various histories. At t = 1, equilibrium strategies are the following. The expert wife invests iff the risky good is productive and always uses the good in equilibrium:

$$\sigma_{1,g}^{E}(h_{1,g}^{E}) = \begin{cases} 1 \text{ if } \eta_{1}^{R} \neq 0\\ 0 \text{ if } \eta_{1}^{R} = 0 \end{cases};$$

$$\sigma_{1,e}^{E}(h_{1,e}^{E}) = \begin{cases} 1 & \text{if } \eta_{1} \neq 0 \text{ or } \left(\eta_{1} = 0 \text{ and } P_{1} \geq \frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+1}\right)\\ 0 & \text{if } \eta_{1} = 0 \text{ and } P_{1} < \frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+1} \end{cases}$$

The non-expert wife invests with probability at least λ , always uses the productive and safe goods but uses the unproductive good with positive probability only when her reputation is not too low:

$$\sigma_{1,g}^{NE}(h_{1,g}^{NE}) = \begin{cases} 1 & \text{if } P_1 > \frac{P_2^*}{P_2^* + \lambda(1 - P_2^*)} \\ \lambda \frac{P_1}{1 - P_1} \frac{1 - P_2^*}{P_2^*} \ge \lambda & \text{if } P_1 \in \left[P_2^*, \frac{P_2^*}{P_2^* + \lambda(1 - P_2^*)}\right]; \\ \frac{P_2^* - P_1 + \lambda P_1(1 - P_2^*)}{(1 - P_1)P_2^*} \ge \lambda & \text{if } P_1 < P_2^* \\ \sigma_{1,e}^{NE}(h_{1,e}^{NE}) = \\ if \eta_1 \neq 0 \text{ or } (\eta_1 = 0 \text{ and } P_1 \ge P_2^*) \\ \left[\frac{P_1(1 - P_2^*)}{P_2^* - P_1 + \lambda P_1(1 - P_2^*)} - 1\right] \frac{\lambda}{1 - \lambda} & \text{if } \eta_1 = 0 \text{ and } P_1 \in \left[\frac{P_2^*}{(1 - \lambda)(1 - P_2^*) + 1}, P_2^*\right] \\ 0 & \text{if } \eta_1 = 0 \text{ and } P_1 < \frac{P_2^*}{(1 - \lambda)(1 - P_2^*) + 1} \end{cases}$$

We will show that the strategies of the wife and the husband form an equilibrium for different values of P_1 for which the wife has different investment and usage strategies:

•
$$P_1 > \frac{P_2^*}{P_2^* + \lambda(1 - P_2^*)},$$

• $P_1 \in \left[P_2^*, \frac{P_2^*}{P_2^* + \lambda(1 - P_2^*)}\right],$
• $P_1 \in \left[\frac{P_2^*}{(1 - \lambda)(1 - P_2^*) + 1}, P_2^*\right)$
• $P_1 < \frac{P_2^*}{(1 - \lambda)(1 - P_2^*) + 1}.$

The proofs are structured as before.

First, suppose
$$P_1 > \frac{P_2^*}{P_2^* + \lambda(1 - P_2^*)} \ge P_2^*$$
.

- $P_2(P_1, g_1 = 1, e_1 = 1) > P_2^*$ and $P_2(P_1, g_1 = 0, e_1 = 1) = 1$, $P_2(P_1, g_1 = 1, e_1 = 0) < P_2(P_1, g_1 = 1, e_1 = 1)$.
- For the usage choice, the non-expert wife always uses the purchased good if $-c + \beta \lambda \eta^R \ge 0$. This condition is satisfied because we assume $\beta \lambda \eta^R (1 \lambda) \ge \lambda \eta^R \eta^S$ and $c \le \frac{\lambda \eta^R \eta^S}{1 \lambda}$. Under this condition, the expert wife also always uses the purchased good.
- For the investment choice, the wife's static best responses are optimal (see lemma 1) as they induce a reputation $P_2 > P_2^*$, which guarantees the future transfers. Therefore, there is no profitable deviation.
- For the husband's strategy, as any posterior P_2 on the equilibrium path lies above P_2^* , $V_2(P_2)$ is linear in P_2 , so $\mathbb{E}V_2(P_2) = \lambda \eta^R + \mathbb{E}P_2(1-\lambda)\eta^S = \lambda \eta^R + P_1(1-\lambda)\eta^S = V_2(P_1)$. Thus, the husband needs to only compare first-stage payoffs from $T_1 = 1$ and $T_1 = 0$. Since $P_1 > P_2^*$, we have $\mathbb{E}[U_1^H(T_1 = 1)|P_1] > \omega$, so the husband prefers to make the transfer, $T_1 = 1$.

Second, suppose $P_1 \in \left[P_2^*, \frac{P_2^*}{P_2^* + \lambda(1 - P_2^*)}\right]$. Denote $\kappa_1^R \equiv \tilde{\sigma}_2^H(h_2^H)$ when P_1 is in this range and $g_1 = 1$.

- Given the strategies, the updated reputation is $P_2(P_1|g_1 = 1, e_1 = 1) = P_2^*$ and $P_2(P_1|g_1 = 0, e_1 = 1) \ge P_2^*, P_2(P_1, g_1 = 1, e_1 = 0) < P_2(P_1, g_1 = 1, e_1 = 1).$
- For the usage choice, the non-expert wife always uses the purchased good if $-c + \beta \kappa_1^R \lambda \eta^R \ge 0$. Once we define κ_1^R , we can show that this condition is satisfied because we assume $\beta \lambda \eta^R (1-\lambda) \ge \lambda \eta^R \eta^S$ and $c \le \frac{\lambda \eta^R \eta^S}{1-\lambda}$. Under this condition, the expert wife also always uses the purchased good.
- For the investment choice, the non-expert wife is mixing if $\lambda \eta^R - (1 - \lambda)c + \beta \kappa_1^R \lambda \eta^R = \eta^S + \beta \lambda \eta^R$. This condition pins down the husband's transfer strategy: $\kappa_1^R = \frac{\eta^S + \beta \lambda \eta^R - \lambda \eta^R + (1 - \lambda)c}{\beta \lambda \eta^R}$. This value of κ_1^R ensures that the condition for usage $-c + \beta \kappa_1^R \lambda \eta^R \geq 0$ holds. The only strategies that form an equilibrium are the husband randomizing for $g_1 = 1$ and transferring with probability 1 for $g_1 = 0$ as otherwise when $P_1 = P_2^*$ and the non-expert wife's investment rate is λ , the non-expert wife would have the incentive to deviate by increasing her investment rate. For the expert wife, buying the risky good is optimal when $\eta_1^R = \eta^R$ because $\eta^R + \beta \kappa_1^R (\lambda \eta^R + (1 - \lambda) \eta^S) > \eta^S + \beta (\lambda \eta^R + (1 - \lambda) \eta^S)$; buying the safe good is optimal when $\eta_1^R = 0$ because $-c + \beta \kappa_1^R (\lambda \eta^R + (1 - \lambda) \eta^S) < \eta^S + \beta (\lambda \eta^R + (1 - \lambda) \eta^S)$.
- For the husband's strategy, again any posterior P_2 lies above or at P_2^* , so the husband only compares first-stage payoffs from $T_1 = 1$ and $T_1 = 0$ (note that it does not matter whether $P_2 > P_2^*$ or $P_2 = P_2^*$ as the value from the transfer is the same as the outside option for $P_2 = P_2^*$). The expected first-stage payoff from $T_1 = 1$ is increasing in P_1 :

$$\mathbb{E}[U_{1}^{H}(T_{1}=1)|P_{1}] = P_{1}(\lambda\eta^{R} + (1-\lambda)\eta^{S}) + (1-P_{1})(\sigma_{1,g}^{NE}\lambda\eta^{R} + (1-\sigma_{1,g}^{NE})\eta^{S})$$

$$\Rightarrow \frac{\partial\mathbb{E}[U_{1}^{H}(T_{1}=1)|P_{1}]}{\partial P_{1}} = \lambda\eta^{R}(1-\sigma_{1,g}^{NE}) + \eta^{S}(\sigma_{1,g}^{NE} - \lambda) + \frac{\partial\sigma_{1,g}^{NE}}{\partial P_{1}}(\lambda\eta^{R} - \eta^{S}) \ge 0$$

where we use $\frac{\partial \sigma_{1,g}^{NE}}{\partial P_1} \ge 0$. Consider the lower boundary of this interval, $P_1 = P_2^*$:

$$\mathbb{E}[U_1^H(T_1=1)|P_1=P_2^*] = P_2^*(\lambda\eta^R + (1-\lambda)\eta^S) + (1-P_2^*)(\lambda^2\eta^R + (1-\lambda)\eta^S) < P_2^*(\lambda\eta^R + (1-\lambda)\eta^S) + (1-P_2^*)\lambda\eta^R = \omega$$

Thus, the husband prefers the outside option, $T_1 = 0$, at $P_1 = P_2^*$ and switches to $T_1 = 1$ at some belief P_1^* that is above P_2^* .

Third, suppose $P_1 \in \left[\frac{P_2^*}{(1-\lambda)(1-P_2^*)+1}, P_2^*\right)$. Denote $\kappa_2^R \equiv \sigma_2^H(h_2^H)$ when P_1 is in this range and $g_1 = 1$. Denote $\kappa_2^S \equiv \sigma_2^H(h_2^H)$ when P_1 is in this range and $g_1 = 0$.

• Given the strategies, the updated reputation is $P_2(P_1|g_1 = 1, e_1 = 1) = P_2(P_1|g_1 = 0, e_1 = 1) = P_2^*, P_2(P_1|g_1 = 1, e_1 = 0) = 0.$

- For the usage choice, the non-expert wife mixes when making the usage choice if $-c + \beta \kappa_2^R \lambda \eta^R = 0$. Thus, $\kappa_2^R = \frac{c}{\beta \lambda \eta^R}$. Under this condition, the expert wife strictly prefers to use the purchased good because her continuation payoff is higher.
- For the investment choice, the non-expert wife mixes if $\lambda \eta^R (1 \lambda) c \sigma_{1,e}^{NE}(h_{1,e}^{NE}) + \beta \kappa_2^R \lambda \eta^R (\lambda + (1 \lambda) \sigma_{1,e}^{NE}(h_{1,e}^{NE})) = \eta^S + \beta \lambda \eta^R \kappa_2^S$. This condition is equivalent to $\kappa_2^S = \frac{\lambda(\eta^R + c) \eta^S}{\beta \lambda \eta^R}$. The expert wife strictly prefers to buy the risky good when $\eta_1^R = \eta^R$ because $\eta^R + \beta \kappa_2^R (\lambda \eta^R + (1 \lambda) \eta^S) > \eta^S + \beta \kappa_2^S (\lambda \eta^R + (1 \lambda) \eta^S)$. The expert wife strictly prefers to buy the safe good when $\eta_1^R = 0$ because $-c + \beta \kappa_2^R (\lambda \eta^R + (1 \lambda) \eta^S) < \eta^S + \beta \kappa_2^S (\lambda \eta^R + (1 \lambda) \eta^S)$.
- For the husband's strategy, note that on the equilibrium path, any posterior P_2 lies below P_2^* . In this region, $V_2(P_2)$ is constant in P_2 , so $\mathbb{E}V_2(P_2) = \omega = V_2(P_1)$. Thus, the husband needs to only compare first-stage payoffs from $T_1 = 1$ and $T_1 = 0$:

$$\mathbb{E}[U_1^H(T_1 = 1)|P_1] = P_1(\lambda \eta^R + (1 - \lambda)\eta^S) + (1 - P_1)(\sigma_{1,g}^{NE}\lambda \eta^R + (1 - \sigma_{1,g}^{NE})\eta^S) < P_1(\lambda \eta^R + (1 - \lambda)\eta^S) + (1 - P_1)\lambda \eta^R < P_2^*(\lambda \eta^R + (1 - \lambda)\eta^S) + (1 - P_2^*)\lambda \eta^R = \omega$$

Thus, the husband prefers the outside option, $T_1 = 0$, in this interval.

Finally, suppose $P_1 < \frac{P_2^*}{(1-\lambda)(1-P_2^*)+1}$. Denote $\kappa_3^S \equiv \sigma_2^H(h_2^H)$ when P_1 is in this range and $g_1 = 0$.

- Given the strategies, the updated reputation is $P_2(P_1|g_1 = 1, e_1 = 1) < P_2^*$, $P_2(P_1|g_1 = 1, e_1 = 0) = 0$, $P_2(P_1|g_1 = 0, e_1 = 1) = P_2^*$.
- For the usage choice, $e_1 = 0$ for wife types when $\eta_1 = 0$ because using the unproductive good cannot increase the reputation enough to reach threshold P_2^* . Using the productive risky and safe goods is dominant, so $e_1 = 1$ for both wife types when $\eta_1 \neq 0$.
- For the investment choice, the non-expert wife mixes if $\lambda \eta^R = \eta^S + \beta \lambda \eta^R \kappa_3^S$. This condition is equivalent to $\kappa_3^S = \frac{\lambda \eta^R \eta^S}{\beta \lambda \eta^R}$. The expert wife strictly prefers to buy the risky good when $\eta_1^R = \eta^R$ because $\eta^R > \eta^S + \beta \kappa_3^S (\lambda \eta^R + (1 \lambda) \eta^S)$. The expert wife strictly prefers to buy the safe good when $\eta_1^R = 0$ because $0 < \eta^S + \beta \kappa_3^S (\lambda \eta^R + (1 \lambda) \eta^S)$.
- For the husband's strategy, note that on the equilibrium path, any posterior P_2 is again below P_2^* . Thus, as in the previous interval, the husband prefers the outside option, $T_1 = 0$, in this interval.

Combining the four intervals, we conclude that the husband uses a threshold strategy:

$$T_1(P_1) = \begin{cases} 1 \text{ if } P_1 \ge P_1^* \\ 0 \text{ if } P_1 < P_1^* \end{cases}$$

where $P_1^* \in \left(P_2^*, \frac{P_2^*}{\lambda(1-P_2^*)+P_2^*}\right)$.

A.5 Proof of Proposition 3

Proof. Suppose the hiding costs are intermediate: $\frac{\lambda \eta^R - \eta^S}{1 - \lambda} < c \leq \beta \lambda \eta^R$. Lemma 1 pins down the equilibrium strategies at t = 2, except for the husband's strategy

Lemma 1 pins down the equilibrium strategies at t = 2, except for the husband's strategy when he is indifferent between making the transfer or not, i.e., at h_2^H such that $P_2 = P_2^*$. For these cases, let the husband randomize with probabilities $\tilde{\sigma}_2^H(h_2^H) \in [0, 1]$, which are defined further in the proof for various histories.

At t = 1, equilibrium strategies are the following. The expert wife invests iff the risky good is productive and always uses the good unless her reputation is very low:

$$\sigma_{1,g}^{E}(h_{1,g}^{E}) = \begin{cases} 1 \text{ if } \eta_{1}^{R} \neq 0\\ 0 \text{ if } \eta_{1}^{R} = 0 \end{cases};$$

$$\sigma_{1,e}^{E}(h_{1,e}^{E}) = \begin{cases} 1 \text{ if } \eta_{1} \neq 0 \text{ or } \left(\eta_{1} = 0 \text{ and } P_{1} \geq \frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+1}\right)\\ 0 \text{ if } \eta_{1} = 0 \text{ and } P_{1} < \frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+1} \end{cases}$$

The non-expert wife invests with probability decreasing in reputation, always uses the productive and safe goods but uses the unproductive good with positive probability only when her reputation is not too low:

$$\sigma_{1,g}^{NE}(h_{1,g}^{NE}) = \begin{cases} 0 & \text{if } P_1 > \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*} \\ \frac{P_2^* - P_1 + \lambda P_1(1-P_2^*)}{(1-P_1)P_2^*} \le \lambda & \text{if } P_1 \in \left[P_2^*, \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*}\right]; \\ \frac{P_2^* - P_1 + \lambda P_1(1-P_2^*)}{(1-P_1)P_2^*} > \lambda & \text{if } P_1 < P_2^* \\ \sigma_{1,e}^{NE}(h_{1,e}^{NE}) = \\ \text{if } \eta_1 \neq 0 \text{ or } (\eta_1 = 0 \text{ and } P_1 \ge P_2^*) \\ \left[\frac{P_1(1-P_2^*)}{P_2^* - P_1 + \lambda P_1(1-P_2^*)} - 1\right] \frac{\lambda}{1-\lambda} & \text{if } \eta_1 = 0 \text{ and } P_1 \in \left[\frac{P_2^*}{(1-\lambda)(1-P_2^*) + 1}, P_2^*\right] \\ 0 & \text{if } \eta_1 = 0 \text{ and } P_1 < \frac{P_2^*}{(1-\lambda)(1-P_2^*) + 1} \end{cases}$$

Note that if the purchased good is productive or safe, it is weakly dominant to use it for both wife types because the cost is zero and reputation drops to $P_2 = 0$ if the good is not used.

We will show that the strategies of the wife and the husband form an equilibrium for different values of P_1 for which the wife has different usage or investment strategies:

$$P_{1} > \frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+P_{2}^{*}}, \quad P_{1} \in \left[P_{2}^{*}, \frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+P_{2}^{*}}\right],$$

$$P_{1} \in \left[\frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+1}, P_{2}^{*}\right], \text{ and }$$

$$P_{1} < \frac{P_{2}^{*}}{(1-\lambda)(1-P_{2}^{*})+1}.$$

The proofs are structured as before.

First, suppose $P_1 > \frac{P_2^*}{(1-\lambda)(1-P_2^*)+P_2^*}$.

- Given the strategies, $P_2(P_1|g_1 = 1, e_1 = 1) = 1$ and $P_2(P_1|g_1 = 0, e_1 = 1) > P_2^*$.
- For the usage choice, if the purchased good is unproductive, both wife types use it because $-c + \beta \lambda \eta^R \ge 0$.
- For the investment choice, it is optimal for the non-expert wife to buy the safe good because $\eta^S + \beta \lambda \eta^R \ge \lambda \eta^R (1 \lambda)c + \beta \lambda \eta^R$. This condition holds by the assumption that $c > \frac{\lambda \eta^R \eta^S}{1 \lambda}$. For the expert wife, if $\eta_1^R = \eta^R$, it is optimal to buy the risky good because $\eta^R + \beta(\lambda \eta^R + (1 \lambda)\eta^S) > \eta^S + \beta(\lambda \eta^R + (1 \lambda)\eta^S)$. If $\eta_1^R = 0$, it is optimal to buy the safe good for the expert wife because $\eta^S + \beta(\lambda \eta^R + (1 \lambda)\eta^S) > -c + \beta(\lambda \eta^R + (1 \lambda)\eta^S)$.
- The wife's strategy in this interval is the same as in proposition 2, so the husband's payoff is also the same. Thus, $T_1 = 1$ in this interval.

Second, suppose $P_1 \in \left[P_2^*, \frac{P_2^*}{(1-\lambda)(1-P_2^*)+P_2^*}\right]$. Denote $\kappa_1^S \equiv \sigma_2^H(h_2^H)$ when P_1 is in this interval and $g_1 = 0$.

- Given the strategies, $P_2(P_1|g_1 = 1, e_1 = 1) \ge P_2^*$ and $P_2(P_1|g_1 = 0, e_1 = 1) = P_2^*$.
- For the usage choice, if the purchased good is unproductive, both wife types use it because $-c + \beta \lambda \eta^R \ge 0$.
- For the investment choice, the non-expert wife mixes if $\eta^S + \beta \kappa_1^S \lambda \eta^R = \lambda \eta^R (1 \lambda)c + \beta \lambda \eta^R$. These conditions pin down the husband's transfer strategy: $\kappa_1^S = \frac{\lambda \eta^R \eta^S (1 \lambda)c + \beta \lambda \eta^R}{\beta \lambda \eta^R}$. For the expert wife, it is optimal to buy the risky good if $\eta_1^R = \eta^R$ because $\eta^R + \beta(\lambda \eta^R + (1 \lambda)\eta^S) > \eta^S + \beta \kappa_1^S(\lambda \eta^R + (1 \lambda)\eta^S)$. If $\eta_1^R = 0$, it is optimal to buy the safe good for the expert wife because $\eta^S + \beta \kappa_1^S(\lambda \eta^R + (1 \lambda)\eta^S) > -c + \beta(\lambda \eta^R + (1 \lambda)\eta^S)$.
- For the husband's strategy, on the equilibrium path, updated reputation is always $\geq P_2^*$. Thus, the husband needs to only consider the first-stage payoff. The expected first-stage payoff is

$$\Delta \mathbb{E}[U_1^H(T_1=1)|P_1] = \lambda \eta^R + P_1 \frac{1-\lambda}{P_2^*} (\eta^S - (1-P_2^*)\lambda \eta^R) - \omega,$$

which is increasing in P_1 . At the lower end of the interval, $P_1 = P_2^*$, the payoff from making the transfer is lower than the outside option:

$$\lambda \eta^R + (1-\lambda)\eta^S - (1-\lambda)(1-P_2^*)\lambda \eta^R < \lambda \eta^R + P_2^*(1-\lambda)\eta^S = \omega$$

Thus, the husband chooses $T_1 = 0$ at $P_1 = P_2^*$ and switches to $T_1 = 1$ at some higher P_1 in this interval.

Third, suppose $P_1 \in \left[\frac{P_2^*}{(1-\lambda)(1-P_2^*)+1}, P_2^*\right)$. Denote $\kappa_2^R \equiv \sigma_2^H(h_2^H)$ when P_1 is in this range and $g_1 = 1$. Denote $\kappa_2^S \equiv \sigma_2^H(h_2^H)$ when P_1 is in this range and $g_1 = 0$.

- Given the strategies, the updated reputation is $P_2(P_1|g_1 = 1, e_1 = 1) = P_2(P_1|g_1 = 0, e_1 = 1) = P_2^*$ and $P_2(P_1|g_1 = 1, e_1 = 0) = 0$.
- For the usage choice, the non-expert wife mixes if $-c + \beta \kappa_2^R \lambda \eta^R = 0$. Thus, $\kappa_2^R = \frac{c}{\beta \lambda \eta^R}$. Under this condition, the expert wife strictly prefers to use the purchased good because her continuation payoff is higher.
- For the investment choice, the non-expert wife mixes if $\lambda \eta^R (1 \lambda) c \sigma_{1,e}^{NE}(h_{1,e}^{NE}) + \beta \kappa_2^R \lambda \eta^R (\lambda + (1 \lambda) \sigma_{1,e}^{NE}(h_{1,e}^{NE})) = \eta^S + \beta \lambda \eta^R \kappa_2^S$. This condition is equivalent to $\kappa_2^S = \frac{\lambda(\eta^R + c) \eta^S}{\beta \lambda \eta^R}$. The expert wife strictly prefers to buy the risky good when $\eta_1^R = \eta^R$ because $\eta^R + \beta \kappa_2^R (\lambda \eta^R + (1 \lambda) \eta^S) > \eta^S + \beta \kappa_2^S (\lambda \eta^R + (1 \lambda) \eta^S)$. The expert wife strictly prefers to buy the safe good when $\eta_1^R = 0$ because $-c + \beta \kappa_2^R (\lambda \eta^R + (1 \lambda) \eta^S) < \eta^S + \beta \kappa_2^S (\lambda \eta^R + (1 \lambda) \eta^S)$.
- For the husband's strategy, as the updated reputation is always $\leq P_2^*$, the husband needs to only consider the first-stage payoff. The expected first-stage payoff is the same as above:

$$\mathbb{E}[U_1^H(T_1=1)|P_1] = \lambda \eta^R + P_1 \frac{1-\lambda}{P_2^*} (\eta^S - (1-P_2^*)\lambda \eta^R)$$

Since the payoff is increasing in P_1 , and it is lower than ω at $P_1 = P_2^*$, it is also lower than ω at all $P_1 < P_2^*$.

Finally, suppose $P_1 < \frac{P_2^*}{(1-\lambda)(1-P_2^*)+1}$. Denote $\kappa_3^S \equiv \sigma_2^H(h_2^H)$ when P_1 is in this range and $g_1 = 0$.

- Given the strategies, the updated reputation is $P_2(P_1|g_1 = 1, e_1 = 1) < P_2^*$, $P_2(P_1|g_1 = 0, e_1 = 1) = P_2^*$ and $P_2(P_1|g_1 = 1, e_1 = 0) = 0$.
- For the usage choice, using the unproductive good cannot increase the reputation enough to reach threshold P_2^* . Therefore, $e_1 = 0$ for both types of wives when $\eta_1 = 0$. Using the productive risky and safe goods is weakly dominant, so $e_1 = 1$ for both types of wives when $\eta_1 \neq 0$.
- For the investment choice, the non-expert wife mixes if $\lambda \eta^R = \eta^S + \beta \lambda \eta^R \kappa_3^S$. This condition is equivalent to $\kappa_3^S = \frac{\lambda \eta^R \eta^S}{\beta \lambda \eta^R}$. The expert wife strictly prefers to buy the risky good when $\eta_1^R = \eta^R$ because $\eta^R > \eta^S + \beta \kappa_3^S (\lambda \eta^R + (1 \lambda) \eta^S)$. The expert wife strictly prefers to buy the safe good when $\eta_1^R = 0$ because $0 < \eta^S + \beta \kappa_3^S (\lambda \eta^R + (1 \lambda) \eta^S)$.
- For the husband's strategy, as above, as the updated reputation is always $< P_2^*$, the payoff is also lower than ω in this interval.

Combining the four intervals, we conclude that the husband uses a threshold strategy:

$$T_1(P_1) = \begin{cases} 1 \text{ if } P_1 \ge P_1^* \\ 0 \text{ if } P_1 < P_1^* \end{cases}$$

where $P_1^* \in \left(P_2^*, \frac{P_2^*}{(1-\lambda)(1-P_2^*) + P_2^*}\right).$

B Technical Appendix

B.1 Sampling

Enumerators were instructed to enroll households using the "left hand rule":

- 1. You will pick a house as a starting point,
- 2. Proceed such that the house you have picked is on your left.
- 3. Count 3 houses and stop at the third house. This house will be selected for sampling.
- 4. You will identify the owners of the household and ask to speak to a husband and wife pair who reside in the household.
- 5. If they are eligible for the survey, conduct the survey. Otherwise skip to the next step.
- 6. Once you are finished with the survey, or if the couple is ineligible, continue in the original direction you were walking in.
- 7. Count three houses, stop at the third house, and repeat steps 4-6.

B.2 Compensation and Prices

Initially, husbands received MWK 300 and women received MWK 200 as baseline compensation for participation in the survey. Women were offered an additional MWK 100 for participating in the signaling activity in the signaling experiment. The hiding prices were MWK 50 (low hiding cost), MWK 100 (intermediate hiding cost) or MWK 150 (high hiding cost). However, enumerators noted during the first month of data collection, that the compensations and prices might be insufficient to incentivize the respondents to thoroughly think about their choices. We thus doubled the compensations given to the respondents for 2/3 of the remaining surveys to allow for geographic variation in the low- and high-compensation versions (in total 28% of surveys had the low compensation and 72% the high compensation). To keep the ratio of prices to compensation constant in the signaling experiment, we also doubled the hiding prices to MWK 100 (low hiding cost), MWK 200 (intermediate hiding cost) or MWK 300 (high hiding cost). In the analysis we express transfers as a percent of total compensation in the transfer experiment and control for baseline compensation fixed-effects in both experiments.

B.3 Defining the Salience Treatment in the Transfer Experiment

Initially, all husbands played the game at the end of the survey, either before or after the MER module (version 1). However, enumerators noted during the first month of data collection (17% of data collected), that the signaling experiment was serving as a large salience treatment itself. We thus moved the transfer at the beginning of the survey in the control arm (version 2) and coded all data collected thus far as assigned to the salience treatment. All results are robust to controlling for version fixed effects.

C Empirical Appendix

C.1 Figures

Figure C.1: Handout to elicit respondent's priors about own and spouse's score



Figure C.2: Percentage of wives participating in the signaling activity for each reported level of risk preference



Notes: The wife's risk preference was the response to the question "You are buying 10 packets of seed. There are two kind of seeds. One will mature with a low yield, but it is guaranteed that all seeds will mature. The other will mature with a higher yield: You will get 3x as much. However, on average, half of all packets produced never mature. That means you could have all good packets or all bad packets or somewhere in between. Out of the 10 packets you will buy, how many packets of the risky seed will you buy?"

Figure C.3: Total forgone earnings in experiment 2 for each rounded mean score prior



Figure C.4: Low hiding cost effectiveness sticker: airtight crop storage bag

PURDUE EXTENSION

Matumba a Purdue Improved Crop Storage (PICS)

Vuto:

Tizilombo timaononga zokolola ndipo zotsatira zake zimakhala:

- Kutsika kolite
- Kupunguka kwa kulemera/kuluza weti
- Kugwiritsa ntchito mankhwala ophera tizilombo mochuluka

Matumba a PICS am<mark>athandiza:</mark>

Matumba a Hermetic triple-layer amateteza zokolola ku tizilombo popanda kugwiritsa ntchito mankhwala.

Zothandiza pa mbewu monga:

Khobwe, Chimanga, Nyemba, Nandolo, Mawere, Mapira, Tirigu, Mtedza, Mpunga, Soya.

Matumba a PICS :

- Amaonjezera phindu/chuma
- Amathandiza kukhala ndi chakudya chokwanira
- Kuthandiza kukhala ndi chakudya cha pamwamba/chakolite yabwino

- Kuthandiza kukhala ndi mbewu za kolite yabwino

English Translation:

The Problem:

Insects damage stored grains, leading to:

- Quality loss
- Weight loss
- Excessive use of chemicals

The PICS Solution:

Hermetic triple-layer bags protect grain from insect damage without using chemicals.

Effective for:

Cowpea, Maize, Beans, Peas, Sorghum, Millett, Wheat, Groundnut, Rice, Soybean

PICS Bags:

- Increase income
- Improve food security
- Maintain food quality
- Maintain seed quality





Figure C.5: High hiding cost effectiveness sticker: children's book

Kuwerengera ana anu ndi chuma cha mtsogolo!

- Luso pakaganizidwe ka ana ang'ono ndi njira yokhayo yofunika yomwe imathandiza ana kudzakhala ndi tsogolo lopambana m'moyo. Izi zimakhudza kuti adzathe kupitiliza maphunziro awo mpaka ku ma college ndi maluso ena apamwamba.
- Luso pakaganizidwe silachibadwa. Izi zikhoza kukhala choncho chifukwa cha Khama la makolo. Nthawi yoyenera kuthandiza ana ndi luso pakaganizidwe ndi pomwe akadali achichepere kungochokera pomwe abadwa.
- Kafukufuku wambiri wapeza kuti ana amene amakhala okonzekera akadali achichepere amayamba sukulu akudziwa mau ochuluka komanso ndi maluso apamwamba.



English Translation:

- Cognitive skills of young children are an important factor in explaining success later-on in life. They affect the likelihood of acquiring higher education and advanced training.

- Cognitive skills are not fixed. They can be influenced by parental efforts. The most effective period for cognitive skill investment by parents is early on in the life of their children – in fact, from birth.

- Numerous studies have found children who are ready to frequently at a very early age enter school with larger vocabularies and more advanced comprehension skills.

Figure C.6: Donated sticker

Kupezeka ku mabanja akumalawi ndi

Stanford University

English Translation: Donated to Malawian families by Stanford University

Figure C.7: Heterogeneity by discretionary transfer size, using the wife's random-forest predicted expertise in experiment 3



Notes: The graph shows the coefficients and confidence intervals from OLS regressions with Huber-White robust SEs. Low/High Transfers correspond to below/above the median. Rows 1 to 3 control for enumerator and compensation fixed effects (and version fixed effects for the transfer experiment) as well as the wife and the husband's age, education, average income in the last two months, variability of income (whether income is the same in most months or varies a lot), risk preferences, math and raven scores, and years married, number of children and number of household members, and MER index. Controls are as reported by the husband in the transfer and signaling experiment and as reported by the wife in the market experiment. Row 3 controls for enumerator and market fixed effects as well as the wife's age, education, average income in the last two months, risk preferences, math score, as well as the husband's average income in the last two months, years married, and the number of children and household members. Coefficients are presented as percentage point deviations from the control means.

Figure C.8: Heterogeneity by number of children in the household (using wife's second-order math beliefs (top) and random-forest predicted types (bottom)



Notes: The graph shows the coefficients and confidence intervals from OLS regressions with Huber-White robust SEs. Rows 1 to 3 control for enumerator and compensation fixed effects (and version fixed effects for the transfer experiment) as well as the wife and the husband's age, education, average income in the last two months, variability of income (whether income is the same in most months or varies a lot), risk preferences, math and raven scores, and years married, number of children and number of household members, and MER index. Controls are as reported by the husband in the transfer and signaling experiment and as reported by the wife in the market experiment. Row 4 controls for enumerator and market fixed effects as well as the wife's age, education, average income in the last two months, risk preferences, math score, as well as the husband's average income in the last two months, years married, and the number of children and household members. Coefficients are presented as percentage point deviations from the control means.

C.2 Tables

Table C.1: Experiment 1: 50 randomly selected example answers to questions 1) Is your wife ever tempted to buy things advertised at the market? and 2) Has it ever happened that your wife bought something that did not work as advertised?

Tempted	Did not work
Charcoal burner	Soap
Pot	Insecticides that wouldn't work
Clothes	Wrapping paper
Door mat	Non-durable cloth
Charcoal stove	Тоу
Rose flower	Pair of trousers
Shoes	Shoes
Not yet but tempted	Piece of cloth poorly sewn on the edges
New dress	Shoes
Soap	Plastic bucket which broke within few days
Medicine from a pharmacy	Clothes
Skirt	Chemical that didn't work
Cooking stove and clothes	Clothes
Snacks for the child but in large quantity	Phone battery
Doll	Skirt
Plates when the goal was to buy a jar	Battery torch which was not working
Trousers for me	Insecticides
Metal plates	Battery torch which could not produce light
Cooking stove	Pesticides that didn't work
Charcoal burner	Torn blouse
Charcoal burner	Battery torch which was not working
Pair of non-durable shoes	Radio with cut wires inside
Dress	Basin
Dress	Cooking stove
Toys	Stolen metal bars which were painted to conceal the rust
Not yet but tempted	Mosquito repellent
Charcoal burner	Torn dress
Charcoal stove	Plastic bucket
Doll	Bucket that leaked
Duvet	Slippers that were not durable
Water treatment solution	Stove
Hats	Expired cooking oil
Cooking stove and pesticides	Chemicals used to kill cockroaches
Kitchen utensils	Headache medicines
Doll for the child	Too tight clothes
Clothes	Non-durable shoes
Mosquito net	Clothes
New drug advertised to help joint pain	Non-durable slippers
Not vet but tempted	Torn trousers
Dress	Used phone (she thought it was new)
Tov	Clothes that lost colour after one wash
Piece of cloth	Too tight skirt
Cooking stove	Container
Basin	Pain killers that did not work
Cooking stove that uses less charcoal	Mosquito pesticide that never even killed ONE mosquito
Shirt for the child	Bucket
Short trousers for me	Shoes in the wrong size (the yendor had told her they would fit anyways)
Toy for her child	Bucket which broke after a few days
Children toys and charcoal burner	Clothes
Cooking stove	Soap

Notes: To create the table, we randomly selected 50 answers to question 1 and 50 answers to question 2 - i.e., the answer choices on the left and on the right do not have to be from the same person. We preferred this to selecting 50 subjects randomly as subjects could answer no to each question - thus, we would provide fewer examples by selecting 50 respondents as opposed to 50 answer choices for each question.
	Mean (SD)	β_{Salience} (SE)	$\beta_{\text{Low MER}}$ (SE)	$\beta_{\text{Low MER*Salience}}$
Variables	(1)	(2)	(3)	(4)
Wife's age				
	30.37	0.67	-0.64	1 30
Husband's age	(8.93)	(0.64)	(1.29)	(1.60)
	35.83	1.31	0.20	0.30
Wife's education	(10.16)	(0.72)	(1.33)	(1.71)
	5.68	0.02	0.57	-0.56
Husband's education	(3.27)	(0.23)	(0.51)	(0.60)
	6.77	0.05	0.54	0.50
Wife's quality score	(3.54)	(0.25)	(0.52)	(0.64)
	4.17	0.02	0.01	0.17
Husband's quality score	(1.20)	(0.09)	(0.17)	(0.21)
	4.16	-0.01	-0.06	0.02
Wife's raven score	(1.25)	(0.09)	(0.19)	(0.23)
	4.05	0.10	0.12	-0.13
Husband's raven score	(1.95)	(0.13)	(0.28)	(0.34)
	4.98	0.32	-0.01	0.16
Wife's math score	(2.01)	(0.14)	(0.29)	(0.35)
YY 1 11 11	3.26	-0.04	0.06	-0.06
Husband's math score	(1.66)	(0.12)	(0.25)	(0.30)
	4.09	0.14	0.08	-0.03
Wife's avg. income last two months (MWK, H's	(1.47)	(0.10)	(0.23)	(0.27)
report)	4,967.04	676.12	-201.09	-1,664.66
	(10, 458.39)	(736.36)	(1,567.70)	(1,790.05)
Husband's avg. income last two months (MWK,	29,770.05	3,944.77	5,572.03	-5,463.49
H's report)	(33,075.29)	(2,284.10)	(5,487.73)	(6,331.62)
W.C. And States of Communication	2.90	0.24	0.32	-0.46
whe's risk preference	(2.44)	(0.17)	(0.36)	(0.43)
Hashend's risk professores	4.60	0.02	-0.38	0.66
Husband's risk preference	(2.98)	(0.20)	(0.41)	(0.49)
Veens meaning	9.91	0.30	-0.19	0.56
rears married	(8.48)	(0.61)	(1.21)	(1.49)
N of Children	2.63	0.07	-0.21	-0.05
N of Children	(1.56)	(0.11)	(0.23)	(0.27)
Household members	5.02	0.08	-0.17	-0.09
HOUSCHOLD INCHIDELS	(1.86)	(0.13)	(0.28)	(0.33)
Aug. transforg (H to W) lost two months (MWV	8,451.97	490.47	901.20	-2,200.67
H's report)	(11,411.97)	(823.33)	(1,766.80)	(2,027.78)

Table C.2: The transfer experiment: Balance, by market expertise reputation (MER) and salience

Notes: The table shows results from OLS regressions with enumerator fixed effects and Huber-White robust SEs. Market Expertise Reputation (MER) defined as before. All MWK values are winsorized at 3 SDs. Wife's and husband's average income as well as transfers as reported by the husband. Regressions control for enumerator fixed effects. No differences are significant after adjusting for false discovery rates (q-values).

	Index	Purchases	Tempted	Manage	Change	$\mathrm{MER} \leq 2$	MER=3	MER=4
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Salience	2.143	1.617	1.636	0.841	0.349	-7.569	4.332	1.616
	(1.720)	(1.691)	(1.743)	(1.862)	(1.667)	(4.483)	(3.081)	(2.126)
Low MER	0.343	0.402	-2.014	-4.453	-2.406			
	(3.374)	(3.773)	(3.256)	(2.728)	(5.618)			
Low MER*Salience	-9.184	-8.266	-4.980	-0.660	3.018			
	(4.231)	(4.742)	(4.043)	(3.427)	(7.032)			
Mean (Control)	68.889	68.823	69.281	70.248	68.920	68.393	65.564	70.768
Observations	1093	1093	1093	1093	1093	186	336	571

Table C.3: Transfer experiment: Effect of reputation salience on amount (%) transferred from the husband to the wife, by MER subcomponent and by MER score

Notes: The table shows results from OLS regressions with Huber-White robust SEs. Market Expertise Reputation (MER) is an index that takes the values 0 to 4, depending on how many of the following questions the husband affirmed: i) his wife had never bought anything that did not work as advertised ("Purchases", 86%), ii) his wife is never tempted to buy advertised goods with uncertain return at the market ("Tempted", 80%), iii) he believes his wife can manage money received from the husband well compared to other women in the community ("Manage", 70%), and iv) his wife can do calculations correctly in her head when she requests change in the market ("Math", 95%). The table shows the effect of the salience treatment for a negative answer to each of the individual MER subcomponents and for the different MER scores. We combine MER scores of 0, 1, and 2 because of small sample sizes: 6 women have an MER of 0 (0.6%), 38 of 1 (3.5%), 142 of 2 (13.0%), 336 of 3 (30.7%) and 571 of 4 (52.2%). All regressions include enumerator, compensation and version fixed effects. Controls include wife and husband's age, education, average income and transfers in the last two months, variability of income (whether income is the same in most months or varies a lot), risk preferences, math and raven scores, and years married, number of children and number of household members.

	Easy (1	N = 550)	Hard (N = 543)	
	Mean	SD	Mean	SD	Diff.
Sponge	0.898	0.303	0.738	0.440	0.160
Bottle	0.638	0.481	0.580	0.494	0.058
Razor	0.593	0.492	0.394	0.489	0.199
Toothbrush	0.902	0.298	0.661	0.474	0.241
Flour	0.725	0.447	0.529	0.500	0.197
Candle	0.933	0.251	0.737	0.441	0.196
Wife's quality score	4.689	1.042	3.639	1.115	1.050
Husband's quality score	4.709	1.033	3.600	1.212	1.109
Prior quality, wife	4.951	1.134	4.297	1.373	0.654
W about H about W	4.798	1.431	4.355	1.512	0.443
W believes H will update negatively (/					

Table C.4: Scores and beliefs on the quality quiz, by quiz difficulty

Two-sided t-tests. "W about H about W" refers to the wife's belief about her husband's belief about her score.

	Whole sample (N=1093)			Participation sample (N=786)				Whole sample (N=1093)
	Panel A: By price and low perceived score							
	Initial score	Participate (%)	Foregone comp.	Initial score	Errors corrected	Hiding fee	Final score	Total forgone
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Non-Expert	-0.238 (0.130)	-3.715 (4,595)	7.430 (9.190)	-0.358 (0.147)	0.267	27.451 (11.352)	-0.090 (0.158)	28.300 (10,935)
Intermediate Cost	0.126 (0.116)	(1.000) 4.359 (3.784)	(0.100) -8.718 (7.568)	0.196 (0.130)	-0.170 (0.075)	(11.664) (11.359)	(0.130) 0.026 (0.131)	(10.000) 3.095 (10.867)
High Cost	(0.110) 0.082 (0.107)	-0.653	(1.306) (7.770)	(0.130) 0.049 (0.126)	-0.241	(11.000) 17.187 (12.625)	(0.101) -0.192 (0.120)	13.779
Non-Expert*Intermediate Cost	(0.107) 0.064 (0.186)	(5.890) -11.955 (6.883)	(7.779) 23.910 (13.767)	(0.120) 0.027 (0.210)	(0.008) -0.062 (0.144)	(12.035) 14.380 (21.786)	(0.129) -0.035 (0.210)	(11.004) 21.223 (17.725)
Non-Expert*High Cost	(0.130) 0.184 (0.174)	(0.883) -9.744 (6.533)	(13.707) 19.488 (13.066)	(0.210) 0.225 (0.207)	(0.144) -0.237 (0.127)	(21.780) -16.052 (22.039)	(0.213) -0.012 (0.217)	(17.723) -3.386 (17.374)
Mean (Low Cost & Expert) P-value (Expert vs. Non-Expert, Intermediate Cost) P-value (Expert vs. Non-Expert, High Cost)	$\begin{array}{c} (0.111) \\ 4.149 \\ 0.183 \\ 0.647 \end{array}$	$\begin{array}{c} (0.000) \\ 76.349 \\ 0.002 \\ 0.004 \end{array}$	$\begin{array}{c} (10.000) \\ 47.303 \\ 0.002 \\ 0.004 \end{array}$	$\begin{array}{c} (0.201) \\ 4.196 \\ 0.026 \\ 0.366 \end{array}$	$\begin{array}{c} (0.121) \\ 0.462 \\ 0.029 \\ 0.663 \end{array}$	$\begin{array}{c} 46.196 \\ 0.023 \\ 0.556 \end{array}$	$\begin{array}{c} (0.211) \\ 4.658 \\ 0.407 \\ 0.493 \end{array}$	82.573 0.000 0.073

Table C.5: Outcomes in the signaling experiment, including controls

			Panel B: I	By price ε	and difficult	y of the qu	iz	
	Initial score	Participate (%)	Forgone comp.	Initial score	Errors corrected	Hiding fee	Final score	Total forgone
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Harder Version	-1.174	5.798 (4.330)	-11.595 (8,660)	-1.136 (0.126)	0.307	30.726 (10.635)	-0.829 (0.141)	12.962 (10.501)
Intermediate Cost	(0.100) 0.061 (0.102)	4.106	-8.211	(0.120) 0.063 (0.120)	-0.210	-2.061	(0.147) -0.147 (0.124)	-8.200
High Cost	0.066	(4.554) -3.149	(9.068) 6.297	0.084	(0.071) -0.219	(10.016) 8.663	(0.124) -0.135	(10.769) 8.582
Harder Version*Intermediate Cost	(0.102) 0.160	(4.575) -9.818	(9.151) 19.636	(0.121) 0.204	$(0.072) \\ 0.069$	(13.158) 42.513	(0.125) 0.272	$(11.650) \\ 43.896$
Harder Version*High Cost	$(0.159) \\ 0.213$	(6.502) -4.101	(13.005) 8.202	(0.183) 0.093	(0.130) -0.200	(19.576) 5.820	(0.200) -0.107	(17.447) 10.680
Mean (Low Cost & Easier Version) P-value (Easier vs. Harder Version, Intermediate Cost)	(0.152) 4.594 0.000	(6.317) 71.875 0.402	(12.633) 56.250 0.402	$(0.178) \\ 4.601 \\ 0.000$	$(0.123) \\ 0.406 \\ 0.000$	$\begin{array}{c} (21.384) \\ 40.580 \\ 0.000 \end{array}$	(0.192) 5.007 0.000	(17.536) 85.417 0.000
P-value (Easier vs. Harder Version, High Cost)	0.000	0.711	0.711	0.000	0.094	0.043	0.000	0.088

Notes: The table shows results from OLS regressions with Huber-White robust SEs. Low Perceived Score is an indicator that takes the value 1 if the wife reports an average weighted score that is lower than 5 (39% of women). The weighted average is calculated as the average across all scores, weighted by the probability assigned to each score by the woman (each woman placed 10 beans on the 7 different scores). Foregone comp. is the amount of money wives left on the table by opting out of the game. All regressions include enumerator and compensation fixed effects. The p-value is the p-value from a Wald test comparing outcomes between low perceived score and high perceived score wives or between the hard and the easy version when the price of hiding is intermediate or high. Controls include wife and husband's age, education, average income in the last two months, variability of income (whether income is the same in most months or varies a lot), risk preferences, math, and raven scores, and years married, number of children and number of household members, and the wife's MER.

	Whole sample (N=1093)			Participation sample $(N=786)$				Whole sample (N=1093)
			Panel A	: By price	e and wife's	education		
	Initial score	Participate (%)	Foregone comp.	Initial score	# Errors corrected	Hiding fee	Final score	Total forgone
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Low Education	-0.268 (0.126)	-9.587 (4.480)	19.174 (8.959)	-0.182 (0.148)	-0.011 (0.101)	-2.412 (10.282)	-0.193 (0.155)	13.592 (10.341)
Intermediate Cost	0.248 (0.119)	0.773 (3.972)	-1.547 (7.945)	0.268 (0.133)	-0.168 (0.087)	21.314 (12.952)	0.099 (0.134)	15.613 (11.783)
High Cost	0.140 (0.111)	-3.823 (4.015)	7.646 (8.030)	0.114 (0.127)	-0.348 (0.078)	6.787 (13,566)	-0.234 (0.132)	10.435 (11.820)
Low Education*Intermediate Cost	-0.224 (0.180)	-3.552 (6.647)	7.105 (13.293)	(0.121) -0.139 (0.204)	-0.050 (0.131)	-9.367 (19.339)	-0.189 (0.209)	-4.305 (17.177)
Low Education*High Cost	0.014 (0.171)	-3.162 (6.387)	6.324 (12.774)	0.028 (0.205)	0.063 (0.119)	(21.212)	(0.091) (0.210)	(17.072)
Mean (Low Cost & High Education) P-value (High vs. Low Education, Intermediate Cost) P-value (High vs. Low Education, High Cost)	4.168 0.000 0.029	79.703 0.008 0.005	$\begin{array}{c} 40.594 \\ 0.008 \\ 0.005 \end{array}$	$\begin{array}{c} (0.200) \\ 4.130 \\ 0.024 \\ 0.279 \end{array}$	$\begin{array}{c} (0.110) \\ 0.553 \\ 0.457 \\ 0.413 \end{array}$	$ \begin{array}{c} (11.112)\\ 55.280\\ 0.472\\ 0.570 \end{array} $	$\begin{array}{c} (0.210) \\ 4.683 \\ 0.007 \\ 0.477 \end{array}$	84.653 0.500 0.071

Table C.6: Outcomes in the signaling experiment, by correlates of expertise

	Panel B: By price and wife's self-esteem								
	Initial score	Participate (%)	Foregone comp.	Initial score	# Errors corrected	Hiding fee	Final score	Total forgone	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Low Self-Esteem	0.996	-4.444	8.888	0.985	-0.095	-9.165	0.890	0.652	
	(0.113)	(5.487)	(10.975)	(0.129)	(0.103)	(10.850)	(0.126)	(11.675)	
Intermediate Cost	0.139	1.173	-2.346	0.219	-0.176	22.273	0.042	15.288	
	(0.101)	(3.579)	(7.159)	(0.114)	(0.077)	(11.411)	(0.120)	(10.233)	
High Cost	0.086	-2.988	5.975	0.065	-0.318	17.728	-0.253	16.121	
	(0.097)	(3.613)	(7.225)	(0.112)	(0.070)	(12.439)	(0.119)	(10.454)	
Low Self-Esteem*Intermediate Cost	0.020	-7.267	14.534	0.020	-0.052	-21.326	-0.032	-7.516	
	(0.174)	(8.169)	(16.338)	(0.192)	(0.134)	(20.130)	(0.184)	(18.810)	
Low Self-Esteem*High Cost	-0.059	-6.734	13.468	0.047	0.009	-18.877	0.056	-1.299	
	(0.160)	(7.489)	(14.978)	(0.186)	(0.122)	(21.334)	(0.188)	(18.171)	
Mean (Low Cost & High Self-Esteem)	3.830	76.471	47.059	3.855	0.570	57.014	4.425	90.657	
P-value (High vs. Low Self-Esteem, Intermediate Cost)	0.000	0.055	0.055	0.000	0.082	0.071	0.000	0.641	
P-value (High vs. Low Self-Esteem, High Cost)	0.000	0.030	0.030	0.000	0.185	0.124	0.000	0.963	

Notes: The table shows results from OLS regressions with Huber-White robust SEs. Low Education is an indicator that takes the value 1 if the wife has fewer than 6 years of education (the median, 46% of women). Low Self-Esteem is an indicator that is 1 if the wife's perceived quality quiz score is lower than her actual score (26% of women). Foregone comp. is the amount of money wives left on the table by opting out of the game. All regressions include enumerator and compensation fixed effects. The p-value is the p-value from a Wald test comparing outcomes between wives with low or high education or between wives with low or high self-esteem when the hiding cost is high.

Table C.7: The signaling experiment: Balance, by low expertise (NE) and hiding cost (intermediate IC or high HC)

	Mean (SD)	$\beta_{\rm NE}$ (SE)	β_{IC} (SE)	$\beta_{\rm HC}$ (SE)	$\beta_{\rm NE \times IC}$ (SE)	$\beta_{\text{NE} \times \text{HC}}$ (SE)
Variables	(1)	(2)	(3)	(4)	(5)	(6)
Wife's age						
Husband's age	30.37 (8.93)	-0.88 (0.95)	0.47 (0.89)	0.74 (0.79)	0.38 (1.42)	1.20 (1.31)
Wife's education	35.83 (10.16)	-1.30 (1.09)	-0.06 (1.00)	1.08 (0.93)	1.82 (1.60)	0.13 (1.49)
Husband's education	5.68 (3.27)	-0.12 (0.34)	0.23 (0.33)	0.17 (0.31)	-0.49 (0.51)	-0.69 (0.47)
Wife's quality score	6.77 (3.54)	0.56 (0.38)	0.01 (0.35)	-0.21 (0.34)	-0.75 (0.55)	-0.77 (0.52)
Husband's quality score	4.17 (1.20)	-0.25 (0.13)	(0.17) (0.12)	0.11 (0.11)	-0.00 (0.18)	0.12 (0.18)
Wife's raven score	4.16 (1.25)	-0.22 (0.14)	(0.01)	-0.14 (0.12)	0.06 (0.20)	-0.08 (0.19)
Husband's raven score	4.05 (1.95)	(0.01) (0.19)	(0.10) (0.19)	(0.18)	-0.18 (0.28)	(0.06) (0.27)
Wife's math score	4.98 (2.01)	-0.48 (0.20)	(0.20)	0.12 (0.19)	(0.31)	(0.13) (0.29)
Husband's math score	3.26 (1.66)	-0.16 (0.18)	(0.25) (0.16)	0.28 (0.16)	-0.33 (0.25)	-0.33 (0.24)
Wife's avg. income last two months (MWK, W's report)	(1.47) 10,659.82	(0.10) (0.16) -604.51	(0.03) (0.15) 1,506.05	(0.13) 519.21	-0.06 (0.23) -2,523.94	-0.11 (0.21) -1,329.13
Husband's avg. income last two months (MWK, W's report)	(17,556.52) 15,506.03 (23,030.80)	(1,871.15) 2,927.65 (2,478.82)	(1,821.95) 2,268.78 (2,310.57)	$(1,657.60) \\511.50 \\(2,052.91)$	(2,664.53) -7,621.71 (3,553.41)	(2,505.84) -2,995.13 (3,409.21)
Wife's risk preference	2.90 (2.44)	-0.13 (0.26)	0.37 (0.24)	0.23 (0.23)	-0.05 (0.37)	0.19 (0.36)
Husband's risk preference	4.60 (2.98)	-0.30 (0.32)	-0.33 (0.29)	-0.07 (0.27)	0.56 (0.45)	0.25 (0.43)
Years married	9.91 (8.48)	(0.89)	0.24 (0.83)	0.70 (0.78)	-0.36 (1.31)	0.28 (1.24)
N of Children	2.63 (1.56)	-0.21 (0.16)	(0.16)	(0.12) (0.15)	(0.24)	-0.02 (0.22)
Household members	(1.86)	-0.28 (0.20)	(0.19)	(0.13) (0.18)	(0.28)	0.03 (0.27)
Avg. transfers (H to W) last two months (MWK, W's report)	(8,097.11) (3.31)	-519.54 (834.24) -0.15	507.42 (774.20) -0.01	(761.49) 0.10	(1,202.14) 0.20	-780.58 (1,144.63) 0.02
MER	(0.86)	(0.10)	(0.08)	(0.08)	(0.14)	(0.13)

Notes: The table shows results from OLS regressions with enumerator fixed effects and Huber-White robust SEs. Non-Expert defined as before. All MWK values are winsorized at 3 SDs. Wife's and husband's average income as well as transfers as reported by the wife. The regressions control for enumerator and compensation fixed-effects. No differences are significant after adjusting for false discovery rates (q-values).

Table C.8: The signaling experiment: Balance, by hard version (HV) and hiding cost (intermediate IC or high HC)

	Mean (SD)	$\beta_{\rm HV}$ (SE)	β_{IC} (SE)	$\beta_{\rm HC}$ (SE)	$\beta_{\rm HV \times IC}$ (SE)	$\beta_{\rm HV \times HC}$ (SE)
Variables	(1)	(2)	(3)	(4)	(5)	(6)
Wife's age						
Husband's age	30.37 (8.93)	1.06 (0.92)	1.42 (0.98)	1.43 (0.86)	-1.74 (1.39)	-0.52 (1.26)
Wife's education	35.83 (10.16)	1.24 (1.06)	1.21 (1.09)	2.14 (1.01)	-1.24 (1.57)	-2.20 (1.45)
Husband's education	5.68	0.16	0.33	0.30	-0.62	-0.84
	(3.27)	(0.34)	(0.35)	(0.33)	(0.51)	(0.47)
Wife's quality score	6.77	-0.42	-0.69	-0.40	0.88	-0.17
	(3.54)	(0.38)	(0.36)	(0.36)	(0.55)	(0.51)
Husband's quality score	4.17	-1.12	0.12	0.10	0.08	0.15
	(1.20)	(0.11)	(0.10)	(0.10)	(0.16)	(0.15)
Wife's raven score	4.16	-1.17	-0.08	-0.17	0.22	0.03
	(1.25)	(0.12)	(0.11)	(0.11)	(0.17)	(0.17)
Husband's raven score	4.05	0.29	0.11	0.22	-0.18	-0.27
	(1.95)	(0.19)	(0.19)	(0.19)	(0.28)	(0.27)
Wife's math score	4.98 (2.01)	0.04 (0.20)	0.17 (0.21)	0.32 (0.20)	$0.06 \\ (0.31)$	-0.36 (0.29)
Husband's math score	3.26	0.16	0.15	0.34	-0.09	-0.43
	(1.66)	(0.18)	(0.17)	(0.18)	(0.25)	(0.25)
Wife's avg. income last two months (MWK, W's report)	4.09	0.10	0.05	0.03	-0.08	-0.22
	(1.47)	(0.15)	(0.16)	(0.15)	(0.23)	(0.21)
	10,659.82	1,538.85	2,134.64	279.93	-3,429.05	-777.95
Husband's avg. income last two months (MWK, W's report)	(17,556.52)	(1,838.89)	(1,966.44)	(1,718.28)	(2,730.86)	(2,494.78)
	15,506.03	2,580.88	1,219.83	-116.41	-3,829.94	-958.75
	(23,030.80)	(2,386.19)	(2,433.71)	(2,200.71)	(3,490.23)	(3,316.66)
Wife's risk preference	2.90	0.21	0.59	0.36	-0.51	-0.13
	(2.44)	(0.24)	(0.26)	(0.25)	(0.36)	(0.35)
Husband's risk preference	4.60	-0.02	-0.13	-0.06	0.03	0.14
	(2.98)	(0.30)	(0.31)	(0.29)	(0.45)	(0.42)
Years married	9.91	0.97	1.31	1.54	-2.53	-1.47
	(8.48)	(0.86)	(0.93)	(0.87)	(1.28)	(1.21)
N of Children	2.63 (1.56)	-0.10 (0.16)	(0.03) (0.17)	(0.05) (0.16)	(0.03) (0.24)	(0.09)
Household members	5.02	-0.06	0.07	0.08	-0.08	0.09
	(1.86)	(0.20)	(0.21)	(0.19)	(0.29)	(0.27)
Avg. transfers (H to W) last two months (MWK, W's report)	4,911.73	759.54	1,122.84	146.72	-1,848.01	-221.20
	(8,097.11)	(823.87)	(833.50)	(764.26)	(1,192.98)	(1,146.98)
	3.31	-0.25	-0.12	0.01	0.37	0.20
MER	(0.86)	(0.09)	(0.09)	(0.08)	(0.13)	(0.12)

Notes: The table shows results from OLS regressions with enumerator fixed effects and Huber-White robust SEs. All MWK values are winsorized at 3 SDs. Wife's and husband's average income as well as transfers as reported by the wife. The regressions control for enumerator and compensation fixed-effects. No differences are significant after adjusting for false discovery rates (q-values).

Variable	Obs	Mean	Std. Dev.
Years married	675	9.58	7.87
Number of children	675	2.49	1.59
Age	675	30.47	8.53
Education	674	7.25	3.18
Wife's avg. income last two months (MWK)	675	15117.49	25907.76
Husband's avg. income last two months (MWK)	675	20396.84	31040.83
Avg. transfers (H to W) last two months (MWK)	675	10292.88	14735.31

Table C.9: Couple characteristics, market experiment

Notes: Kwacha values are winsorized at 3 SDs.

	Whole sample (N=1093)			Participation sample (N=786)				Whole sample (N=1093)
	Panel A: By price and low score							
	Initial score	Participate (%)	Foregone comp.	Initial score	# Errors corrected	Hiding fee	Final score	Total forgone
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Non-Expert	-2.148 (0.075)	-3.089 (4.878)	6.178 (9.756)	-2.122 (0.089)	0.413 (0.132)	39.013 (13.209)	-1.709 (0.152)	34.610 (12.034)
Intermediate Cost	0.065 (0.066)	0.056 (3.853)	-0.113 (7.707)	0.104 (0.074)	-0.142 (0.059)	12.511 (9.136)	-0.039 (0.081)	8.518 (9.117)
High Cost	0.031 (0.061)	(5.866) (3.757)	(11.733) (7.514)	0.014 (0.071)	-0.232	10.791 (10.063)	-0.218	16.502 (9.167)
Non-Expert*Intermediate Cost	0.064 (0.113)	(3.131) -2.756 (7.240)	5.513 (14 480)	0.126 (0.130)	-0.108 (0.174)	(10.000) 25.582 (26.068)	(0.019) (0.203)	(22.291) (21.216)
Non-Expert*High Cost	(0.110) 0.050 (0.105)	(1.210) 1.377 (7.115)	-2.754 (14 230)	(0.100) 0.070 (0.127)	-0.268 (0.158)	(20.000) 12.302 (27.956)	(0.200) -0.197 (0.194)	3.088 (21.446)
Mean (Low Cost & Expert) P-value (Expert vs. Non-Expert, Intermediate Cost) P-value (Expert vs. Non-Expert, High Cost)	$\begin{array}{c} (0.100) \\ 4.738 \\ 0.000 \\ 0.000 \end{array}$	(1.110) 75.781 0.276 0.743	$ \begin{array}{c} (11.233) \\ 48.438 \\ 0.276 \\ 0.743 \end{array} $	4.727 0.000 0.000	$\begin{array}{c} 0.130) \\ 0.418 \\ 0.006 \\ 0.085 \end{array}$	$\begin{array}{c} 41.753 \\ 0.004 \\ 0.036 \end{array}$	$\begin{array}{c} (0.101) \\ 5.144 \\ 0.000 \\ 0.000 \end{array}$	80.078 0.001 0.032

Table C.10: Outcomes in the signaling experiment, by alternative measures of expertise

		Pan	el B: By pri	ice and a	ecuracy of b	eliefs abou	it score	
	Initial score	Participate (%)	Foregone comp.	Initial score	# Errors corrected	Hiding fee	Final score	Total forgone
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Inaccurate Beliefs	-1.294	-2.292	4.584	-1.298	0.081	9.869	-1.217	10.307
	(0.115)	(4.545)	(9.090)	(0.130)	(0.112)	(11.195)	(0.149)	(10.880)
Intermediate Cost	-0.033	4.031	-8.063	0.052	-0.162	19.869	-0.110	8.718
	(0.091)	(4.018)	(8.035)	(0.104)	(0.072)	(11.152)	(0.099)	(10.611)
High Cost	-0.073	-4.892	9.784	0.013	-0.331	1.267	-0.318	7.073
	(0.086)	(4.083)	(8.166)	(0.101)	(0.063)	(10.819)	(0.102)	(9.834)
Inaccurate Beliefs*Intermediate Cost	0.474	-11.432	22.865	0.360	-0.062	-4.751	0.298	11.673
	(0.177)	(6.736)	(13.472)	(0.197)	(0.144)	(21.167)	(0.211)	(18.044)
Inaccurate Beliefs*High Cost	0.521	-1.301	2.602	0.283	0.026	28.300	0.308	22.132
	(0.166)	(6.519)	(13.037)	(0.189)	(0.131)	(22.920)	(0.207)	(18.269)
Mean (Low Cost & Accurate Beliefs)	4.577	76.577	46.847	4.576	0.529	52.941	5.106	87.387
P-value (High vs. Low Accuracy, Intermediate Cost)	0.000	0.006	0.006	0.000	0.828	0.777	0.000	0.128
P-value (High vs. Low Accuracy, High Cost)	0.000	0.442	0.442	0.000	0.120	0.057	0.000	0.028

Notes: The table shows results from OLS regressions with Huber-White robust SEs. Non-Expert is an indicator that takes the value 1 if the wife has a quality score that is lower than 4 (the median, 28% of women). Inaccurate beliefs is an indicator that is 1 if the absolute distance between the wife's belief about her quality score and her actual quality score is larger than 1 (the median, 40% of women). Foregone comp. is the amount of money wives left on the table by opting out of the game. All regressions include enumerator and compensation fixed effects. The p-value is the p-value from a Wald test comparing outcomes between Non-Expert and Expert wives or between wives with accurate or inaccurate beliefs when the hiding cost is high.

	Mean (SD)	$egin{array}{c} eta_{\mathrm{NE}} \ (\mathrm{SE}) \end{array}$	$egin{aligned} & eta_{\mathrm{Donated}} \ & (\mathrm{SE}) \end{aligned}$	$egin{aligned} & eta_{\mathrm{D}.*\mathrm{NE}} \ & (\mathrm{SE}) \end{aligned}$	$\beta_{\text{Effectiveness}}$ (SE)	$\beta_{\rm Eff.*NE}$ (SE)	$\beta_{\text{D.&Eff.}}$ (SE)	$\beta_{(D.\&Eff.)*NE}$ (SE)
Variables	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Age	30.47	1.25	0.37	-1.43	0.77	-0.30	0.81	-1.60
	(8.53)	(1.36)	(1.16)	(1.80)	(1.20)	(1.92)	(1.21)	(1.94)
Education	7.25	-1.39	0.39	-0.26	0.14	-0.71	-0.43	0.25
	(3.18)	(0.52)	(0.42)	(0.67)	(0.41)	(0.71)	(0.49)	(0.73)
Math Score	3.62	-0.74	-0.10	0.01	0.02	-0.14	-0.17	-0.11
	(1.48)	(0.23)	(0.20)	(0.31)	(0.20)	(0.32)	(0.21)	(0.33)
Wife's avg. income last two months (MWK)	15,117.49	-3,185.91	2,230.19	-988.70	-3,013.56	2,558.45	-5,539.24	-577.60
	(25, 907.76)	(4, 333.25)	(4, 115.68)	(6,076.41)	(3,853.76)	(6, 116.96)	(3, 384.75)	(5,219.28)
Husband's avg. income last two months (MWK)	20,396.84	-6,951.45	-2,043.04	-1,299.85	-6,685.62	2,291.62	-6,089.95	2,605.20
	(31,040.83)	(5, 149.33)	(4, 821.72)	(6,880.30)	(4, 495.60)	(6,909.47)	(4,589.39)	(7,040.73)
Risk Preferences	3.56	-0.54	0.11	0.10	0.48	-0.47	0.30	-0.05
	(2.70)	(0.36)	(0.40)	(0.53)	(0.42)	(0.54)	(0.39)	(0.54)
Years married	9.58	2.06	0.48	-1.98	2.00	-2.70	1.50	-1.25
	(7.87)	(1.07)	(1.01)	(1.53)	(1.12)	(1.73)	(1.11)	(1.75)
Number of children	2.49	0.18	-0.11	-0.13	0.53	-0.68	0.27	-0.49
	(1.59)	(0.22)	(0.20)	(0.30)	(0.29)	(0.40)	(0.22)	(0.32)
Household members	4.92	0.31	-0.03	-0.36	0.70	-0.98	0.27	-0.50
	(1.95)	(0.30)	(0.27)	(0.39)	(0.34)	(0.48)	(0.27)	(0.41)
Avg. transfers (H to W) last two months (MWK)	10,292.88	-3,265.25	748.86	-1,516.36	-3,908.02	4,838.72	-2,164.09	-713.67
~ ` ` ` ` ` ` ` ` ` ` ` ` ` ` ` ` ` ` `	(14, 735.31)	(2,332.99)	(2,445.84)	(3, 312.62)	(1, 899.48)	(3, 159.14)	(2,091.75)	(3,073.95)

Table C.11: The market experiment: Balance, by expertise (NE=non-expert wife) and sticker treatments

Notes: The table shows results from OLS regressions with enumerator fixed effects and Huber-White robust SEs. All MWK values are winsorized at 3 SDs. Non-Expert defined as before. Wife's and husband's average income as well as transfers are reported by the wife, the agent of the market experiment. The regressions control for enumerator fixed effects. No differences are significant after adjusting for false discovery rates (q-values).